# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# **FORM 10-Q**

$\times$	QUARTERLY REPORT PURSUAN	Γ TO SECTION 13 OR 15(d) OF THE S	SECURITIES EXCHANGE ACT OF 1934	
		For the quarterly period ended June 30,	2023	
		OR		
	TRANSITION REPORT PURSUAN	T TO SECTION 13 OR 15(d) OF THE S	SECURITIES EXCHANGE ACT OF 1934	
	1	For the transition period from to		
		Commission file number 001-34504		
	ADDIC		DOD ATION	
		IOMECARE COR		
		(Exact name of registrant as specified in its	charter)	
	Delaware		20-5340172	
	(State or other jurisdictio	n of	(I.R.S. Employer	
	incorporation or organiza		Identification No.)	
	6303 Cowboys Way, Suite	600		
	Frisco, TX		75034	
	(Address of principal executiv	•	(Zip Code)	
		(469) 535-8200 (Registrant's telephone number, includin	g area code)	
		Securities registered pursuant to Section 12(b) of		
	Title of each class	Trading Symbol(s)	Name of each exchange on which registere	d
	Common Stock, \$0.001 par value	ADUS	The Nasdaq Global Market	
		unt (1) has filed all reports required to be filed l	hy Section 12 on 15(d) of the Securities Eyekange A	et of
		shorter period that the registrant was required t	by Section 13 or 15(d) of the Securities Exchange A to file such reports), and (2) has been subject to such	
			ve Data File required to be submitted pursuant to R period that the registrant was required to submit su	
			ler, a non-accelerated filer, a smaller reporting comp "smaller reporting company" and "emerging growt	
Larg	e Accelerated Filer		Accelerated Filer	
	Accelerated Filer		Smaller Reporting Company	
Eme	rging Growth Company			
new		y check mark if the registrant has elected not to rided pursuant to Section 13(a) of the Exchang	o use the extended transition period for complying vec Act. $\square$	with any
	-	ant is a shell company (as defined in Rule 12b- coration had 16,214,653 shares of Common St		

# ADDUS HOMECARE CORPORATION

# FORM 10-Q

# INDEX

PART I. FINANCIAL INFORMATION	3
Item 1. Financial Statements (Unaudited)	3
Condensed Consolidated Balance Sheets as of June 30, 2023 and December 31, 2022	3
Condensed Consolidated Statements of Income For the Three and Six Months Ended June 30, 2023 and 2022	4
Condensed Consolidated Statement of Stockholders' Equity For the Three and Six Months Ended June 30, 2023 and 2022	5
Condensed Consolidated Statements of Cash Flows For the Six Months Ended June 30, 2023 and 2022	7
Notes to Condensed Consolidated Financial Statements	8
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	20
Item 3. Quantitative and Qualitative Disclosures About Market Risk	35
Item 4. Controls and Procedures	35
PART II. OTHER INFORMATION	36
<u>Item 1. Legal Proceedings</u>	36
Item 1A. Risk Factors	36
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	36
Item 3. Defaults Upon Senior Securities	36
Item 4. Mine Safety Disclosures	36
<u>Item 5. Other Information</u>	36
<u>Item 6. Exhibits</u>	37
2	

# PART I – FINANCIAL INFORMATION

# Item 1. Financial Statements

# ADDUS HOMECARE CORPORATION AND SUBSIDIARIES

# CONDENSED CONSOLIDATED BALANCE SHEETS As of June 30, 2023 and December 31, 2022 (Amounts and Shares in Thousands, Except Per Share Data) (Unaudited)

	Ju	ne 30, 2023	December 31, 2022		
Assets					
Current assets					
Cash	\$	84,188	\$	79,961	
Accounts receivable, net of allowances		104,252		125,501	
Prepaid expenses and other current assets		19,350		17,345	
Total current assets		207,790	'	222,807	
Property and equipment, net of accumulated depreciation and amortization		19,607	<u> </u>	21,182	
Other assets					
Goodwill		583,656		582,837	
Intangibles, net of accumulated amortization		68,859		72,188	
Operating lease assets, net		48,472		38,980	
Total other assets		700,987		694,005	
Total assets		928,384	\$	937,994	
Liabilities and stockholders' equity					
Current liabilities					
Accounts payable	\$	20,699	\$	22,092	
Accrued payroll		47,795		44,937	
Accrued expenses		31,966		27,507	
Operating lease liabilities, current portion		11,334		10,801	
Government stimulus advances		9,959		12,912	
Accrued workers' compensation insurance		12,149		12,897	
Total current liabilities		133,902		131,146	
Long-term liabilities					
Long-term debt, net of debt issuance costs		78,702		131,772	
Long-term operating lease liabilities		43,214		35,479	
Other long-term liabilities		6,215		6,057	
Total long-term liabilities		128,131		173,308	
Total liabilities	\$	262,033	\$	304,454	
Stockholders' equity					
Common stock—\$.001 par value; 40,000 authorized and 16,214 and 16,128 shares					
issued and outstanding as of June 30, 2023 and December 31, 2022, respectively	\$	16	\$	16	
Additional paid-in capital		398,492		393,208	
Retained earnings		267,843		240,316	
Total stockholders' equity		666,351		633,540	
Total liabilities and stockholders' equity	\$	928,384	\$	937,994	

# **CONDENSED CONSOLIDATED STATEMENTS OF INCOME**For the Three and Six Months Ended June 30, 2023 and 2022

(Amounts and Shares in Thousands, Except Per Share Data)
(Unaudited)

	For	For the three months ended June 30,			Fo	For the six months ended June 30			
		2023	2022			2023		2022	
Net service revenues	\$	259,980	\$	236,940	\$	511,579	\$	463,574	
Cost of service revenues		177,662		161,342		350,846		317,790	
Gross profit		82,318		75,598		160,733		145,784	
General and administrative expenses		57,397		55,095		113,757		108,247	
Depreciation and amortization		3,382		3,609		6,829		7,130	
Total operating expenses		60,779	-	58,704	-	120,586	-	115,377	
Operating income		21,539		16,894		40,147		30,407	
Interest income		(291)		(108)		(397)		(166)	
Interest expense		2,331		1,986		4,792		3,806	
Total interest expense, net		2,040		1,878		4,395		3,640	
Income before income taxes		19,499		15,016		35,752		26,767	
Income tax expense		4,647		3,766		8,225		7,047	
Net income	\$	14,852	\$	11,250	\$	27,527	\$	19,720	
Net income per common share						,			
Basic income per share	\$	0.93	\$	0.71	\$	1.72	\$	1.25	
Diluted income per share	\$	0.91	\$	0.70	\$	1.69	\$	1.22	
Weighted average number of common shares and potential common									
shares outstanding:									
Basic		16,002		15,854		15,975		15,833	
Diluted		16,283		16,131		16,304		16,113	

# CONDENSED CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY

For the Three and Six Months Ended June 30, 2023 (Amounts and Shares in Thousands) (Unaudited)

		For the Three Months Ended June 30, 2023										
	Common Stock			A	Additional Paid-in Capital	Retained Earnings		Sto	Total ockholders' Equity			
	Shares	A	mount									
Balance at April 1, 2023	16,204	\$	16	\$	395,879	\$	252,991	\$	648,886			
Issuance of shares of common stock under restricted stock award agreements	10		_		_		_		_			
Forfeiture of shares of common stock under restricted stock award agreements	_		_		_		_		_			
Stock-based compensation	_		_		2,613		_		2,613			
Shares issued for exercise of stock options	_		_		_		_		_			
Net income	_		_		_		14,852		14,852			
Balance at June 30, 2023	16,214	\$	16	\$	398,492	\$	267,843	\$	666,351			

			For the Six	Mon	ths Ended Jun	e 30, 2	023		
	Common Stock			A	Additional Paid-in Capital	Retained Earnings		Sto	Total ockholders' Equity
	Shares	A	mount						
Balance at January 1, 2023	16,128	\$	16	\$	393,208	\$	240,316	\$	633,540
Issuance of shares of common stock under restricted stock award agreements	86		_		_		_		_
Forfeiture of shares of common stock under restricted stock award agreements	_		_		_		_		_
Stock-based compensation	_		_		5,259		_		5,259
Shares issued for exercise of stock options	_		_		25		_		25
Net income	_		_		_		27,527		27,527
Balance at June 30, 2023	16,214	\$	16	\$	398,492	\$	267,843	\$	666,351

# CONDENSED CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY

For the Three and Six Months Ended June 30, 2022 (Amounts and Shares in Thousands) (Unaudited)

			For the Thr	ee Mo	nths Ended Ju	ne 30,	2022		
	Common Stock			A	Additional Paid-in Capital	Retained		Sto	Total ockholders' Equity
	Shares		Amount						
Balance at April 1, 2022	16,068	\$	16	\$	383,001	\$	202,761	\$	585,778
Issuance of shares of common stock under restricted stock award agreements	15		_		_		_		_
Forfeiture of shares of common stock under restricted stock award agreements	(3)		_		_		_		_
Stock-based compensation	_		_		2,680		_		2,680
Shares issued for exercise of stock options	1		_		69		_		69
Net income							11,250		11,250
Balance at June 30, 2022	16,081	\$	16	\$	385,750	\$	214,011	\$	599,777

			For the Six	Mon	ths Ended Jun	e 30, 2	022	
	Common Stock				Additional Paid-in Capital	Retained Earnings		Total ckholders' Equity
	Shares	A	mount					
Balance at January 1, 2022	15,940	\$	16	\$	380,037	\$	194,291	\$ 574,344
Issuance of shares of common stock under restricted stock award agreements	130		_		_		_	_
Forfeiture of shares of common stock under restricted stock award agreements	(3)		_		_		_	_
Stock-based compensation	_		_		5,165		_	5,165
Shares issued for exercise of stock options	14		_		548		_	548
Net income			_				19,720	19,720
Balance at June 30, 2022	16,081	\$	16	\$	385,750	\$	214,011	\$ 599,777

# CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

For the Six Months Ended June 30, 2023 and 2022 (Amounts in Thousands) (Unaudited)

For the Six Months

	Ended June 30,				
		2023		2022	
Cash flows from operating activities:					
Net income	\$	27,527	\$	19,720	
Adjustments to reconcile net income to net cash provided by (used in) operating activities, net of acquisitions:					
Depreciation and amortization		6,829		7,130	
Deferred income taxes		165		291	
Stock-based compensation		5,259		5,165	
Amortization of debt issuance costs under the credit facility		430		430	
Provision for credit losses		313		318	
Impairment of operating lease assets		_		1,174	
Changes in operating assets and liabilities, net of acquisitions:					
Accounts receivable		21,395		20,249	
Prepaid expenses and other current assets		(4,018)		7,909	
Government stimulus advances		(3,382)		12,562	
Accounts payable		(597)		1,398	
Accrued payroll		2,788		(5,756)	
Accrued expenses and other long-term liabilities		3,704		(8,088)	
Net cash provided by operating activities		60,413		62,502	
Cash flows from investing activities:					
Acquisitions of businesses, net of cash acquired		(940)		(84,490)	
Purchases of property and equipment		(1,771)		(1,538)	
Net cash used in investing activities		(2,711)		(86,028)	
Cash flows from financing activities:					
Payments on revolver loan — credit facility		(53,500)		(60,000)	
Proceeds from borrowings on revolver — credit facility				35,000	
Payments on term loan — credit facility		_		_	
Cash received from exercise of stock options		25		548	
Other		_		_	
Net cash used in financing activities		(53,475)		(24,452)	
Net change in cash		4,227		(47,978)	
Cash, at beginning of period		79,961		168,895	
Cash, at end of period	\$		\$	120,917	
Supplemental disclosures of cash flow information:					
Cash paid for interest	\$	4,416	\$	3,403	
Cash paid for income taxes	-	6,767	•		
Acquisition consideration payable included in accrued expenses		_		1,605	
T. T				_,	

# Notes to Condensed Consolidated Financial Statements (Unaudited)

## 1. Nature of Operations, Consolidation, and Presentation of Financial Statements

Addus HomeCare Corporation ("Holdings") and its subsidiaries (together with Holdings, the "Company", "we", "us" or "our") operate as a multistate provider of three distinct but related business segments providing in-home services. In its personal care services segment, the Company provides non-medical assistance with activities of daily living, primarily to persons who are at increased risk of hospitalization or institutionalization, such as the elderly, chronically ill or disabled. In its hospice segment, the Company provides physical, emotional and spiritual care for people who are terminally ill as well as related services for their families. In its home health segment, the Company provides services that are primarily medical in nature to individuals who may require assistance during an illness or after hospitalization and include skilled nursing and physical, occupational and speech therapy. The Company's payors include federal, state and local governmental agencies, managed care organizations, commercial insurers and private individuals.

# **Basis of Presentation**

The accompanying Unaudited Condensed Consolidated Financial Statements and related notes have been prepared in accordance with the rules and regulations of the U.S. Securities and Exchange Commission ("SEC") for Quarterly Reports on Form 10-Q. The accompanying balance sheet as of December 31, 2022 has been derived from the Company's audited financial statements for the year ended December 31, 2022 previously filed with the SEC. Accordingly, these financial statements do not include all of the information and note disclosures required by accounting principles generally accepted in the United States of America ("GAAP") for annual financial statements and should be read in conjunction with our consolidated financial statements and notes thereto for the year ended December 31, 2022 included in our Annual Report on Form 10-K, which includes information and disclosures not included herein.

In the opinion of management, these financial statements reflect all adjustments of a normal, recurring nature necessary for the fair statement of our financial position, results of operations, and cash flows for the interim periods presented in conformity with GAAP. Our results for any interim period are not necessarily indicative of results for a full year or any other interim period.

### **Principles of Consolidation**

These Unaudited Condensed Consolidated Financial Statements include the accounts of Addus HomeCare Corporation, and its subsidiaries. All intercompany balances and transactions have been eliminated in consolidation.

## 2. Summary of Significant Accounting Policies

# **Estimates**

The financial statements are prepared by management in conformity with GAAP and include estimated amounts and certain disclosures based on assumptions about future events. The Company's critical accounting estimates include the following areas: revenue recognition, goodwill and intangibles in business combinations and when required, the quantitative assessment of goodwill. Actual results could differ from those estimates.

# **Computation of Weighted Average Shares**

The following table sets forth the computation of basic and diluted common shares:

	For the Three Months (Amounts in th		For the Six Months I (Amounts in the	,
	2023	2022	2023	2022
Weighted average number of shares outstanding for basic per share				
calculation	16,002	15,854	15,975	15,833
Effect of dilutive potential shares:				
Stock options	237	241	245	237
Restricted stock awards	44	36	84	43
Adjusted weighted average shares outstanding for diluted per share calculation	16,283	16,131	16,304	16,113
Anti-dilutive shares:				
Stock options	61	111	61	111
Restricted stock awards	77	81	1	19

### Recently Adopted Accounting Pronouncements

In October 2021, the Financial Accounting Standards Board ("FASB") issued ASU No. 2021-08, Accounting for Contract Assets and Contract Liabilities from Contracts with Customers (Topic 805). This ASU requires an acquirer in a business combination to recognize and measure contract assets and contract liabilities (deferred revenue) from acquired contracts using the revenue recognition guidance in Topic 606. At the acquisition date, the acquirer applies the revenue model as if it had originated the acquired contracts. The ASU was adopted prospectively on January 1, 2023. The additional disclosures required did not have a material impact on our consolidated financial statements.

# **Recently Issued Accounting Pronouncements**

In March 2020, the FASB issued ASU 2020-04, Reference Rate Reform (Topic 848): Facilitation of the Effects of Reference Rate Reform on Financial Reporting. ASU 2020-04 provides optional expedients and exceptions for applying GAAP to contract modifications and hedging relationships, and other transactions subject to meeting certain criteria, that reference the London Inter-Bank Offered Rate ("LIBOR") or another reference rate expected to be discontinued. The ASU provides companies with optional guidance to ease the potential accounting burden associated with transitioning away from reference rates that are expected to be discontinued. Therefore, it was in effect for a limited time through December 31, 2022. The ASU could be adopted no later than December 1, 2022 with early adoption permitted. As discussed further in Note 7 and pursuant to the Third Amendment to Amended and Restated Credit Agreement dated as of April 26, 2023, the Company amended its credit facility to replace LIBOR with the secured overnight financing rate as administered by the Federal Reserve Bank of New York ("SOFR") as the benchmark reference rate for loans under its credit facility. The transition to SOFR did not and is not expected to have a material impact on the Company's results of operations or liquidity.

## 3. Leases

Amounts reported on the Company's Unaudited Condensed Consolidated Balance Sheets for operating leases were as follows:

	June 30, 2	023	December 31	l, 2022
Operating lease assets, net	\$	48,472	\$	38,980
Short-term operating lease liabilities		11,334		10,801
Long-term operating lease liabilities		43,214		35,479
Total operating lease liabilities	\$	54,548	\$	46,280

# Lease Costs

Components of lease costs were reported in general and administrative expenses in the Company's Unaudited Condensed Consolidated Statements of Income as follows:

	For the Three Mon (Amounts in		For the Six Months Ended June 30, (Amounts in Thousands)				
	 2023		2022		2023	2022	
Operating lease costs	\$ 3,292	\$	3,103	\$	6,334	\$	5,904
Short-term lease costs	284		911		700		1,662
Total lease costs	 3,576		4,014		7,034		7,566
Less: sublease income	(700)		(176)		(1,399)		(353)
Total lease costs, net	\$ 2,876	\$	3,838	\$	5,635	\$	7,213

# Lease Term and Discount Rate

Weighted average remaining lease terms and discount rates were as follows:

	June 30, 2023	December 31, 2022
Operating leases:		
Weighted average remaining lease term (years)	6.55	5.82
Weighted average discount rate	5.25 %	3.98%

# **Maturity of Lease Liabilities**

Remaining operating lease payments as of June 30, 2023 were as follows:

	Ope	erating Leases
	(Amou	nts in Thousands)
Due in the 12-month period ended June 30,		
2024	\$	13,803
2025		11,485
2026		8,535
2027		6,605
2028		5,532
Thereafter		19,698
Total future minimum rental commitments		65,658
Less: Imputed interest		(11,110)
Total lease liabilities	\$	54,548

# Supplemental cash flows information

	For the Six Months Ended June 30, (Amounts in Thousands)					
		2023		2022		
Cash paid for amounts included in the measurement of lease liabilities:						
Operating cash flows from operating leases	\$	7,010	\$		6,476	
Right-of-use assets obtained in exchange for lease obligations:						
Operating leases	\$	14,415	\$		11,319	

# 4. Goodwill and Intangible Assets

A summary of the goodwill and related adjustments is provided below:

		Hospice		Hospice Persona		Hospice Personal Care Home H		Personal Care		Personal Care		pice Personal Care		me Health	Total	
				(Amounts in	Thousa	nds)										
Goodwill as of December 31, 2022	\$	397,728	\$	152,688	\$	32,421	\$	582,837								
Additions for acquisitions		_		599				599								
Adjustments to previously recorded goodwill		_		_		220		220								
Goodwill as of June 30, 2023	\$	397,728	\$	153,287	\$	32,641	\$	583,656								

On January 1, 2023, the Company completed the acquisition of Coastal Nursecare of Florida, Inc. ("CareStaff") for approximately \$1.0 million. With the purchase of CareStaff, the Company expanded its personal care services segment in Florida. In connection with the Carestaff acquisition, the Company recognized goodwill in its personal care segment of \$0.6 million during the six months ended June 30, 2023.

The Company's identifiable intangible assets consist of customer and referral relationships, trade names and trademarks, non-competition agreements and state licenses. Amortization is computed using straight-line and accelerated methods based upon the estimated useful lives of the respective assets, which range from one to twenty-five years. Customer and referral relationships are amortized systematically over the periods of expected economic benefit, which range from five to ten years.

The carrying amount and accumulated amortization of each identifiable intangible asset category consisted of the following as of June 30, 2023:

	Customer and referral relationships	Trade names and trademarks (A	Non- competition agreements mounts in Thousand	State Licenses	Total
Intangible assets with indefinite lives				27,108	27,108
Intangible assets subject to amortization:					
Gross carrying amount	44,672	52,046	6,785	12,670	116,174
Accumulated amortization	(38,840)	(22,366)	(5,193)	(8,023)	(74,423)
Intangible assets subject to amortization, net	5,832	29,680	1,592	4,647	41,751
Total intangible assets at June 30, 2023	\$ 5,832	\$ 29,680	\$ 1,592	\$ 31,755	\$ 68,859

Amortization expense related to the intangible assets was \$1.7 million and \$3.5 million for the three and six months ended June 30, 2023, respectively, and \$1.8 million and \$3.6 million for the three and six months ended June 30, 2022, respectively. The weighted average remaining useful lives of identifiable intangible assets as of June 30, 2023 was 9.6 years.

## 5. Details of Certain Balance Sheet Accounts

Prepaid expenses and other current assets consisted of the following:

	June	30, 2023	Decembe	r 31, 2022			
	(Amounts in Thousan						
Prepaid payroll	\$	8,240	\$	7,566			
Prepaid workers' compensation and liability insurance		3,571		3,399			
Prepaid licensing fees		4,638		3,722			
Workers' compensation insurance receivable		582		666			
Other		2,319		1,992			
Total prepaid expenses and other current assets	\$	19,350	\$	17,345			

Accrued expenses consisted of the following:

	June 30, 2023	December 31, 2022
	(Amounts in	n Thousands)
Accrued health benefits	6,921	5,152
Payor advances (1)	2,136	4,473
Accrued professional fees	5,933	3,576
Accrued payroll and other taxes	8,747	6,175
Other	8,229	8,131
Total accrued expenses	\$ 31,966	\$ 27,507

<sup>(1)</sup> Represents the deferred portion of payments received from payors for COVID-19 reimbursements which will be recognized as we incur specific COVID-19 related expenses (including expenses related to securing and maintaining adequate personnel) or will be returned to the extent such related expenses are not incurred.

# 6. Government Actions to Mitigate COVID-19's Impact

In March 2020, the World Health Organization declared the novel coronavirus ("COVID-19") outbreak a global pandemic. Although the acute phase of the COVID-19 pandemic has faded, the World Health organization has ended its public health emergency declaration, and vaccines and booster shots for the COVID-19 virus have become widely available in the United States, COVID-19 has continued to result in a significant number of hospitalizations, and the future course of the pandemic remains uncertain. However, compared to earlier periods, the number of COVID-19 infections and related hospitalizations has significantly declined. We will continue to closely monitor the impact of COVID-19 on all aspects of our business, including the impacts to our employees, patients and suppliers.

In recognition of the significant threat to the liquidity of financial markets posed by the COVID-19 pandemic, the Federal Reserve and Congress took dramatic actions to provide liquidity to businesses and the banking system in the United States. One of the primary sources of relief for healthcare providers is the Coronavirus Aid, Relief, and Economic Security Act ("CARES Act"), which was expanded by the Paycheck Protection Program and Health Care Enhancement ("PPPHCE") Act, and the Consolidated Appropriations Act ("CAA"). Another relief package with numerous provisions that affect healthcare providers is the American Rescue Plan Act of 2021 ("ARPA").

# ARPA Spending Plans

The ARPA provides for \$350 billion in relief funding for eligible state, local, territorial, and Tribal governments to mitigate the fiscal effects of the COVID-19 public health emergency. Additionally, the law provides for a 10-percentage point increase in federal matching funds for Medicaid home and community-based services ("HCBS") from April 1, 2021, through March 31, 2022, provided the state satisfied certain conditions. States are permitted to use the state funds equivalent to the additional federal funds through March 31, 2025. States must use the monies attributable to this matching fund increase to supplement, not supplant, their level of state spending for the implementation of activities enhanced under the Medicaid HCBS in effect as of April 1, 2021.

HCBS spending plans for the additional matching funds vary by state, but common initiatives in which the Company is participating include those aimed at strengthening the provider workforce (e.g., efforts to recruit, retain, and train direct service providers). The Company is required to properly and fully document the use of such funds in reports to the state in which the funds originated. Funds may be subject to recoupment if not expended or if they are expended on non-approved uses. During the six months ended June 30, 2023, the Company received additional state funding provided by the ARPA in an aggregate amount of \$1.9 million. The Company recorded revenue of \$0.2 million for certain states that met the revenue recognition criteria. The Company deferred the remaining \$1.7 million, which was received from states with specific spending plans and reporting requirements. The Company utilized \$2.4 million and 4.8 million of these funds during the three and six months ended June 30, 2023, primarily for caregivers and adding support to recruiting and retention efforts, included as a reduction of cost of service revenues in the Company's Unaudited Condensed Consolidated Statements of Income. As of June 30, 2023, the deferred portion of ARPA funding of \$10.0 million is included within Government stimulus advances on the Company's Unaudited Condensed Consolidated Balance Sheets.

### Medicare sequester

The CARES Act and related laws temporarily lifted the Medicare sequester which would have otherwise reduced payments to Medicare providers by 2%, as required by the Budget Control Act of 2011, from May 1, 2020, through March 31, 2022. The sequestration payment adjustment was phased back in with a 1% reduction beginning April 1, 2022, and returned to 2% on July 1, 2022. These sequestration cuts have been extended through the first six months of 2032.

The ARPA increases the federal budget deficit in a manner that triggers an additional statutorily mandated sequestration under the Pay-As-You-Go Act of 2010 ("PAYGO Act"). As a result, an additional Medicare payment reduction of up to 4% was required to take effect in January 2022. However, Congress delayed implementation of this payment reduction until 2025.

In the hospice segment, Medicare sequester relief resulted in an increase in net service revenues of \$0.0 million and \$0.5 million for the three months ended June 30, 2023 and 2022, respectively and \$0.0 million and \$1.4 million for the six months ended June 30, 2023 and 2022 respectively. In the home health segment, Medicare sequester relief resulted in an increase in net service revenues of \$0.0 million and \$0.1 for the three months ended June 30, 2023 and 2022 respectively and \$0.0 million and \$0.3 million for the six months ended June 30, 2023 and 2022, respectively.

# 7. Long-Term Debt

Long-term debt consisted of the following:

	June 30, 2023	Dec	cember 31, 2022		
	(Amounts in	Thousan	ıds)		
Revolving loan under the credit facility	\$ \$ 81,353 \$				
Less unamortized issuance costs	(2,651)		(3,081)		
Long-term debt	\$ 78,702	\$	131,772		

#### Amended and Restated Senior Secured Credit Facility

On October 31, 2018, the Company entered into the Amended and Restated Credit Agreement, with certain lenders and Capital One, National Association, as a lender and as agent for all lenders, as amended by the First Amendment to Amended and Restated Credit Agreement, dated as of September 12, 2019, as further amended by the Second Amendment to Amended and Restated Credit Agreement, dated as of July 30, 2021, and as further amended by the Third Amendment to Amended and Restated Credit Agreement, dated as of April 26, 2023 (as described below, the "Third Amendment") (as amended, the "Credit Agreement," as used throughout this Quarterly Report on Form 10-Q, "credit facility" shall mean the credit facility evidenced by the Credit Agreement). The credit facility consists of a \$600.0 million revolving credit facility and a \$125.0 million incremental loan facility, which incremental loan facility may be for term loans or an increase to the revolving loan commitments. The maturity of this credit facility is July 30, 2026. On April 26, 2023, the Company entered into the Third Amendment to replace LIBOR with SOFR as the benchmark reference rate for loans under its credit facility. The Third Amendment did not amend any other terms of the Credit Agreement. Interest on the credit facility may be payable at (x) the sum of (i) an applicable margin ranging from 0.75% to 1.50% based on the applicable senior net leverage ratio plus (ii) a base rate equal to the greatest of (a) the rate of interest last quoted by The Wall Street Journal as the "prime rate," (b) the sum of the federal funds rate plus a margin of 0.50% and (c) the sum of Term SOFR (as published by the CME Group Benchmark Administrative Limited) for an interest period of one month for such applicable day plus 0.10% (not to be less than 0.00%), plus a margin of 1.00% or (y) the sum of (i) an applicable margin ranging from 1.75% to 2.50% based on the applicable senior net leverage ratio plus (ii) the rate per annum equal to the sum of Term SOFR (as published by the CME Group Benchmark Administrative Limited) for the applicable interest period plus 0.10% (not to be less than zero). Swing loans may not be SOFR loans. The transition to SOFR did not and is not expected to have a material impact on the Company's results of operations or liquidity.

Addus HealthCare, Inc. ("Addus HealthCare") is the borrower, and its parent, Holdings, and substantially all of Holdings' subsidiaries are guarantors under this credit facility, and it is collateralized by a first priority security interest in all of the Company's and the other credit parties' current and future tangible and intangible assets, including the shares of stock of the borrower and subsidiaries. The Credit Agreement contains affirmative and negative covenants customary for credit facilities of this type, including limitations on the Company with respect to liens, indebtedness, guaranties, investments, distributions, mergers and acquisitions and dispositions of assets. The availability of additional draws under this credit facility is conditioned, among other things, upon (after giving effect to such draws) the Total Net Leverage Ratio (as defined in the Credit Agreement) not exceeding 3.75:1.00. In certain circumstances, in connection with a Material Acquisition (as defined in the Credit Agreement), the Company can elect to increase its Total Net Leverage Ratio compliance covenant to 4.25:1.00 for the then current fiscal quarter and the three succeeding fiscal quarters.

The Company pays a fee ranging from 0.20% to 0.35% based on the applicable senior net leverage ratio times the unused portion of the revolving loan portion of the credit facility.

The Credit Agreement contains customary affirmative covenants regarding, among other things, the maintenance of records, compliance with laws, maintenance of permits, maintenance of insurance and property and payment of taxes. The Credit Agreement also contains certain customary financial covenants and negative covenants that, among other things, include a requirement to maintain a minimum Interest Coverage Ratio (as defined in the Credit Agreement), a requirement to stay below a maximum Total Net Leverage Ratio (as defined in the Credit Agreement) and a requirement to stay below a maximum permitted amount of capital expenditures. The Credit Agreement also contains restrictions on guarantees, indebtedness, liens, investments and loans, subject to customary carve outs, a restriction on dividends (provided that Addus HealthCare may make distributions to the Company in an amount that does not exceed \$7.5 million in any year absent of an event of default, plus limited exceptions for tax and administrative distributions), a restriction on the ability to consummate acquisitions (without the consent of the lenders) under its credit facility subject to compliance with the Total Net Leverage Ratio (as defined in the Credit Agreement) thresholds, restrictions on mergers, dispositions of assets, and affiliate transactions, and restrictions on fundamental changes and lines of business.

During the six months ended June 30, 2023, the Company did not draw on its credit facility and repaid \$53.5 million under the revolving credit facility. During the six months ended June 30, 2022, the Company drew \$35.0 million under its credit facility to fund, in part, the JourneyCare acquisition and repaid \$60 million under the revolving credit facility.

At June 30, 2023, the Company had a total of \$81.4 million of revolving loans, with an interest rate of 6.95%, outstanding on its credit facility. After giving effect to the amount drawn on its credit facility, approximately \$8.0 million of outstanding letters of credit and borrowing limits based on an advance multiple of adjusted EBITDA (as defined in the Credit Agreement), the Company had \$409.3 million of capacity and \$319.9 million available for borrowing under its credit facility. As of December 31, 2022, the Company had a total of \$134.9 million of revolving loans, with an interest rate of 6.13%, outstanding on its credit facility.

As of June 30, 2023, the Company was in compliance with all financial covenants under the Credit Agreement.

# 8. Income Taxes

The effective income tax rates were 23.8% and 25.1% for the three months ended June 30, 2023 and 2022, respectively. The effective income tax rates were 23.0% and 26.3% for the six months ended June 30, 2023 and 2022, respectively. The difference between our federal statutory rate of 21% and our effective income tax rates is principally due to the inclusion of state taxes and non-deductible compensation, partially offset by the use of federal employment tax credits.

# 9. Commitments and Contingencies

# **Legal Proceedings**

From time to time, the Company is subject to legal and/or administrative proceedings incidental to its business.

On June 2, 2021, the Company received a \$6.5 million Request for Repayment from Palmetto, GBA, LLC ("Palmetto"), a Medicare administrative contractor, regarding Ambercare Hospice Inc. ("Ambercare"), our subsidiary that provides hospice services in New Mexico. In 2018, the Office of Audit Services ("OAS"), under the HHS Office of Inspector General, initiated a clinical review of certain hospice claims billed during a timeframe from January 1, 2016 to December 31, 2017. The OAS review concluded that certain payments to Ambercare for hospice services during the review period were made in error. The Company acquired Ambercare in May 2018 and has a contractual right to full indemnification from any potential losses from the OAS review through the terms of the Ambercare purchase agreement. The Company disputes the results of the OAS review and related asserted billing errors and is in the process of filing administrative appeals. At this stage, the Company cannot predict the ultimate outcome of the appeal process.

It is the opinion of management that the outcome of pending legal and/or administrative proceedings will not have a material effect on the Company's Unaudited Condensed Consolidated Balance Sheets and Unaudited Condensed Consolidated Statements of Income.

## 10. Segment Information

Operating segments are defined as components of a company that engage in business activities from which it may earn revenues and incur expenses, and for which separate financial information is available and is regularly reviewed by the Company's chief operating decision makers, to assess the performance of the individual segments and make decisions about resources to be allocated to the segments. The Company operates as a multi-state provider of three distinct but related business segments providing in-home services.

In its personal care segment, the Company provides non-medical assistance with activities of daily living, primarily to persons who are at increased risk of hospitalization or institutionalization, such as the elderly, chronically ill or disabled. In its hospice segment, the Company provides physical, emotional and spiritual care for people who are terminally ill as well as related services for their families. In its home health segment, the Company provides services that are primarily medical in nature to individuals who may require assistance during an illness or after hospitalization and include skilled nursing and physical, occupational and speech therapy.

The tables below set forth information about the Company's reportable segments, along with the items necessary to reconcile the segment information to the totals reported in the accompanying Unaudited Condensed Consolidated Financial Statements. Segment assets are not reviewed by the Company's chief operating decision maker function and therefore are not disclosed below.

Segment operating income consists of revenue generated by a segment, less the direct costs of service revenues and general and administrative expenses that are incurred directly by the segment. Unallocated general and administrative costs are those costs for functions performed in a centralized manner and therefore not attributable to a particular segment. These costs include accounting, finance, human resources, legal, information technology, corporate office support and facility costs and overall corporate management.

			FOF U	ie i iiree Montii	s Ended .	Julie 30, 2023			
	(Amounts in Thousands)								
	Personal Care		Hospice		Hospice Home I			Total	
Net service revenues	\$	198,314	\$	50,210	\$	11,456	\$	259,980	
Cost of services revenues		143,972		26,606		7,084		177,662	
Gross profit		54,342		23,604		4,372		82,318	
General and administrative expenses		16,267		12,768		2,641		31,676	
Segment operating income	\$	38,075	\$	10,836	\$	1,731	\$	50,642	

For the Three Months Ended June 20, 2022

			For tl	For the Three Months Ended June 30, 2022									
		(Amounts in Thousands)											
	Personal Care Hospice		Personal Care Hospice		Ho	me Health		Total					
Net service revenues	\$	174,330	\$	52,074	\$	10,536	\$	236,940					
Cost of services revenues		128,682		25,522		7,138		161,342					
Gross profit		45,648		26,552		3,398		75,598					
General and administrative expenses		15,447		13,036		2,501		30,984					
Segment operating income	\$	30,201	\$	13,516	\$	897	\$	44,614					

	For the Three Mon	ths Ended J	une 30,
	 2023		2022
	(Amounts in	Thousands	)
Segment reconciliation:			
Total segment operating income	\$ 50,642	\$	44,614
Items not allocated at segment level:			
Other general and administrative expenses	25,721		24,111
Depreciation and amortization	3,382		3,609
Interest income	(291)		(108)
Interest expense	2,331		1,986
Income before income taxes	\$ 19,499	\$	15,016

		For the Six Months Ended June 30, 2023									
		(Amounts in Thousands)									
	Pe	Personal Care Hospice		are Hospice Home Health		Home Health		Total			
Net service revenues	\$	388,346	\$	99,292	\$	23,941	\$	511,579			
Cost of services revenues		282,355		53,873		14,618		350,846			
Gross profit		105,991		45,419		9,323		160,733			
General and administrative expenses		32,202		25,783		5,521		63,506			
Segment operating income	\$	73,788	\$	19,636	\$	3,803	\$	97,227			

			For t	he Six Months	Ended J	une 30, 2022	
				(Amounts in	Thousa	nds)	
	Pe	rsonal Care		Hospice	Ho	me Health	Total
Net service revenues	\$	343,962	\$	99,801	\$	19,811	\$ 463,574
Cost of services revenues		254,973		48,963		13,854	317,790
Gross profit		88,989		50,838		5,957	145,784
General and administrative expenses		30,451		24,748		4,860	60,059
Segment operating income	\$	58,538	\$	26,090	\$	1,097	\$ 85,725

	For the Six Mont	or the Six Months Ended June 30, 2022 (Amounts in Thousands)		
	 2023		2022	
	(Amounts in	Thousands	)	
Segment reconciliation:				
Total segment operating income	\$ 97,227	\$	85,725	
Items not allocated at segment level:				
Other general and administrative expenses	50,251		48,188	
Depreciation and amortization	6,829		7,130	
Interest income	(397)		(166)	
Interest expense	4,792		3,806	
Income before income taxes	\$ 35,752	\$	26,767	

# 11. Significant Payors

The Company's revenue by payor type was as follows:

Personal Care Segment	Fe	or the Three Mo	nths I	Ended June 30,		For the Six Months Ended June 30,					
	 2023	}		2022	?		202	23	202	2022	
	Amount Thousands)	% of Segment Net Service Revenues	(iı	Amount n Thousands)	% of Segment Net Service Revenues	Amoun (in Thousa		% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues	
State, local and other											
governmental programs	\$ 100,399	50.6	%\$	85,462	49.0 %	\$ 195	,721	50.4%	5\$ 169,370	49.2 %	
Managed care organizations	91,276	46.0		80,577	46.2	179	,176	46.1	157,967	45.9	
Private pay	4,137	2.2		4,610	2.7	8	3,363	2.2	9,236	2.7	
Commercial insurance	1,637	0.8		2,093	1.2	3	3,306	0.9	4,117	1.2	
Other	865	0.4		1,588	0.9	1	,781	0.4	3,272	1.0	
Total personal care segment net service revenues	\$ 198,314	100.0	%\$	174,330	100.0 %	\$ 388	3,346	100.0 %	\$ 343,962	100.0%	

Hospice Segment		F	or the Three Mon	ths Ended June 30,						
		2023	3	202	.2	202	23	2022		
		Amount Thousands)	% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues	
Medicare	\$	45,566	90.7 %	6\$ 47,152	90.59	%\$ 90 <b>,</b> 122	90.8 %	\$ 90,637	90.8%	
Commercial insurance		2,726	5.4	2,726	5.2	5,273	5.3	4,970	5.0	
Managed care organizations		1,567	3.1	1,968	3.8	3,214	3.2	3,683	3.7	
Other		352	8.0	228	0.5	684	0.7	511	0.5	
Total hospice segment net service revenues	<b>\$</b>	50,210	100.0 %	ó\$ 52,074	100.0	<b>%</b> \$ 99,293	100.0 %	\$ 99,801	100.0 %	

Home Health Segment	F	or the Three Mon	onths Ended June 30,				For the Six Months Ended June 30,					
	2023	}	2022			2023				2022		
	mount 'housands)	% of Segment Net Service Revenues		ount ousands)	% of Segment Net Service Revenues		Amount Thousands)	% of Segment Net Service Revenues	(iı	Amount n Thousands)	% of Segment Net Service Revenues	
Medicare	\$ 8,716	76.1 %	\$	7,592	72.1 %	\$	17,988	75.1 %	\$	14,404	72.7 %	
Managed care organizations	2,251	19.6		2,262	21.5		4,789	20.0		4,166	21.0	
Other	489	4.3		682	6.4		1,164	4.9		1,241	6.3	
Total home health segment net service revenues	\$ 11,456	100.0 %	\$	10,536	100.0%	\$	23,941	100.0%	\$	19,811	100.0 %	

The Company derives a significant amount of its revenue from its operations in Illinois, New Mexico and New York. The percentages of segment revenue for each of these significant states were as follows:

Personal Care Segment	F	or the Three Mon	nths Ended June 30,				1				
	2023	3	2022				2023	B	2022		
	Amount Thousands)	% of Segment Net Service Revenues	(in	Amount n Thousands)	% of Segment Net Service Revenues	(i	Amount n Thousands)	% of Segment Net Service Revenues	Amou		% of Segment Net Service Revenues
Illinois	\$ 103,864	52.4%	\$	88,797	50.9%	\$	202,279	52.1 %	\$ 17	3,480	50.4%
New Mexico	27,907	14.1		26,473	15.2		56,381	14.5	5	1,912	15.1
New York (1)	23,841	12.0		21,127	12.1		45,727	11.8	4	2,513	12.4
All other states	42,702	21.5		37,933	21.8		83,959	21.6	7	6,057	22.1
Total personal care segment net service revenues	\$ 198,314	100.0 %	\$	174,330	100.0 %	\$	388,346	100.0 %	\$ 34	3,962	100.0 %

In 2019, New York initiated a new Request For Offer ("RFO") process to competitively procure CDPAP fiscal intermediaries. The Company was not selected in the initial RFO process. We submitted a formal protest in response to the selection process, which was filed and accepted in March 2021. The New York fiscal year 2023 state budget, passed in April 2022, amends the current Fiscal Intermediary RFO process to authorize all fiscal intermediaries that submitted an RFO application and served at least 200 clients in New York City or 50 clients in other counties between January 1, 2020 and March 31, 2020 to contract with the New York State Department of Health and continue to operate in all counties contained in their application, if the fiscal intermediary submits an attestation and supporting information to the New York State Department of Health no later than November 29, 2022. The Company submitted an attestation on November 22, 2022. Under this provision, the Company is allowed to continue to contract with all of its current payors for CDPAP services as of the anticipated contract award date of April 1, 2023. On June 6, 2023, the New York State Department of Health notified the Company of a contract award. The Company has resumed services under this contract to all current payors.

Hospice Segment		F	or the Three Mon	nths Ended June 30,				For the Six Months Ended June 30,					
		2023		2022				2023	3	2022			
	-	amount 'housands)	% of Segment Net Service Revenues		Amount Thousands)	% of Segment Net Service Revenues	(i	Amount in Thousands)	% of Segment Net Service Revenues		mount housands)	% of Segment Net Service Revenues	
Ohio	\$	19,332	38.5 %	\$	17,245	33.1%	\$	37,783	38.1 %	\$	33,574	33.7 %	
Illinois		11,606	23.1		13,561	26.0		23,087	23.3		16,078	16.1	
New Mexico		6,540	13.0		7,846	15.1		13,026	13.1		23,102	23.1	
All other states		12,732	25.4		13,422	25.8		25,397	25.5		27,047	27.1	
Total hospice segment net service revenues	\$	50,210	100.0 %	\$	52,074	100.0 %	\$	99,293	100.0 %	\$	99,801	100.0 %	

Home Health Segment	F	or the Three Mont	onths Ended June 30,				For the Six Months Ended June 30,					
	2023		2022				202	3	2022			
	amount Thousands)	% of Segment Net Service Revenues		nount ousands)	% of Segment Net Service Revenues		Amount Thousands)	% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues		
New Mexico	\$ 8,093	70.6 %	\$	9,070	86.1 %	\$	17,209	71.9 %	\$ 16,579	83.7 %		
Illinois	3,363	29.4		1,466	13.9		6,732	28.1	3,232	16.3		
Total home health segment net service revenues	\$ 11,456	100.0 %	\$	10,536	100.0 %	\$	23,941	100.0 %	\$ 19,811	100.0 %		

A substantial portion of the Company's revenue and accounts receivable are derived from services performed for federal, state and local governmental agencies. We derive a significant amount of our net service revenues in Illinois, which represented 45.7%, and 43.8% of our net service revenues for the three months ended June 30, 2023, and 2022, respectively, and accounted for 45.4% and 41.6% of our net service revenues for the six months ended June 30, 2023 and 2022, respectively. The Illinois Department on Aging, the largest payor program for the Company's Illinois personal care operations, accounted for 21.5% and 20.7% of the Company's net service revenues for the three months ended June 30, 2023 and 2022, respectively, and accounted for 21.5% and 20.7% of the Company's net service revenues for the six months ended June 30, 2023 and 2022, respectively.

The related receivables due from the Illinois Department on Aging represented 13.0% and 18.0% of the Company's net accounts receivable at June 30, 2023 and December 31, 2022, respectively.

# 12. Subsequent Events

On August 1, 2023, the Company completed the acquisition of American Home Care, LLC, a Tennessee limited liability company ("AHC") and its subsidiaries, Homecare, LLC, a Tennessee limited liability company ("Homecare"), Tennessee Valley Home Care, LLC (d/b/a Tennessee Quality Care – Home Health), a Tennessee limited liability company ("TQC – Home Health"), and Tri-County Home Health and Hospice, LLC (d/b/a Tennessee Quality Care - Hospice), a Tennessee limited liability company ("TQC – Hospice", and collectively with AHC, Homecare, and TQC – Home Health "Tennessee Quality Care") for approximately \$106 million, with funding provided by drawing on the Company's revolving credit facility pursuant to the terms of the Membership Interests Purchase Agreement dated as of June 28, 2023. With the purchase of Tennessee Quality Care, the Company expanded its services within its hospice and home health segments in Tennessee. The initial accounting is not yet complete as the Company is currently in the process of valuing the assets acquired and liabilities assumed in the transaction. Therefore, the related business combination disclosures have not been presented.

### ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

You should read the following discussion together with our unaudited condensed consolidated financial statements and the related notes included elsewhere in this quarterly report on Form 10-Q. This discussion contains forward-looking statements about our business and operations. Statements that are predictive in nature, that depend upon or refer to future events or conditions or that include words like "believes," "belief," "expects," "plans," "anticipates," "intends," "projects," "estimates," "may," "might," "would," "should" and similar expressions are intended to be forward-looking statements as defined by the Private Securities Litigation Reform Act of 1995. These statements are based on the beliefs and assumptions of our management based on information currently available to management. Such forward-looking statements are subject to risks, uncertainties and other important factors that could cause actual results and the timing of certain events to differ materially from future results expressed or implied by such forward-looking statements. These risks and uncertainties include, but are not limited to: the impact of macroeconomic conditions, elevated global inflation and interest rates, legislative developments, trade disruptions and supply chain disruptions on our business and our customers' businesses; financial market instability and disruptions to the banking system due to bank failures, particularly in light of the closures of Silicon Valley Bank and Signature Bank in March 2023; business disruptions due to natural disasters, acts of terrorism, pandemics (including ongoing issues related to the COVID-19 pandemic), riots, civil insurrection or social unrest, looting, protests, strikes or street demonstrations; changes in operational and reimbursement processes and payment structures at the state or federal levels; changes in Medicaid, Medicare, other government program and managed care organizations policies and payment rates, and the timeliness of reimbursements received under government programs; changes in, or our failure to comply with, existing, federal and state laws or regulations, or our failure to comply with new government laws or regulations on a timely basis; competition in the healthcare industry; the geographical concentration of our operations; changes in the case mix of consumers and payment methodologies; operational changes resulting from the assumption by managed care organizations of responsibility for managing and paying for our services to consumers; the nature and success of future financial and/or delivery system reforms; changes in estimates and judgments associated with critical accounting policies; our ability to maintain or establish new referral sources; our ability to renew significant agreements or groups of agreements; our ability to attract and retain qualified personnel; federal, state and city minimum wage pressure, including any failure of any governmental entity to enact a minimum wage offset and/or the timing of any such enactment; changes in payments and covered services due to the overall economic conditions and deficit reduction measures by federal and state governments, and our expectations regarding these changes; cost containment initiatives undertaken by federal and state governmental and other thirdparty payors; our ability to access financing through the capital and credit markets; our ability to meet debt service requirements and comply with covenants in debt agreements; our ability to integrate and manage our information systems; any security breaches, cyber-attacks, loss of data, or cybersecurity threats or incidents, and any actual or perceived failures to comply with legal requirements related to the privacy of confidential consumer data and other sensitive information; the size and growth of the markets for our services, including our expectations regarding the markets for our services; the acceptance of privatized social services; eligibility standards and limits on services imposed by state governmental agencies; the potential for litigation, audits and investigations; discretionary determinations by government officials; our ability to successfully implement our business model to grow our business; our ability to continue identifying, pursuing, consummating and integrating acquisition opportunities and expand into new geographic markets; the impact of acquisitions and dispositions on our business, including the potential inability to realize the benefits of potential acquisitions; the potential impact of the discontinuation of LIBOR and the transition to SOFR; the effectiveness, quality and cost of our services; our ability to successfully execute our growth strategy; changes in tax rates; the impact of inclement weather or natural disasters; and various other matters, many of which are beyond our control. In addition, these forward-looking statements are subject to the risk factors set forth in Part I, Item 1A of our Annual Report on Form 10-K for the period ended December 31, 2022, filed with the SEC on February 28, 2023. You should carefully review all of these factors. Moreover, our business may be materially adversely affected by factors that are not currently known to us, by factors that we currently consider immaterial or by factors that are not specific to us, such as general economic conditions. These forward-looking statements were based on information, plans and estimates at the date of this report, and we assume no obligation to update any forward-looking statements to reflect changes in underlying assumptions or factors, new information, future events or other changes, except as may be required by law.

# Overview

We are a home care services provider operating three segments: personal care, hospice and home health. Our services are principally provided inhome under agreements with federal, state and local government agencies, managed care organizations, commercial insurers and private individuals. Our consumers are predominantly "dual eligible," meaning they are eligible to receive both Medicare and Medicaid benefits. Managed care revenues accounted for 36.6% and 35.8% of our net service revenues during the three months ended June 30, 2023 and 2022, respectively, and 36.6% and 35.8% of our net service revenues during the six months ended June 30, 2023 and 2022, respectively.

A summary of certain consolidated financial results is provided in the table below.

	For	the Three Mon	ths End	led June 30,	F	or the Six Mont	ths Ended June 30,		
		2023		2022		2023		2022	
Net service revenues by segment:		(Amounts in	Thousa	ands)		(Amounts in	Thousa	ınds)	
Personal care	\$	198,314	\$	174,330	\$	388,346	\$	343,962	
Hospice		50,210		52,074		99,292		99,801	
Home health		11,456		10,536		23,941		19,811	
Total net service revenues	\$	259,980	\$	236,940	\$	511,579	\$	463,574	
Net income	\$	14,852	\$	11,250	\$	27,527	\$	19,720	

As of June 30, 2023, we provided our services in 22 states through 204 offices. We served approximately 62,000 and 64,000 discrete individuals, respectively, during the six months ended June 30, 2023 and 2022. Our personal care segment also includes staffing services, with clients including assisted living facilities, nursing homes and hospice facilities.

### Acquisitions

In addition to our organic growth, we have grown through acquisitions that have expanded our presence in current markets, with the goal of having all three levels of in-home care in our markets or facilitating our entry into new markets where in-home care has been moving to managed care organizations.

On February 1, 2022, we completed the acquisition of the operations of JourneyCare Inc. ("JourneyCare"). The purchase price was approximately \$86.6 million, including the amount of acquired excess cash held by JourneyCare at the closing of the acquisition (approximately \$0.4 million). The JourneyCare acquisition was funded with a combination of a \$35.0 million draw on the Company's revolving credit facility and available cash. With the JourneyCare acquisition, the Company expanded its hospice services in Illinois.

On October 1, 2022, we completed the acquisition of Apple Home HealthCare, LTD ("Apple Home") for \$12.7 million, with funding provided by drawing on the Company's revolving credit facility. With the purchase of Apple Home, the Company expanded clinical services for its home health segment in Illinois.

On January 1, 2023, we completed the acquisition of Coastal Nursecare of Florida, Inc. ("CareStaff") for approximately \$1.0 million, with funding provided by available cash. With the purchase of CareStaff, the Company expanded its personal care services in Florida.

On August 1, 2023, we completed the acquisition of Tennessee Quality Care for approximately \$106 million, with funding provided by drawing on the Company's revolving credit facility. With the purchase of Tennessee Quality Care, the Company expanded its services within its hospice and home health segment in Tennessee.

# **COVID-19 Pandemic Update**

Compared to earlier periods, the number of COVID-19 infections and related hospitalizations has significantly declined. However, given the longer-term uncertainties associated with the COVID-19 pandemic, it is difficult to predict the effect and ultimate impact of the COVID-19 pandemic on the Company.

For the three and six months ended June 30, 2023, COVID-19 related expenses in our personal care segment were approximately \$0.6 million and \$1.3 million, respectively. For the three and six months ended June 30, 2022, COVID-19 related expenses in our personal care segment were approximately \$1.1 million and \$2.8 million, respectively. COVID-19 related expenses are included in cost of service revenue on the Consolidated Statements of Income. Additionally, we recognized revenue of \$1.8 million and \$3.0 million attributable to temporary rate increases from certain payors in our personal care segment for the six months ended June 30, 2023 and 2022, respectively.

As of June 30, 2023, the Company deferred the recognition of \$2.9 million of payments received from payors for COVID-19 reimbursement included within accrued expenses, which will be recognized as we incur specific expenses related to the pandemic, such as expenses related to acquiring additional personal protective equipment ("PPE") and COVID-19 related paid time off, or will be returned to the extent COVID-19-related expenses are not incurred.

The federal public health emergency declared by HHS expired May 11, 2023. We will continue to assess the impact and consequences of the COVID-19 pandemic and government responses to the pandemic, including the implementation of the CARES Act, the PPPHCE Act, the CAA, the ARPA, other stimulus and relief legislation, and the President's National COVID-19 Preparedness Plan, on our business, results of operations, financial condition and cash flows. Given the dynamic nature of these circumstances, we cannot currently predict with certainty the extent to which our business, results of operations, financial condition or liquidity will ultimately be impacted by the pandemic but it is not expected to have a material adverse impact. See Part I, Item 1A—Risk Factors—Risks Related to Economic Conditions and the COVID-19 Pandemic of our Annual Report on Form 10-K for the year ended December 31, 2022, filed with the SEC on February 28, 2023.

See "Liquidity and Capital Resources" below for additional information regarding funds received related to COVID-19 pandemic relief.

## Recruiting

As the labor market has tightened and unemployment has declined in comparison to earlier levels, the competition for new caregivers, including skilled healthcare staff, and support staff has increased. In addition, the United States economy continues to experience significant inflationary pressures and a competitive labor market. To the extent that we continue to experience a shortage of caregivers, it may hinder our ability to fully meet the continuing demand for both our non-clinical and clinical services. The increased staffing challenges, including COVID-19 related quarantine requirements and inflationary pressures, resulted in increased labor costs to satisfy our staffing requirements during the three and six months ended June 30, 2023, compared to 2022 in our non-clinical and clinical operations.

# Revenue by Payor and Significant States

Our payors are principally federal, state and local governmental agencies and managed care organizations. The federal, state and local programs under which the agencies operate are subject to legislative and budgetary changes and other risks that can influence reimbursement rates. We are experiencing a transition of business from government payors to managed care organizations, which we believe aligns with our emphasis on coordinated care and the reduction of the need for acute care.

Our revenue by payor and significant states by segment were as follows:

Personal Care Segment		For	the Three Months E	nded June 30,		For the Six Months Ended June 30,					
3		2023		2022		2023		2022			
	(ir	Amount n Thousands)	% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues		
State, local and other											
governmental programs	\$	100,399	50.6 %\$	85,462	49.0 %	<b>(\$</b> 195,721	50.4 %	169,370	49.2%		
Managed care organizations		91,276	46.0	80,577	46.2	179,176	46.1	157,967	45.9		
Private pay		4,137	2.2	4,610	2.7	8,363	2.2	9,236	2.7		
Commercial insurance		1,637	8.0	2,093	1.2	3,306	0.9	4,117	1.2		
Other		865	0.4	1,588	0.9	1,781	0.4	3,272	1.0		
Total personal care segment net service revenues	\$	198,314	100.0 %\$	174,330	100.0 %	<b>388,346</b>	100.0 %	343,962	100.0 %		
Illinois	\$	103,864	52.4 %\$	88,797	50.9 %	\$ 202,279	52.1 %	5 173,480	50.4%		
New Mexico		27,907	14.1	26,473	15.2	56,381	14.5	51,912	15.1		
New York (1)		23,841	12.0	21,127	12.1	45,727	11.8	42,513	12.4		
All other states		42,702	21.5	37,933	21.8	83,959	21.6	76,057	22.1		
Total personal care segment net service revenues	\$	198,314	100.0 %\$	174,330	100.0 %	\$ 388,346	100.0 %	343,962	100.0 %		

<sup>(1)</sup> Prior to June 6, 2023, when the New York State Department of Health notified the Company that it had received a contract award, the Company had suspended materially all of its new patient admissions under the New York CDPAP program.

Hospice Segment		For t	he Three Months	Ended June 30,		For the Six Months Ended June 30,					
		2023		2022		2023		2022			
	(i	Amount n Thousands)	% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues		
Medicare	\$	45,566	90.7 %	\$ 47,152	90.5 %	<b>6</b> \$ 90,122	90.8 %	90,637	90.8%		
Commercial insurance		2,726	5.4	2,726	5.2	5,273	5.3	4,970	5.0		
Managed care organizations		1,567	3.1	1,968	3.8	3,214	3.2	3,683	3.7		
Other		352	8.0	228	0.5	684	0.7	511	0.5		
Total hospice segment net service revenues	\$	50,210	100.0 %	\$ 52,074	100.0 %	<b>\$</b> 99,293	100.0 %	99,801	100.0 %		
Ohio	\$	19,332	38.5 %	\$ 17,245	33.1 %	<b>37,783</b>	38.1 %	33,574	33.7 %		
Illinois		11,606	23.1	13,561	26.0	23,087	23.3	16,078	16.1		
New Mexico		6,540	13.0	7,846	15.1	13,026	13.1	23,102	23.1		
All other states		12,732	25.4	13,422	25.8	25,397	25.5	27,047	27.1		
Total hospice segment net service revenues	\$	50,210	100.0%	\$ 52,074	100.0 %	<b>\$</b> 99,293	100.0 %	99,801	100.0 %		

Home Health Segment	For	the Three Months	s Ended June 30,		For the Six Months Ended June 30,					
-	2023		2022		2023		2022			
	Amount (Thousands)	% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues	Amount (in Thousands)	% of Segment Net Service Revenues		
Medicare	\$ 8,716	76.1 %\$	7,592	72.1 %	<b>6</b> \$ 17,988	75.1 %	14,404	72.7 %		
Managed care organizations	2,251	19.6	2,262	21.5	4,789	20.0	4,166	21.0		
Other	489	4.3	682	6.4	1,164	4.9	1,241	6.3		
Total home health segment net service revenues	\$ 11,456	100.0 %\$	5 10,536	100.0 %	%\$ 23,941	100.0 %	\$ 19,811	100.0 %		
New Mexico	\$ 8,093	70.6 %\$	9,070	86.1 %	<b>(</b> \$ 17,209	71.9 %	\$ 16,579	83.7 %		
Illinois	3,363	29.4	1,466	13.9	6,732	28.1	3,232	16.3		
Total home health segment net service revenues	\$ 11,456	100.0 %\$	10,536	100.0 %	<b>6</b> \$ 23,941	100.0 %	\$ 19,811	100.0 %		

We derive a significant amount of our net service revenues in Illinois, which represented 45.7% and 43.8% of our net service revenues for the three months ended June 30, 2023 and 2022, respectively, and accounted for 45.4% and 41.6% of our net service revenues for the six months ended June 30, 2023 and 2022, respectively.

A significant amount of our net service revenues are derived from one payor, the Illinois Department on Aging, the largest payor program for our Illinois personal care operations, which accounted for 21.5% and 20.7% of our net service revenues for the three months ended June 30, 2023 and 2022, respectively, and accounted for 21.5% and 20.7% of the Company's net service revenues for the six months ended June 30, 2023 and 2022, respectively.

# **Changes in Reimbursement Rates**

Illinois

On November 26, 2019, the City of Chicago voted to approve additional increases in the Chicago minimum wage to \$14 per hour beginning July 1, 2020 and to \$15 per hour beginning July 1, 2021. In each subsequent year, the City is required to raise the wage based on increases in the Consumer Price Index ("CPI") subject to a cap and other requirements. On July 1, 2022, the rate was adjusted to \$15.80 based on the increase in the CPI.

The Illinois fiscal year 2022 budget included an increase of hourly rates for in-home care services to \$24.96, to be effective January 1, 2022. On July 12, 2021, in connection with the temporary increase in federal funding for Medicaid home and community-based services authorized by the ARPA, the State of Illinois submitted its Initial Spending Plan and Narrative to CMS for approval. That plan included the acceleration by two months of the rate increase to \$24.96 from January 1, 2022, to November 1, 2021. The Company recognized \$3.6 million related to the rate increase for the year ended December 31, 2021.

The Illinois fiscal year 2023 budget included an increase of hourly rates for in-home care services to \$25.66, to be effective January 1, 2023. This increase offsets the \$0.40 increase in Chicago minimum wage that occurred on July 1, 2022. In addition, CMS approved a waiver amendment proposal submitted by the Illinois Department of Healthcare and Family Services with regard to its Persons who are Elderly program, further increasing in-home care rates to \$26.92, effective March 1, 2023.

Our business will benefit from the rate increases noted above as planned for 2023, but there is no assurance that there will be additional offsetting rate increases in Illinois for fiscal years beyond fiscal year 2023, and our financial performance will be adversely impacted for any periods in which an additional offsetting reimbursement rate increase is not in effect.

# Impact of Changes in Medicare and Medicaid Reimbursement

# Home Health

Home health services provided to Medicare beneficiaries are paid under the Medicare Home Health Prospective Payment System ("HHPPS"), which uses national, standardized 30-day period payment rates for periods of care that meet a certain threshold of home health visits (periods of care that do not meet the visit threshold are paid a per-visit payment rate for providing care). Although payment is made for each 30-day period, the HHPPS permits continuous 60-day certification periods through which beneficiaries are verified as eligible for the home health benefit. The daily home health payment rate is adjusted for case-mix and area wage levels. CMS uses the Patient-Driven Groupings Model ("PDGM") as the case-mix classification model to place periods of care into payment categories, classifying patients based on clinical characteristics and their resource needs. An outlier adjustment may be paid for periods of care where costs exceed a specific threshold amount.

CMS updates the HHPPS payment rates each calendar year. For calendar year 2023, CMS estimates that Medicare payments to home health agencies will increase by 0.7%. This is based on a home health payment update percentage of 4.0, which reflects a 4.1% market basket update reduced by a productivity adjustment of negative 0.1 percentage points, and an estimated 3.5% decrease associated with the transition to the PDGM that is intended to help achieve budget-neutrality on a prospective basis, among other changes. Home health providers that do not comply with quality data reporting requirements are subject to a 2-percentage point reduction to their market basket update. In addition, beginning January 1, 2022, Medicare requires home health agencies to submit a one-time Notice of Admission ("NOA") for each patient that establishes that the beneficiary is under a Medicare home health period of care. Failure to submit the NOA within five calendar days from the start of care will result in a reduction to the 30-day period payment amount for each day from the start of care date until the date the NOA is submitted.

CMS began implementing a nationwide expansion of the Home Health Value-Based Purchasing ("HHVBP") Model in January 2022. Under the model, home health agencies will receive increases or decreases to their Medicare fee-for-service payments of up to 5%, based on performance against specific quality measures relative to the performance of other home health providers. Data collected in each performance year will impact Medicare payments two years later. Calendar year 2023 is the first performance year under the expanded HHVBP Model, which will affect payments in calendar year 2025.

In certain states, payment of claims may be impacted by the Review Choice Demonstration for Home Health Services, a program intended to identify and prevent fraud, reduce the number of Medicare appeals and improve provider compliance with Medicare program requirements. The program applies to home health agencies in Illinois, Ohio, North Carolina, Florida and Texas and may expand, in the future, into additional states. Providers in states subject to the Review Choice Demonstration may initially select from the following claims review and approval processes: pre-claim review, post-payment review or a minimal post-payment review with a 25% payment reduction. Home health agencies that maintain high compliance levels will be eligible for additional options that may be less burdensome. We are currently unable to predict what impact, if any, this program may have on our result of operations or financial position.

The IMPACT Act requires HHS, together with the Medicare Payment Advisory Commission, to work toward a unified payment system for post-acute care services provided by home health agencies, inpatient rehabilitation facilities, skilled nursing facilities, and long-term care hospitals. A unified post-acute care payment system would pay post-acute care providers under a single framework according to a patient's characteristics, rather than based on the post-acute care setting where the patient receives treatment. As required under the statute, CMS and the HHS Office of the Assistant Secretary for Planning and Evaluation issued a report presenting a prototype for a unified post-acute care payment model in July 2022. CMS noted in its report the need for additional analyses and acknowledged that the universal implementation of a unified post-acute care payment system would require congressional action. The Medicare Payment Advisory Commission ("MedPAC") submitted a report to Congress in June 2023, concluding that designing a unified payment system is feasible, but cautioning that implementation of related policies would be complex. As Congress and other policymakers evaluate next steps, MedPAC suggested that CMS consider smaller-scale site-neutral policies to address some of the overlap in patients treated in different settings.

# Hospice

Hospice services provided to Medicare beneficiaries are paid under the Medicare Hospice Prospective Payment System, under which CMS sets a daily rate for each day a patient is enrolled in the hospice benefit. CMS updates these rates each federal fiscal year. Effective October 1, 2023, CMS will increase hospice payment rates by 3.1%. This reflects a 3.3% market basket increase and a negative 0.2 percentage point productivity adjustment. Hospices that do not satisfy quality reporting requirements will be subject to a 4-percentage point reduction to the market basket update.

Overall payments made by Medicare to each hospice provider number are subject to an inpatient cap and an aggregate cap, which is set each federal fiscal year. The inpatient cap limits the number of days of inpatient care to no more than 20% of total patient care days. The aggregate cap, which limits the total Medicare reimbursement that a hospice may receive based on an annual per-beneficiary cap amount and the number of Medicare patients served, was updated to \$33,494.01 for federal fiscal year 2024. If a hospice's Medicare payments exceed its inpatient or aggregate caps, it must repay Medicare the excess amount.

# New York Consumer Directed Personal Assistance Program ("CDPAP")

The CDPAP is a self-directed care alternative program that allows eligible individuals who need help with activities of daily living or skilled nursing services to choose their caregivers. We provide support services as a CDPAP fiscal intermediary.

In April 2022, the New York legislature passed the fiscal year 2023 state budget, which amended the Fiscal Intermediary Request For Offer ("RFO") process to authorize all fiscal intermediaries that submitted an RFO application and served at least 200 clients in New York City or 50 clients in other counties between January 1, 2020, and March 31, 2020, but that were not initially awarded a contract, to contract with the New York State Department of Health. These fiscal intermediaries are permitted to continue operating in all counties contained in their RFO application, provided they submitted an attestation and supporting information to the New York State Department of Health no later than November 29, 2022. The Company submitted an attestation on November 22, 2022, which allowed the Company to continue its CDPAP fiscal intermediary operations. On June 6, 2023, the New York State Department of Health notified the Company that it had received a contract award. Under this contract, the Company is providing services to all current payors and has resumed new fee-for-service patient admissions through County Social Service Departments in the CDPAP program.

# CMS Proposed Rule: "Ensuring Access to Medicaid Services"

In May 2023, CMS published a proposed rule, intended to improve access to services for Medicaid beneficiaries, that includes provisions related to HCBS payments. Specifically, in an effort to address workforce shortages, the proposed rule would (if finalized in its proposed form) require that a minimum of 80% of Medicaid payments in a state for home health aide, personal care services and some similar services be spent on compensation to direct care workers, in addition to related payment transparency requirements. CMS has proposed allowing states four years to implement changes required by a final rule. The ultimate impact of the 80% requirement, if finalized, could be adverse for periods after implementation, but other aspects of the rule could also benefit our business by improving access to services, depending on the policies ultimately set forth in any final rule. The comment period for the proposed rule ended July 1, 2023. The Company filed a comment letter on the proposed rule before this deadline, as did many other organizations, states and stakeholders.

# Components of our Statements of Income

Net Service Revenues

We generate net service revenues by providing our services directly to consumers and primarily on an hourly basis in our personal care segment, on a daily basis in our hospice segment and on an episodic basis in our home health segment. We receive payment for providing such services from our private consumers and payors, including federal, state and local governmental agencies, managed care organizations and commercial insurers.

In our personal care segment, net service revenues are principally provided based on authorized hours, determined by the relevant agency, at an hourly rate, which is either contractual or fixed by legislation, and are recognized at the time services are rendered. In our hospice segment, net service revenues are provided based on daily rates for each of the levels of care and are recognized as services are provided. In our home health segment, net service revenues are based on an episodic basis at a stated rate and recognized based on the number of days elapsed during a period of care within the reporting period. We also record estimated implicit price concessions (based primarily on historical collection experience) related to uninsured accounts to record revenues.

### Cost of Service Revenues

We incur direct care wages, payroll taxes and benefit-related costs in connection with providing our services. We also provide workers' compensation and general liability coverage for our employees. Employees are also reimbursed for their travel time and related travel costs in certain instances.

# General and Administrative Expenses

Our general and administrative expenses include our costs for operating our network of local agencies and our administrative offices. Our agency expenses consist of costs for supervisory personnel, our community care supervisors and office administrative costs. Personnel costs include wages, payroll taxes and employee benefits. Facility costs include rents, utilities, and postage, telephone and office expenses. Our corporate and support center expenses include costs for accounting, information systems, human resources, billing and collections, contracting, marketing and executive leadership. These expenses consist of compensation, including stock-based compensation, payroll taxes, employee benefits, legal, accounting and other professional fees, travel, general insurance, rents, provision for doubtful accounts and related facility costs. Expenses related to streamlining our operations such as costs related to terminated employees, termination of professional services relationships, other contract termination costs and asset write-offs are also included in general and administrative expenses.

# Depreciation and Amortization Expenses

Depreciable assets consist principally of furniture and equipment, network administration and telephone equipment and operating system software. Depreciable and leasehold assets are depreciated or amortized on a straight-line method over their useful lives or, if less and if applicable, their lease terms. We amortize our intangible assets with finite lives, consisting of customer and referral relationships, trade names, trademarks and non-competition agreements, using straight line or accelerated methods based upon their estimated useful lives.

# Interest Expense

Interest expense is reported when incurred and principally consists of interest and unused credit line fees on the credit facility.

### Income Tax Expense

All of our income is from domestic sources. We incur state and local taxes in the states in which we operate. The difference between our federal statutory rate of 21% and our effective income tax rates is principally due to the inclusion of state taxes and non-deductible compensation, partially offset by the use of federal employment tax credits.

# Results of Operations — Consolidated

Three Months Ended June 30, 2023 Compared to Three Months Ended June 30, 2022

The following table sets forth, for the periods indicated, consolidated results of operations.

	1	For the Three Mont				
	202	23	202	2	Chang	ge
	Amount	% Of Net Service Revenues	Amount	% Of Net Service Revenues	Amount	%
		(Am	ounts in Thousands	, Except Percenta	iges)	
Net service revenues	\$ 259,980	100.0 %	\$ 236,940	100.0	% \$ 23,040	9.7 %
Cost of service revenues	177,662	68.3	161,342	68.1	16,320	10.1
Gross profit	82,318	31.7	75,598	31.9	6,720	8.9
General and administrative expenses	57,397	22.1	55,095	23.3	2,302	4.2
Depreciation and amortization	3,382	1.3	3,609	1.5	(227)	(6.3)
Total operating expenses	60,779	23.4	58,704	24.8	2,075	3.5
Operating income	21,539	8.3	16,894	7.1	4,645	27.5
Interest income	(291)	(0.1)	(108)	_	(183)	169.6
Interest expense	2,331	0.9	1,986	0.8	345	17.4
Total interest expense, net	2,040	0.8	1,878	0.8	162	8.6
Income before income taxes	19,499	7.5	15,016	6.3	4,483	29.9
Income tax expense	4,647	1.8	3,766	1.6	881	23.4
Net income	\$ 14,852	5.7 %	\$ 11,250	4.7	% <b>\$</b> 3,602	32.0 %

Net service revenues increased by 9.7% to \$260.0 million for the three months ended June 30, 2023 compared to \$236.9 million for the three months ended June 30, 2022. Revenue increased by \$23.9 million in our personal care segment, decreased by \$1.9 million in our hospice segment and increased by \$1.0 million in our home health segment during the three months ended June 30, 2023, compared to the same period in 2022. The increase in our personal care segment was mainly due to an increase in revenues per billable hour for the three months ended June 30, 2023.

Gross profit, expressed as a percentage of net service revenues, decreased to 31.7% for the three months ended June 30, 2023, compared to 31.9% for the same period in 2022, due to a decline in revenue and gross profit in the hospice segment combined with an increase in direct payroll and benefits as a percentage of net service revenues of 0.7% for the three months ended June 30 2023.

General and administrative expenses increased to \$57.4 million for the three months ended June 30, 2023, as compared to \$55.1 million for the three months ended June 30, 2022. The increase in general and administrative expenses was primarily due to an increase in administrative employee wages, bonus, taxes and benefit costs of \$2.5 million, offset by a decrease in acquisition-related expense of \$0.2 million. General and administrative expenses, expressed as a percentage of net service revenues decreased to 22.1% for the three months ended June 30, 2023, from 23.3% for the three months ended June 30, 2022.

Interest expense increased to \$2.3 million for the three months ended June 30, 2023 from \$2.0 million for the three months ended June 30, 2022. The increase in interest expense was primarily due to increased interest rates under our credit facility for the three months ended June 30, 2023 compared to the three months ended June 30, 2022.

The effective income tax rate was 23.8% and 25.1% for the three months ended June 30, 2023 and 2022, respectively.

Six Months Ended June 30, 2023 Compared to Six Months Ended June 30, 2022

The following table sets forth, for the periods indicated, our consolidated results of operations.

		For the six mor				
	202	3	202	2	Chang	<u> </u>
	Amount	% Of Net Service Revenues	Amount	% Of Net Service Revenues	Amount	%
			(Amounts in Thousands	ges)		
Net service revenues	\$ 511,579	100.0	% \$ 463,574	100.0	% \$ 48,005	10.4 %
Cost of service revenues	350,846	68.6	317,790	68.6	33,056	10.4
Gross profit	160,733	31.4	145,784	31.4	14,949	10.3
General and administrative expenses	113,757	22.2	108,247	23.4	5,510	5.1
Depreciation and amortization	6,829	1.3	7,130	1.5	(301)	(4.2)
Total operating expenses	120,586	23.6	115,377	24.9	5,209	4.5
Operating income	40,147	7.7	30,407	6.5	9,740	32.0
Interest income	(397)	(0.1)	(166)		(231)	139.0
Interest expense	4,792	0.9	3,806	8.0	986	25.9
Total interest expense, net	4,395	0.9	3,640	0.8	755	20.7
Income before income taxes	35,752	7.0	26,767	5.8	8,985	33.6
Income tax expense	8,225	1.6	7,047	1.5	1,178	16.7
Net income	\$ 27,527	5.4	% \$ 19,720	4.3	% \$ 7,807	39.6 %

Net service revenues increased by 10.4% to \$511.6 million for the six months ended June 30, 2023 compared to \$463.6 million for the six months ended June 30, 2022. Net service revenue increased by \$44.3 million in our personal care segment and by \$4.1 million in our home health segment during the six months ended June 30, 2023, compared to the same period in 2022. During the six months ended June 30, 2023, the increase in our personal care segment revenue was primarily due to an increase in revenue per patient day, attributable to the rate increase discussed above, compared to the same period in 2022.

Gross profit, expressed as a percentage of net service revenues, remained at 31.4% for the six months ended June 30, 2023.

General and administrative expenses increased to \$113.8 million for the six months ended June 30, 2023 as compared to \$108.2 million for the six months ended June 30, 2022. The increase in general and administrative expenses was primarily due to acquisitions and wage increases that resulted in an increase in administrative employee wages, bonus, taxes and benefit costs of \$5.5 million. General and administrative expenses, expressed as a percentage of net service revenues decreased to 22.2% for the six months ended June 30, 2023, from 23.4% for the six months ended June 30, 2022.

Interest expense increased to \$4.8 for the six months ended June 30, 2023, as compared to \$3.8 million for the six months ended June 30, 2022. The increase in interest expense was primarily due to increased interest rates under our credit facility for the six months ended June 30, 2023, compared to the six months ended June 30, 2022.

The effective income tax rate was 23.0% and 26.3% for the six months ended June 30, 2023 and 2022, respectively.

# Results of Operations - Segments

The following tables and related analysis summarize our operating results and business metrics by segment:

#### **Personal Care Segment**

			For the T	hree Month	s Ended June 30,		For the Six Months Ended June 30,									
		202	23	2	022	Cha	nge	2023		2022		Change				
	A	Amount	% of Segment Net Service Revenues	Amount	% of Segment Net Service Revenues	Amount	%	Amount	% of Segment Net Service Revenues	Amount	% of Segment Net Service Revenues	Amount	%			
			(Amounts i	n Thousands	s, Except Percent	ages)			(Amounts in	n Thousand	s, Except Percenta	ges)				
Operating Results																
Net service revenues	\$	198,314	100.0 %	174,330	100.0 %	\$ 23,984	13.8 % \$	388,346	100.0 % \$	343,962	100.0 % \$	44,384	12.9 %			
Cost of services revenues		143,972	72.6	128,682	73.8	15,290	11.9	282,355	72.7	254,973	74.1	27,382	10.7			
Gross profit		54,342	27.4	45,648	26.2	8,694	19.0	105,991	27.3	88,989	25.9	17,002	19.1			
General and administrative expenses		16,267	8.2	15,447	8.9	820	5.3	32,202	8.3	30,451	8.9	1,751	5.8			
Segment operating income	\$	38,075	19.2 %	30,201	17.3 %	\$ 7,874	26.1 % \$	73,788	19.0 % \$	58,538	17.0 % \$	15,250	26.1 %			
Business Metrics (Actual Numbers, Except Billable Hours in Thousands)																
Locations at period end								157		161						
Average billable census * (1)		39,099		37,501		1,598	4.3 %	38,707		37,041		1,666	4.5 %			
Billable hours * (2)		7,681		7,373		308	4.2	15,274		14,474		800	5.5			
Average billable hours per census per month * (2)		65.3		65.2		0.1	0.2	65.6		64.8		0.8	1.2			
Billable hours per business day * (2)		118,177		113,426		4,751	4.2	117,491		112,198		5,293	4.7			
Revenues per billable hour * (2)	\$	25.57	;	23.58		\$ 1.99	8.4 % \$	25.27	\$	23.61	\$	1.66	7.0 %			
Same store growth revenue % * (3)		12.6 %	ó	2.5 9	6			11.7 %	ó	1.7 %	ó					

- (1) Average billable census is the number of unique clients receiving a billable service during the year and is the total census divided by months in operation during the period.
- (2) Billable hours is the total number of hours served to clients during the period. Average billable hours per census per month is billable hours divided by average billable census. Billable hours per day is total billable hours divided by the number of business days in the period. Revenues per billable hour is revenue, attributed to billable bonus hours, divided by billable hours.
- (3) Same store growth reflects the change in year-over-year revenue for the same store base. We define the same store base to include those stores open for at least 52 full weeks. This measure highlights the performance of existing stores, while excluding the impact of acquisitions, new store openings and closures, and ARPA associated revenue from this calculation.
- \* Management deems these metrics to be key performance indicators. Management uses these metrics to monitor our performance, both in our existing operations and acquisitions. Many of these metrics serve as the basis of reported revenues and assessment of these provide direct correlation to the results of operations from period to period and facilitate comparison with the results of our peers. Historical trends established in these metrics can be used to evaluate current operating results, identify trends affecting our business, determine the allocation of resources and assess the quality and potential variability of our cash flows and earnings. We believe they are useful to investors in evaluating and understanding our business but should not be used solely in assessing the Company's performance. These key performance indicators should not be considered superior to, as a substitute for or as an alternative to, and should be considered in conjunction with, the GAAP financial measures presented herein to fully evaluate and understand the business as a whole. These measures may not be comparable to similarly titled performance indicators used by other companies.

The personal care segment derives a significant amount of its net service revenues from operations in Illinois, which represented 45.7% and 43.8% of our net service revenues for the three months ended June 30, 2023 and 2022, respectively, and accounted for 45.4% and 41.6% of our net service revenues for the six months ended June 30, 2023 and 2022, respectively. One payor, the Illinois Department on Aging, accounted for 21.5% and 20.7% of net service revenues for the three months ended June 30, 2023 and 2022, respectively, and accounted for 21.5% and 20.7% of net service revenues for the six months ended June 30, 2023 and 2022, respectively.

Net service revenues from state, local and other governmental programs accounted for 50.6% and 49.0% of personal care segment net service revenues for the three months ended June 30, 2023 and 2022, respectively. Managed care organizations accounted for 46.0% and 46.2% of personal care segment net service revenues for the three months ended June 30, 2023 and 2022, respectively, with commercial insurance, private pay and other payors accounting for the remainder of personal care segment net service revenues. Net service revenues from state, local and other governmental programs accounted for 50.4% and 49.2% of net service revenues for the six months ended June 30, 2023 and 2022, respectively. Managed care organizations accounted for 46.1% and 45.9% of personal care segment net service revenues for the six months ended June 30, 2023 and 2022, respectively with commercial insurance, private pay and other payors accounting for the remainder of net service revenues.

Personal care segment net service revenues increased by 13.8% and 12.9% for the three and six months ended June 30, 2023, respectively, compared to the three and six months ended June 30, 2022. Net service revenues included a 7.0% and 5.3% increase in revenues per billable hour for the three and six months ended June 30, 2023, respectively, mainly attributed to rate increases discussed above, as compared to the three and six months ended June 30, 2022. The Company experienced an increase in New York net service revenues of \$0.4 million and \$0.7 million for the three and six months ended June 30, 2023, primarily driven by an increase in participation in the New York CDPAP program as discussed above, compared to 2022.

Gross profit, expressed as a percentage of net service revenues, increased to 27.4% for the three months ended June 30, 2023 from 26.2% for the three months ended June 30, 2022. This increase was primarily due to decreases in direct payroll and benefits as a percentage of net service revenues of 1.1% for the three months ended June 30, 2023. Gross profit expressed as a percentage of net service revenues, increased to 27.3% for the six months ended June 30, 2023 from 25.9% for the six months ended June 30, 2022. This increase was primarily due to a decrease of 0.8% in direct payroll and benefits as a percentage of net service revenues for the six months ended June 30, 2023 as compared to the six months ended June 30, 2022.

General and administrative expenses, expressed as a percentage of net service revenues, was 8.2% and 8.9% for the six months ended June 30, 2023 and 2022, respectively.

# **Hospice Segment**

	For the Three Months Ended June 30,									For the Six Months Ended June 30,									
		20	)23	2022				Chan	ge	2023				2022			Change		
	Ā	Amount	% of Segment Net Service Revenues	A	mount	% o Segme Net Ser Reven	ent vice	Amount	%	Amo	ount	% of Segme Net Serv Revenu	nt rice	Amount	% Segn Net Se Reve	nent ervice	Amount	Ç	%
			(Amounts	n Tl	ousands, E	xcept Pe	rcentages	s)				(Aı	nounts i	n Thousands	, Except	Percentag	es)		
Operating Results																			
Net service revenues	\$	50,210	100.0 %	\$	52,074		100.0 % 5	(1,864)	-3.6 % \$	9	9,292	1	00.0 % \$	99,801		100.0 % \$	(509)		-0.5 %
Cost of services revenues		26,606	53.0		25,522		49.0	1,084	4.2	5	3,873		54.3	48,963		49.1	4,910		10.0
Gross profit		23,604	47.0		26,552		51.0	(2,948)	-11.1	4	5,418		45.7	50,838		50.9	(5,420)	-	10.7
General and administrative expenses		12,768	25.4		13,036		25.0	(268)	-2.1	2	25,783		26.0	24,748		24.8	1,035		4.2
Segment operating income	\$	10,836	21.6 %	\$	13,516		26.0 % 5	(2,680)	-19.8 % \$	5 1	9,636		19.8 % \$	26,090		26.1 % \$	(6,454)		24.7 %
Business Metrics (Actual Numbers)																			
Locations at period end											34			33					
Admissions * (1)		3,076			3,281			(205)	-6.2 %		6,400			6,596			(196)		-3.0 %
Average daily census * (2)		3,225			3,333			(108)	-3.2		3,210			3,323			(113)		-3.4
Average discharge length of stay * (3)	)	94			84			10	12.1		91			84			7		8.4
Patient days * (4)		293,502			303,289			(9,787)	-3.2	58	31,053			578,777			2,276		0.4
Revenue per patient day * (5)	\$	174.32		\$	171.70		9	2.62	1.5 %	\$ 1	75.26		\$	172.43		\$	2.83		1.6 %
Organic growth																			
- Revenue * <sup>(6)</sup>		(1.1) %	6		2.5 %						0.5 %			3.4 %	6				
<ul> <li>Average daily census * <sup>(6)</sup></li> </ul>		(3.2) %	ó		6.1 %						1.4 %			6.6 %	ó				

- (1) Represents referral process and new patients on service during the period.
- (2) Average daily census is total patient days divided by the number of days in the period.
- (3) Average length of stay is the average number of days a patient is on service, calculated upon discharge, and is total patient days divided by total discharges in the period.
- (4) Patient days is days of service for all patients in the period.
- (5) Revenue per patient day is hospice revenue divided by the number of patient days in the period.
- (6) Revenue organic growth and average daily census organic growth reflect the change in year-over-year revenue and average daily census for the same store base. We define the same store base to include those stores open for at least 52 full weeks. These measures highlight the performance of existing stores, while excluding the impact of acquisitions, new store openings and closures.

\* Management deems these metrics to be key performance indicators. Management uses these metrics to monitor our performance, both in our existing operations and acquisitions. Many of these metrics serve as the basis of reported revenues and assessment of these provide direct correlation to the results of operations from period to period and facilitate comparison with the results of our peers. Historical trends established in these metrics can be used to evaluate current operating results, identify trends affecting our business, determine the allocation of resources and assess the quality and potential variability of our cash flows and earnings. We believe they are useful to investors in evaluating and understanding our business but should not be used solely in assessing the Company's performance. These key performance indicators should not be considered superior to, as a substitute for or as an alternative to, and should be considered in conjunction with, the GAAP financial measures presented herein to fully evaluate and understand the business as a whole. These measures may not be comparable to similarly titled performance indicators used by other companies.

The hospice segment generates net service revenues by providing care to patients with a life expectancy of six months or less, as well as related services for their families. Hospice offers four levels of care, as defined by Medicare, to meet the varying needs of patients and their families. The four levels of hospice include routine home care, continuous home care, general inpatient care and respite care. Our hospice segment principally provides routine home care and continuous home care services, and with the JourneyCare acquisition, expanded into providing general inpatient care services. In our hospice segment, net service revenues from Medicare accounted for 90.7% and 90.5% of total hospice segment net service revenues for the three months ended June 30, 2023 and 2022, respectively, and 90.8% for both the six months ended June 30, 2023 and 2022. Net service revenues from managed care organizations accounted for 3.1% and 3.8% of total hospice segment net service revenues for each the three months ended June 30, 2023 and 2022, respectively, and 3.2% and 3.7% for the six months ended June 30, 2023 and 2022, respectively.

Hospice net service revenues decreased by \$1.9 million and \$0.5 for the three and six months ended June 30, 2023, respectively, compared to the three and six months ended June 30, 2022, primarily attributed to minimal patient day growth.

Gross profit, expressed as a percentage of net service revenues, was 47.0% and 51.0% for the three months ended June 30, 2023 and 2022, respectively, and 45.7% and 50.9%, for the six months ended June 30, 2023 and 2022, respectively. For the three and six months ended June 30, 2023, the decrease was mainly attributed to increases in direct employee wages, taxes and benefit costs of 2.9% and 3.2% respectively.

The hospice segment's general and administrative expenses primarily consist of administrative employee wages, taxes and benefit costs, rent, information technology and office expenses. General and administrative expenses, expressed as a percentage of net service revenues was 25.4% and 25.0% for the three months ended June 30, 2023 and 2022, respectively, and 26.0% and 24.8% for the six months ended June 30, 2023 and 2022, respectively. General and administrative expenses as a percentage of net service revenues for the three months ended June 30, 2023 was relatively consistent with the prior period. The increase in general and administrative expenses as a percentage of net service revenues for the six months ended June 30, 2023 was due to a \$1.3 million increase in administrative employee wages, taxes and benefit costs.

# Home Health Segment

		For the	Three Months	Ended June 30,		For the Six Months Ended June 30,									
	2	023	20	22	Chan	ige	20	23	20	122	Chang	je			
	Amount	% of Segment Net Service Revenues	Amount	% of Segment Net Service Revenues	Amount	_%	Amount	% of Segment Net Service Revenues	Amount	% of Segment Net Service Revenues	Amount	%			
		(Amounts	in Thousands,	Except Percentag	ges)			(Amounts in	1 Thousands	, Except Percentag	ges)				
Operating Results															
Net service revenues	\$ 11,456	100.0 %		100.0 %		8.7 % \$		100.0 % \$	19,811	100.0 % 5		20.8 %			
Cost of services revenues	7,084	61.8	7,138	67.7	(54)	-0.8	14,618	61.1	13,854	69.9	764	5.5			
Gross profit	4,372	38.2	3,398	32.3	974	28.7	9,323	38.9	5,957	30.1	3,366	56.5			
General and administrative expenses	2,641	23.1	2,501	23.7	140	5.6	5,521	23.1	4,860	24.5	661	13.6			
Segment operating income	\$ 1,731	15.1 %	\$ 897	8.5 %	\$ 834	93.0 % \$	3,803	15.9 % \$	1,097	5.5 % 5	\$ 2,706	246.6 %			
Business Metrics (Actual Numbers)															
Locations at period end							13		12						
New admissions * (1)	3,439		3,351		88	2.6 %	7,332		6,687		645	9.6 %			
Recertifications * (2)	1,595		1,409		186	13.2	3,144		2,725		419	15.4			
Total volume * (3)	5,034		4,760		274	5.8	10,476		9,412		1,064	11.3			
Visits * (4)	68,293		68,452		(159)	-0.2 %	146,121		133,665		12,456	9.3 %			
Organic growth															
- Revenue * (5)	(10.9) %	ó	24.6 %				0.7 %	)	12.6 %	ó					
- Admissions * (5)	(17.5) %	ó	25.2 %				(10.5) %	)	13.9 %	ó					

- (1) Represents new patients during the period.
- (2) A home health certification period is an episode of care that begins with a start of care visit and continues for 60 days. If at the end of the initial episode of care, the patient continues to require home health services, a recertification is required. This represents the number of recertifications during the period.
- (3) Total volume is total admissions and total recertifications in the period.

- (4) Represents number of services to patients in the period.
- (5) Revenue organic growth and admissions organic growth reflect the change in year-over-year revenue and admissions for the same store base. We define the same store base to include those stores open for at least 52 full weeks. These measures highlight the performance of existing stores, while excluding the impact of acquisitions, new store openings and closures.
- \* Management deems these metrics to be key performance indicators. Management uses these metrics to monitor our performance, both in our existing operations and acquisitions. Many of these metrics serve as the basis of reported revenues and assessment of these provide direct correlation to the results of operations from period to period and facilitate comparison with the results of our peers. Historical trends established in these metrics can be used to evaluate current operating results, identify trends affecting our business, determine the allocation of resources and assess the quality and potential variability of our cash flows and earnings. We believe they are useful to investors in evaluating and understanding our business but should not be used solely in assessing the Company's performance. These key performance indicators should not be considered superior to, as a substitute for or as an alternative to, and should be considered in conjunction with, the GAAP financial measures presented herein to fully evaluate and understand the business as a whole. These measures may not be comparable to similarly titled performance indicators used by other companies.

The home health segment generates net service revenues by providing home health services on a short-term, intermittent or episodic basis to individuals, generally to treat an illness or injury. Net service revenues from Medicare accounted for 76.1% and 72.1%, managed care organizations accounted for 19.6% and 21.5% and other accounted for 4.3% and 6.4% of total home health segment net service revenues for the three months ended June 30, 2023 and 2022, respectively. Net service revenues from Medicare accounted for 75.1% and 72.7%, managed care organizations accounted for 20.0% and 21.0% and other accounted for 4.9% and 6.3% of total home health segment net service revenues for the six months ended June 30, 2023 and 2022, respectively. Home health services provided to Medicare beneficiaries are paid under the Medicare Home Health Prospective Payment System, which uses national, standardized 30-day period payment rates for periods of care. CMS uses the PDGM as the case-mix classification model to place periods of care into payment categories, classifying patients based on clinical characteristics. An outlier adjustment may be paid for periods of care in which costs exceed a specific threshold amount.

Home Health net service revenues increased by \$0.9 million and \$4.1 million for the three and six months ended June 30, 2023, respectively, compared to the three and six months ended June 30, 2022. Total visits increased for the three months and six months ended June 30, 2023, mainly attributed to the acquisition of Apple Home HealthCare on October 1, 2022.

Gross profit, expressed as a percentage of net service revenues, was 38.2% and 32.3% for the three months ended June 30, 2023 and 2022, respectively, and 38.9% and 30.1%, for the six months ended June 30, 2023 and 2022, respectively. For the three and six months ended June 30, 2023, the increase was primarily due to an decrease in direct employee wages, taxes and benefit costs of 13.5% and 13.7%, respectively.

The home health segment's general and administrative expenses primarily consist of administrative employee wages, taxes and benefit costs, rent, information technology and office expenses. General and administrative expenses, expressed as a percentage of net service revenues, was 23.1% and 23.7% for the three months ended June 30, 2023 and 2022, respectively, and 23.1% and 24.5% for the six months ended June 30, 2023 and 2022, respectively. The decrease in general and administrative expenses as a percentage of net service revenues was primarily due to acquisitions that resulted in more efficient administrative of employee wages, taxes and benefit costs for the three and six months ended June 30, 2023, respectively.

# **Liquidity and Capital Resources**

# Overview

Our primary sources of liquidity are cash on hand and cash from operations and borrowings under our credit facility. At June 30, 2023 and December 31, 2022, we had cash balances of \$84.1 million and \$80.0 million, respectively. At June 30, 2023, we had a \$600.0 million revolving credit facility and a \$125.0 million incremental loan facility, which may be for term loans or an increase to the revolving loan commitments. The maturity of this credit facility is July 30, 2026.

During the six months ended June 30, 2023, we used approximately \$0.9 million in cash to fund the CareStaff acquisition and repaid \$53.5 million under our revolving credit facility. As of June 30, 2023, we had a total of \$81.4 million in revolving loans, with an interest rate of 6.95% outstanding on our credit facility and after giving effect to the amount drawn on our credit facility, approximately \$8.0 million of outstanding letters of credit and borrowing limits based on an advance multiple of adjusted EBITDA (as defined in the Credit Agreement), we had \$409.3 million of capacity and \$319.9 million available for borrowing under our credit facility. At December 31, 2022, we had a total of \$134.9 million revolving credit loans, with an interest rate of 6.13%, outstanding on our credit facility.

Our credit facility requires us to maintain a total net leverage ratio not exceeding 3.75:1.00. At June 30, 2023, we were in compliance with our financial covenants under the Credit Agreement. Although we believe our liquidity position remains strong, we can provide no assurance that we will remain in compliance with the covenants in our Credit Agreement, and in the future, it may prove necessary to seek an amendment with the bank lending group under our credit facility. Additionally, there can be no assurance that we will be able to raise additional funds on terms acceptable to us, if at all.

See Note 7 to the Notes to Condensed Consolidated Financial Statements, Long-Term Debt, for additional details of our long-term debt.

# **COVID-19 Pandemic**

The federal public health emergency declared by HHS as a result of the COVID-19 pandemic expired May 11, 2023, reflecting the evolution of the COVID-19 public health situation from its acute emergency phase. Earlier in the pandemic, federal and state governments passed legislation, promulgated regulations, and took other administrative actions intended to assist healthcare providers in providing care to COVID-19 patients and other patients during the public health emergency. These temporary measures, most of which have been reduced or terminated, included relief from Medicare conditions of participation requirements for healthcare providers, relaxation of licensure requirements for healthcare professionals, relaxation of privacy restrictions for telehealth remote communications, promoting use of telehealth by expanding the scope of services for which Medicare reimbursement is available, and limited waivers of fraud and abuse laws for activities related to COVID-19 during the emergency period.

# ARPA Spending Plans

The ARPA, which became law on March 11, 2021, provided for \$350 billion in relief funding for eligible state, local, territorial and tribal governments to mitigate the fiscal effects of the COVID-19 public health emergency. Additionally, the law provided for a 10 percentage point increase in federal matching funds for Medicaid HCBS from April 1, 2021, through March 31, 2022, provided the state satisfied certain conditions. States are permitted to use the state funds equivalent to the additional federal funds through March 31, 2025. States must use the monies attributable to this matching fund increase to supplement, not supplant, their level of state spending for the implementation of activities enhanced under the Medicaid HCBS in effect as of April 1, 2021.

HCBS spending plans for the additional matching funds vary by state, but common initiatives in which the Company is participating include those aimed at strengthening the provider workforce (e.g., efforts to recruit, retain, and train direct service providers). The Company is required to properly and fully document the use of such funds in reports to the state in which the funds originated. Funds may be subject to recoupment if not expended or if they are expended on non-approved uses. During the six months ended June 30, 2023, the Company received state funding provided by the ARPA in an aggregate amount of \$1.9 million. The Company recorded revenue of \$0.2 million for certain states that met the revenue recognition criteria. The Company deferred the remaining \$1.7 million, which was received from states with specific spending plans and reporting requirements. The Company utilized \$2.4 million and \$4.8 million of these funds during the three and six months ended June 30, 2023, primarily for caregivers and adding support to recruiting and retention efforts, included as a reduction of cost of service revenues in the Company's Consolidated Statements of Income. As of June 30, 2023, the deferred portion of ARPA funding of \$10.0 million is included within Government stimulus advances on the Company's Unaudited Condensed Consolidated Balance Sheets.

# Medicare sequester

The CARES Act and related legislation temporarily lifted the Medicare sequester that would have otherwise reduced payments to Medicare providers by 2% as required by the Budget Control Act of 2011, from May 1, 2020 through March 31, 2022. The sequestration payment adjustment was phased back in, returning to a 2% reduction on July 1, 2022. These sequestration cuts have been extended through the first six months of 2032.

In our hospice segment, Medicare sequester relief resulted in an increase in net service revenues of \$0.0 million and \$0.5 million for the three months ended June 30, 2023 and 2022, respectively, and \$0.0 million and \$1.4 million for the six months ended June 30, 2023 and 2022, respectively. In our home health segment, Medicare sequester relief resulted in an increase in net service revenues of \$0.0 million and \$0.1 million for the three months ended June 30, 2023 and 2022, respectively and \$0.0 and \$0.3 million for the six months ended June 30, 2023 and 2022, respectively.

The ARPA increases the federal budget deficit in a manner that triggers an additional statutorily mandated sequestration under the PAYGO Act. As a result, an additional Medicare payment reduction of up to 4% was required to take effect in January 2022. However, Congress has delayed implementation of this payment reduction until 2025. We cannot currently determine if, or to what extent, our business, results of operations, financial condition or liquidity will ultimately be impacted by mandated sequestration triggers under the PAYGO Act, or if or when the mandated sequestration will occur.

See Note 6 to the Notes to Condensed Consolidated Financial Statements, COVID-19 Pandemic, for additional details of the COVID-19 pandemic.

### Cash Flows

The following table summarizes changes in our cash flows:

	 For the Six Months Ended June 30,						
	2023		2022				
	(Amounts in Thousands)						
Net cash provided by operating activities	\$ 60,413	\$	62,502				
Net cash used in investing activities	(2,711)		(86,028)				
Net cash used in financing activities	(53,475)		(24,452)				

Six Months Ended June 30, 2023 Compared to Six Months Ended June 30, 2022

Cash flows from operating activities represent the inflow of cash from our payors and the outflow of cash for payroll and payroll taxes, operating expenses, interest and taxes. Net cash provided by operating activities was \$60.4 million for the six months ended June 30, 2023, compared to net cash provided by operating activities of \$62.5 million for the same period in 2022. The decrease in cash provided by operations was primarily due to the timing of receipts on accounts receivable and the timing of government stimulus funds. The changes in accounts receivable were primarily related to the growth in revenue and a decrease in days sales outstanding ("DSO") during the six months ended June 30, 2023 compared to 2022, as described below. The related receivables due from the Illinois Department on Aging represented 13.0% and 18.0% of the Company's net accounts receivable at June 30, 2023 and June 30, 2022, respectively, as discussed below.

Net cash used in investing activities for the six months ended June 30, 2023, primarily consisted of \$0.9 million of net cash used for the CareStaff acquisition and \$1.8 million for property and equipment purchases, which were primarily related to our ongoing investments in our technology infrastructure. Net cash used in investing activities for the six months ended June 30, 2022, primarily consisted of \$84.5 million of net cash used for the JourneyCare acquisition and \$1.5 million of cash used for the purchase of property and equipment.

Net cash used in financing activities for the six months ended June 30, 2023, primarily related to a \$53.5 million payment on the revolver portion of our credit facility. Net cash used in financing activities for the six months ended June 30, 2022, primarily related to a \$60.0 million payment on the revolver portion of our credit facility, partially offset by borrowings of \$35.0 million on the revolver portion of our credit facility to fund, in part, the JourneyCare acquisition. For the six months ended June 30, 2023 and 2022, net cash provided by financing activities included cash received from the exercise of stock options of \$0.25 million and \$0.5 million, respectively.

# **Outstanding Accounts Receivable**

Gross accounts receivable as of June 30, 2023 and December 31, 2022 were approximately \$105.7 million and \$127.1 million, respectively. Outstanding accounts receivable, net of allowance for credit losses, decreased by \$21.3 million as of June 30, 2023 as compared to December 31, 2022. Accounts receivable for the Illinois Department on Aging decreased approximately \$9.1 million during the six months ended June 30, 2023. Our collection procedures include review of account aging and direct contact with our payors. We have historically not used collection agencies. An uncollectible amount is written off to the allowance account after reasonable collection efforts have been exhausted.

We calculate our DSO by taking the trade accounts receivable outstanding, net of allowance for credit losses for doubtful accounts, divided by the net service revenues for the last quarter, multiplied by the number of days in that quarter. Our DSOs were 36 days and 45 days at June 30, 2023 and December 31, 2022, respectively. The DSOs for our largest payor, the Illinois Department on Aging, were 22 days and 42 days at June 30, 2023 and December 31, 2022, respectively.

# **Off-Balance Sheet Arrangements**

As of June 30, 2023, we did not have any off-balance sheet guarantees or arrangements with unconsolidated entities.

# **Critical Accounting Policies and Estimates**

There have been no material changes to our critical accounting policies and estimates previously disclosed under the caption "Management's Discussion and Analysis of Financial Condition and Results of Operations — Critical Accounting Policies and Estimates" set forth in Part II, Item 7 of our Annual Report on Form 10-K for the period ended December 31, 2022, filed on February 28, 2023.

# **Recently Issued Accounting Pronouncements**

Refer to Note 2 to the Notes to Condensed Consolidated Financial Statements (Unaudited) for further discussion.

### ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

We are exposed to market risk associated with changes in interest rates on our variable rate long-term debt. As of June 30, 2023, we had outstanding borrowings of approximately \$81.4 million on our credit facility, all of such borrowings were subject to variable interest rates. If the variable rates on this debt were 100 basis points higher than the rate applicable to the borrowing during the three and six month periods ended June 30, 2023, our net income would have decreased by \$0.2 million, or \$0.01 per diluted share, and \$0.6 million, or \$0.04 per diluted share, respectively. We do not currently have any derivative or hedging arrangements, or other known exposures, to changes in interest rates.

### ITEM 4. CONTROLS AND PROCEDURES

# **Evaluation of Disclosure Controls and Procedures**

Our management, with the participation of our Chief Executive Officer and our Chief Financial Officer, evaluated the effectiveness of our disclosure controls and procedures as of June 30, 2023. The term "disclosure controls and procedures," as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended (the "Exchange Act") means controls and other procedures of an issuer that are designed to ensure that information required to be disclosed by an issuer in the reports that it files or submits under the Exchange Act, is recorded, processed, summarized, and reported within the time periods specified in the Securities and Exchange Commission's rules and forms. Disclosure controls and procedures include, without limitation, controls and procedures designed to ensure that information required to be disclosed by an issuer in the reports that it files or submits under the Exchange Act is accumulated and communicated to the issuer's management, including its principal executive and principal financial officers, or persons performing similar functions, as appropriate to allow timely decisions regarding required disclosure. Management recognizes that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving their objectives and management necessarily applies its judgment in evaluating the cost-benefit relationship of possible controls and procedures.

Based on the evaluation of our disclosure controls and procedures, our Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures were effective as of June 30, 2023.

# Changes in Internal Control Over Financial Reporting

There were no changes in our internal control over financial reporting identified in connection with the evaluation required by Rule 13a-15(d) and 15d-15(d) of the Exchange Act that occurred during the fiscal quarter ended June 30, 2023 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

### PART II - OTHER INFORMATION

### Item 1. Legal Proceedings

# Legal Proceedings

From time to time, we are subject to legal and/or administrative proceedings incidental to our business. It is the opinion of management that the outcome of pending legal and/or administrative proceedings will not have a material effect on our financial position and results of operations.

Further information with respect to this Item may be found in Note 9 to the Condensed Consolidated Financial Statements in Part I, Item 1—"Financial Statements (Unaudited)," which is incorporated herein by reference.

## Item 1A. Risk Factors

Investing in our common stock involves a high degree of risk. You should carefully consider the risk factors discussed under the caption "Risk Factors" set forth in Part I, Item 1A, of our Annual Report on Form 10-K for the year ended December 31, 2022, filed on February 28, 2023. There have been no material changes to the risk factors previously disclosed under the caption "Risk Factors" in our Annual Report on Form 10-K. Additional risks and uncertainties not currently known to us or that we currently deem to be immaterial also may materially adversely affect our business, financial condition or operating results.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

None.

Item 3. Defaults Upon Senior Securities

None.

Item 4. Mine Safety Disclosures

None.

#### Item 5. Other Information

Not applicable. Without limiting the generality of the foregoing, during the quarter ended June 30, 2023, no director or Section 16 officer adopted or terminated any Rule 10b5-1 trading arrangements, as defined in Item 408(a) of Regulation S-K.

## **Table of Contents**

#### **Exhibits** Item 6.

## EXHIBIT INDEX

			Incorporate	d by Reference	
Exhibit Number	Description of Document	Form	File No.	Date Filing	Exhibit Number
<u>3.1</u>	Amended and Restated Certificate of Incorporation of the Company dated as of October 27, 2009.	10-Q	001-34504	11/20/2009	3.1
3.2	Amended and Restated Bylaws of the Company, as amended by the First Amendment to the Amended and Restated Bylaws.	10-Q	001-34504	05/9/2013	3.2
<u>4.1</u>	Form of Common Stock Certificate.	S-1	333-160634	10/2/2009	4.1
<u>10.1</u>	Addus HomeCare Corporation Amended and Restated 2017 Omnibus Incentive Plan.	8-K	001-34504	06/15/2023	10.1
10.2**	Membership Interests Purchase Agreement, dated June 28, 2023, by and among Addus HealthCare, Inc., HHH Newco Holdings, LLC, American Health Companies, LLC, American Home Care, LLC, Homecare, LLC, Tennessee Valley Home Care, LLC, and Tri-County Home Health and Hospice, LLC.				
<u>31.1</u>	Certification of Chief Executive Officer Pursuant to Rule 13a-14(a) of the Securities Exchange Act of 1934 as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.				
<u>31.2</u>	Certification of Chief Financial Officer Pursuant to Rule 13a-14(a) of the Securities Exchange Act of 1934 as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.				
32.1	Certification of Chief Executive Officer Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.				
32.2	Certification of Chief Financial Officer Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.				
101.INS	Inline XBRL Instance Document (the instance document does not appear in the Interactive Data File because its XBRL tags are embedded within the Inline XBRL document).				
101.SCH	Inline XBRL Taxonomy Extension Schema Document.				
101.CAL	Inline XBRL Taxonomy Calculation Linkbase Document.				
101.LAB	Inline XBRL Taxonomy Label Linkbase Document.				
101.PRE	Inline XBRL Presentation Linkbase Document.				
101.DEF	Inline XBRL Taxonomy Extension Definition Linkbase Document.				
104	Cover Page Interactive Data File (embedded within the Inline XBRL document and contained in Exhibit 101).				

<sup>\*</sup> Management compensatory plan or arrangement

\*\* Schedules and exhibits have been omitted pursuant to Item 601 of Regulation S-K. The Company hereby undertakes to furnish supplementally a copy of any of the omitted schedules and exhibits upon request by the Securities and Exchange Commission.

## **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

	ADDUS HOME	ADDUS HOMECARE CORPORATION		
Date: August 1, 2023	Ву:	/s/ R. DIRK ALLISON		
		R. Dirk Allison Chairman and Chief Executive Officer (As Principal Executive Officer)		
Date: August 1, 2023	Ву:	/s/ BRIAN POFF		
		Brian Poff Chief Financial Officer (As Principal Financial Officer)		
	38			

# **Membership Interests Purchase Agreement**

by

Addus HealthCare, Inc. ("Buyer"),
American Health Companies, LLC ("Seller Parent"),

HHH Newco Holdings, LLC ("Seller"),

and

each Acquired Company

(as defined herein)

**Dated** 

June 28, 2023

## TABLE OF CONTENTS

1.

2.

3.

4.

DEFIN 1.1	Definitions 1
1.2	Usage. 14
	AND TRANSFER OF MEMBERSHIP INTERESTS; CLOSING 16
2.1	Membership Interests 16
2.2	Purchase Price 16
2.3	Closing Purchase Price Adjustments 16
2.4	Post-Closing Purchase Price Adjustment 17
2.5	Closing 19
2.6	Closing Obligations 19
2.7	Withholding 22
REPR	ESENTATIONS AND WARRANTIES OF SELLER PARENT AND SELLER 22
3.1	Corporate Organization 22
3.2	Capitalization/ Ownership of Membership Interests 22
3.3	Authority; No Violation 23
3.4	Consents and Approvals 24
3.5	Financial Statements 24
3.6	Brokers or Finders 25
3.7	Absence of Certain Changes or Events 25
3.8	Condition and Sufficiency of Assets 27
3.9	Inventory 27
3.10	Legal Proceedings 27
3.11	Taxes and Tax Returns 28
3.12	Employees 30
3.13	Employee Benefits 31
3.14	Compliance with Applicable Law 33
3.15	Certain Contracts 34
3.16	Real Estate 36
3.17	Undisclosed Liabilities; Indebtedness 37
3.18	Insurance 37
3.19	Intellectual Property 38
3.20	Transactions with Related Persons 39
3.21	Environmental Liability 39
3.22	Healthcare Law Matters 40
3.23	Directors and Officers 44
3.24	Material Contracted Care Partners, Suppliers and Payors 44
3.25	No Further Representations 45
DEDDI	ECENTATIONS AND MADDANTIES OF DIVED. 45
4.1	ESENTATIONS AND WARRANTIES OF BUYER 45 Organization and Good Standing 46
4.1 4.2	Enforceability and Authority; No Conflict 46
4.4	i
	1

	4.3	Investment Intent 46
	4.4	Certain Proceedings 46
	4.5	Brokers or Finders 47
	4.6	Due Diligence 47
	4.7	Non-reliance of Buyer 47
	4.8	Sufficiency of Funds 47
	4.9	R&W Binder 47
5.	COVI	ENANTS OF SELLER PRIOR TO CLOSING DATE 48
	5.1	Access and Investigation 48
	5.2	Operation of the Businesses of the Acquired Companies 48
	5.3	Filings and Notifications; Cooperation 48
	5.4	Exclusive Dealing 49
	5.5	Financial Information 49
	5.6	401(k) Plan 49
5.	COVI	ENANTS OF BUYER PRIOR TO CLOSING DATE 49
	6.1	Filings and Notifications; Cooperation 49
	6.2	R&W Binder 50
7.	POST	-CLOSING COVENANTS 51
	7.1	Cooperation and Proceedings; Access to Records 51
	7.2	Insurance 51
	7.3	Confidentiality 51
	7.4	Exculpation, Indemnification, D&O Insurance 52
	7.5	Public Announcements 53
	7.6	COBRA Liability 53
3.	CONI	DITIONS PRECEDENT TO BUYER'S OBLIGATION TO CLOSE 54
	8.1	Accuracy of Seller Parent's and Seller's Representations 54
	8.2	Seller Parent's and Seller's Performance 54
	8.3	Bring Down Certificate 54
	8.4	Consents 54
	8.5	Governmental Authorizations 54
	8.6	No Legal Prohibition 54
	8.7	Closing Deliveries 54
	8.8	Frustration of Closing Conditions 55
<del>)</del> .	CONI	DITIONS PRECEDENT TO SELLER PARENT'S AND SELLER'S OBLIGATIONS TO CLOSE
	55	
	9.1	Accuracy of Buyer's Representations 55
	9.2	Buyer's Performance 55
	9.3	Bring Down Certificate 55
	9.4	Consents 55
	9.5	No Legal Prohibition 55
	9.6	Closing Deliveries 55
	9.7	Frustration of Closing Conditions 55

10.	TERMIN	TERMINATION 56			
	10.1	Termination Events 56			
	10.2	Effect of Termination 56			
11.	SURVIV	AL; REMEDIES 56			
	11.1	Survival of Representations, Warranties and Covenants 56			
	11.2	Remedies; Specific Performance. 57			
	11.3	Exclusive Remedy 58			
12.	TAX MA	TTERS 58			
	12.1	General 58			
13.	MISCEL	LANEOUS 60			
	13.1	Expenses 60			
	13.2	Seller's Disclosure Statement 60			
	13.3	Further Assurance 60			
	13.4	Entire Agreement 61			
	13.5	Modification 61			
	13.6	Assignments and Successors 61			
	13.7	No Third Party Rights 61			
	13.8	Governing Law 61			
	13.9	Jurisdiction; Service of Process 61			
	13.10	Waiver of Jury Trail 62			
	13.11	Attorneys' Fees 62			
	13.12	Enforcement of Agreement 62			
	13.13	No Waiver 62			
	13.14	Notices 62			
	13.15	Post-Closing Representation of Seller 63			
	13.16	Attorney Client Privilege Carve Out 64			
	13.17	Severability 64			
	13.18	Time of Essence 64			
	13.19	Counterparts and Electronic Signature 64			

#### MEMBERSHIP INTERESTS PURCHASE AGREEMENT

This Membership Interests Purchase Agreement ("**Agreement**") is made as of June 28, 2023 by Addus HealthCare, Inc., an Illinois corporation ("**Buyer**"), American Health Companies, LLC, a Tennessee limited liability company ("**Seller Parent**"), HHH Newco Holding, LLC, a Michigan limited liability company ("**Seller**"), and each of the Acquired Companies (defined below). Buyer, Seller Parent, Seller, and each Acquired Company are sometimes referred to herein, collectively, as the "**Parties**" and, individually, as a "**Party**."

#### **RECITALS**

WHEREAS, Seller Parent owns directly all of the issued and outstanding membership interests in the Company (the "**Membership Interests**");

WHEREAS, the Company provides home health and hospice services in Tennessee through one or more direct or indirect Subsidiaries (the "**Business**");

WHEREAS, prior to the Closing and pursuant to the Pre-Closing Reorganization, Seller Parent shall contribute the Membership Interests of the Company to Seller such that at Closing, Seller shall own directly all of the Membership Interests of the Company; and

WHEREAS, Seller and Seller Parent desire to sell, and Buyer desires to purchase from Seller the Business via an acquisition of all of the Membership Interests.

The Parties, intending to be legally bound, agree as follows:

#### 1. DEFINITIONS AND USAGE

#### 1.1Definitions

For purposes of this Agreement, the following terms have the meanings specified or referred to in this Section 1.1:

"401(k) Plan"—as defined in Section 5.6.

"Acquired Companies"—means the Company and its subsidiaries, Homecare, LLC, a Tennessee limited liability company; Tennessee Valley Home Care, LLC (d/b/a Tennessee Quality Care – Home Health), a Tennessee limited liability company; and Tri-County Home Health and Hospice, LLC (d/b/a Tennessee Quality Care - Hospice), a Tennessee limited liability company. "Acquired Company" means any one of the Acquired Companies.

**"Adjusted Closing Cash Amount"** – a net amount equal to (A) the Gross Purchase Price, <u>minus</u> (B) the Final Closing Debt Amount, <u>minus</u> (C) the Final Seller's Expenses Amount, <u>plus</u> (D) the Final Cash on Hand Amount, <u>plus</u> (E) the excess, if any, of the amount of Final Net Working Capital Amount over the amount of the Net Working Capital Target, <u>minus</u> (F) the excess, if any, of the amount of the Net Working Capital Target over the amount of the Final Net Working Capital Amount.

- "Affiliate" —of any Person means any Person directly or indirectly controlling, controlled by or under common control with such Person. The term "control" (including the terms "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of at least ten percent (10%) of the voting securities of such other Person, by contract or otherwise.
  - "**Agreement**"—as defined in the first paragraph of this Agreement.
- "ARP Funds" means all stimulus funds received by any Acquired Company prior to the Closing pursuant to the American Rescue Plan Act.
- "Business Day"—any day other than (a) Saturday or Sunday or (b) any other day on which national banks in Nashville, Tennessee, are generally permitted or required to be closed.
  - "Buyer"—as defined in the first paragraph of this Agreement.
  - "Buyer Prepared Returns"—as defined in Section 12.1(a).
- **"Buyer's Closing Documents"**—the documents required to be executed or delivered by Buyer at the Closing pursuant to <u>Section 2.6(b)</u>.
- **"Cash on Hand"** all cash and cash equivalents (including marketable securities and short-term investments) held by the Acquired Companies as of the measurement date, including the amounts of any received but uncleared checks, drafts and wires received by the Acquired Companies or others on their behalf prior to such time, less the amount of any Restricted Cash.
  - "Closing"—as defined in Section 2.3.
  - "Closing Balance Sheet"—as defined in Section 2.4(a).
  - "Closing Cash Amount"—as defined in Section 2.2(b).
  - "Closing Date"—the date on which the Closing occurs.
  - "Closing Documents"—as defined in <u>Section 11.1</u>.
  - "Closing Statement"— as defined in Section 2.4(a).
  - "Code"—the Internal Revenue Code of 1986, as amended.
  - "Company"— means American Home Care, LLC, a Tennessee limited liability company.
  - "Company Benefit Plan" as defined in Section 3.13(a).
  - "Company ERISA Affiliate" as defined in Section 3.13(a).

"Company Intellectual Property"—the Owned Intellectual Property, Intellectual Property subject to an IP Licenses and any other Intellectual Property used or held for use by the Acquired Companies or necessary for the operation of the business as currently conducted.

"Company Material Contract" —as defined in Section 3.15(a).

"Company Systems"—as defined in Section 3.19(c).

"Confidential Information"—as defined in Section 7.3(a).

"Consent"—any approval, consent, ratification, grant, waiver, exemption or other authorization.

"Contemplated Transactions"—the transactions contemplated by this Agreement.

"Contract"—any agreement, contract, lease, license or other legally binding instrument or obligation (whether written or oral).

"COVID-19 Assistance" — as defined in Section 3.17(b).

"**Data Privacy and Security Laws**" — all Legal Requirements governing, regulating, or protecting the privacy, security, use, disclosure, maintenance or transmission of Personal Information.

"**Debt"**— with respect to any Person without duplication, and solely to the extent not included in the calculation of Net Working Capital, (i) all obligations of such Person for borrowed money, whether current or long-term, secured or unsecured, (ii) all obligations of such Person created or arising under any conditional sale or other title retention agreement with respect to property acquired by such Person, (iii) all deferred obligations of such Person to make payment for all or part of the purchase price of property or services, whether accrued, absolute, contingent, unliquidated or otherwise, known or unknown, whether due or to become due (including the maximum potential amount payable under notes, earn-out payments, change in control payments, or similar payments), (iv) all obligations under leases which shall have been or should be, in accordance with GAAP, recorded as finance or synthetic leases in respect of which such Person is liable as lessee, (v) any obligation of such Person in respect of bankers' acceptances, performance bonds or letters of credit (whether drawn or undrawn), (vi) any obligations secured by Encumbrances on property acquired by such Person, whether or not such obligations were assumed by such Person at the time of acquisition of such property, (vii) obligations under any interest rate, currency or other hedging or swap agreement, (viii) the Pre-Closing Tax Accrual Amount, (ix) any obligation set forth on Section 1.1 of the Seller's Disclosure Statement, (x) all obligations of a type referred to in clauses (i) through (viii) above which are directly or indirectly guaranteed by such Person or which it has agreed (contingently or otherwise) to purchase or otherwise acquire or in respect of which it has otherwise assured a credit against loss, and (xi) interest, principal, prepayment penalty, fees, redemption costs, or expenses, to the extent due or owing in respect of those items listed in clauses (i) through (x) above, whether resulting from their payment or discharge or otherwise.

"Debt Payoff Letters"—as defined in Section 2.6(a)(vi).

**"Effective Time"** — 12:01 a.m. central time on the Closing Date.

**"Encumbrance"**—any charge, claim, community or other marital property interest, condition, equitable interest, lien, option, pledge, security interest, mortgage, right of way, easement, encroachment, servitude, right of first option, right of first refusal, or similar restriction, including any restriction on use, voting, transfer, receipt of income, or exercise of any other attribute of ownership.

"**End Date**"—as defined in <u>Section 10.1(d)</u>.

**"Environment"**—soil, land surface and subsurface strata, surface waters (including navigable and non-navigable inland and ocean waters), ground waters, drinking water supply, stream sediments, ambient air, indoor air, plant and animal life, and any other environmental medium or natural resource.

"Environmental Law"—any Legal Requirement that provides for or relates to: (i) the use of any Hazardous Material, the Release or Threat of Release of Hazardous Material, violation of discharge or emission limits or other prohibitions, or any Hazardous Activity or any activity, such as resource extraction or construction, that could have a significant effect on the Environment; (ii) preventing or reducing to acceptable levels the Release of Hazardous Material into the Environment; (iii) reducing the quantities, or minimizing or controlling the hazardous characteristics, of Hazardous Material that are generated; (iv) reducing the risks involved in the transportation of Hazardous Material; (v) the cleanup of Hazardous Material that has been Released, preventing its Release, or addressing the Threat of Release, or paying the costs of such actions; or (vi) making a Person compensate any other Person for damage done to its health or property or the Environment or permitting self-appointed representatives of the public interest to recover for injuries done to public assets or resources.

**"Equity Security"**—in respect of any Person, (a) any capital stock or similar security, (b) any security convertible into or exchangeable for any security described in clause (a), (c) any option, warrant, or other right to purchase or otherwise acquire any security described in clauses (a), (b), or (c), and (d) any "equity security" within the meaning of the Exchange Act.

"ERISA"—the Employee Retirement Income Security Act of 1974, as amended.

"Estimated Closing Balance Sheet"—as defined in Section 2.3(a).

"Estimated Closing Debt Amount"—as defined in Section 2.3(a).

"Estimated Closing Statement"—as defined in Section 2.3(a).

"Estimated Net Working Capital Amount"—as defined in Section 2.3(a).

"Estimated Seller's Expenses Amount"—as defined in Section 2.3(a).

"Exchange Act"—the Securities Exchange Act of 1934, as amended.

"Expense Payoff Letters"—as defined in Section 2.6(a)(vii).

- **"Final Cash on Hand Amount"** Cash on Hand of the Acquired Companies as of the Effective Time as finally determined pursuant to <u>Section 2.4</u>.
- **"Final Closing Debt Amount"** Debt of the Acquired Companies as of the Effective Time as finally determined pursuant to <u>Section 2.4</u>.
- **"Final Net Working Capital Amount"** Net Working Capital as of the Effective Time as finally determined pursuant to <u>Section 2.4</u>.
- "**Final Seller's Expenses Amount**"— Seller's Expenses as of the Effective Time as finally determined pursuant to Section 2.4.

## **"Financial Statements"** – as defined in <u>Section 3.5</u>.

- **"Fraud**"—actual (knowing, and not constructive) and intentional fraud with respect to the making of a representation or warranty set forth in <u>Article 3</u> or <u>Article 4</u>. For the avoidance of doubt, the term "Fraud" does not include any claim for equitable fraud, promissory fraud, unfair dealings fraud, or any other tort (including a claim for fraud) based solely on negligence, gross negligence or reckless conduct.
- "GAAP"—generally accepted accounting principles in the United States as in effect from time to time, as historically applied by the Acquired Companies.
- "Governmental Authorization"—any (a) Consent, license, registration, certificate, grant, waiver, exemption, provider or supplier number, or permit issued, granted, given, or otherwise made available by or under the authority of any Governmental Body or pursuant to any Legal Requirement; or (b) right under any Contract with any Governmental Body.

## "Governmental Body"—any:

- (a) nation, state, county, city, town, borough, village, district, or other jurisdiction;
- (b) federal, state, local, municipal, foreign, multinational, or other government;
- (c) governmental or quasi-governmental authority of any nature (including any agency, branch, department, board, commission, court, tribunal, or other entity exercising governmental or quasi-governmental powers);
- (d) body exercising, or entitled or purporting to exercise, any administrative, executive, judicial, legislative, police, regulatory, or taxing authority or power, whether local, national, or international; or
  - (e) official of any of the foregoing.

For the avoidance of doubt, and without limitation, "Governmental Body" includes Medicare administrative contractors, zone program integrity contractors, unified program integrity contractors, supplemental medical review contractors and the like.

"Government Program" means any United States federal, state or local health care or reimbursement program administered by a Governmental Body, including, without limitation, any "Federal Health Care Program" as defined in 42 U.S.C. §1320a-7b(f), including Medicare, state Medicaid programs, state CHIP programs, TRICARE and similar or successor programs with or for the benefit of, or sponsored, in whole or in part, by any Governmental Body.

"Gross Purchase Price" — as defined in Section 2.2(a).

"Hazardous Activity"— the distribution, generation, handling, importing, management, manufacturing, processing, production, refinement, Release, storage, transfer, transportation, treatment, or use of Hazardous Material and any other act, business, operation, or activity that increases the danger, or poses a risk of harm, to the Environment.

"Hazardous Material"— any substance, material, or waste that is or will foreseeably be regulated by any Governmental Body, including any material, substance, or waste that is defined or classified as a "hazardous waste," "hazardous material," "hazardous substance," "extremely hazardous waste," "pollutant," "restricted hazardous waste," "contaminant," "toxic waste," "pollutant," or "toxic substance" under any provision of Environmental Law, including petroleum, petroleum products, asbestos, presumed asbestos-containing material or asbestos-containing material, urea formaldehyde, per- and polyfluoroalkyl substances or polychlorinated biphenyls.

"Healthcare Laws"—means all Legal Requirements that govern, regulate, restrict or relate to hospice, home health, and private duty nursing, prescribing or dispensing medicines or controlled substances, healthcare referrals, billing and submission of healthcare claims, conditions of participation and conditions of payment in Government Programs, fraudulent, abusive or unlawful practices in connection with the provision of healthcare items or services or the billing for or claims for reimbursement for such items or services, coding, coverage, reimbursement, claims submission, billing and collections, insurance fraud, the administration of health care claims or benefits, processing or payment for health care services or treatment, healthcare claims processing, medical necessity, medical privacy and security, patient confidentiality, confidentiality of health records, patient inducements, patient referrals, anti-kickback, anti-referral, fee-splitting, false claims, healthcare advertising and marketing, medical waste requirements, professional conduct, informed consent, quality and safety, standards of care, credentialing, licensure, and certification, or other healthcare-related matters, including HIPAA and state Legal Requirements governing the confidentiality or security of individual health information, Title XVIII of the Social Security Act, 42 U.S.C. §§ 1395 et seq. (the Medicare statute), including the Ethics in Patient Referrals Act, as amended, or Stark Law; Title XIX of the Social Security Act, 42 U.S.C. §§ 1396 et seq. (the Medicaid statute), including, without limitation, Section 1128J(d) thereof; TRICARE, 10 U.S.C. § 1071 et seq.; the Federal Health Care Program Anti-Kickback Statute, 42 U.S.C. § 1320a 7b(b); the Beneficiary Inducement Statute, 42 U.S.C. § 1320a 7a(a)(5); the False Claims Act, as amended, 31 U.S.C. §§ 3729-3733; the Program Fraud Civil Remedies Act, 31 U.S.C. §§ 3801 et seq.; the Anti Kickback Act of 1986, 41 U.S.C. §§ 51 58; the Civil Monetary Penalties Law, 42 U.S.C. §§ 1320a 7a and 1320a 7b; the Exclusion Laws, 42 U.S.C. § 1320a 7; the Federal Controlled Substances Act, 21 U.S.C. § 801 et seq.; criminal false claim statutes, e.g., 18 U.S.C. § 1001, 42 U.S.C. § 1320a 7b(a); the Deficit Reduction Act of 2005, P.L. 109 171, 120 Stat. 4; the Affordable Care Act, P.L. 111 148 and 111 152; the CARES Act; state corporate practice of medicine Legal Requirements, all applicable licensing, survey, accreditation, and certificate of

need Legal Requirements; all applicable regulations, rules, guidance, policies, ordinances and orders promulgated thereunder; and any similar state and local statutes, regulations, rules, guidance, policies, ordinances, orders or other Legal Requirements that address the subject matter of the foregoing.

"Healthcare Permits"—as defined in Section 3.22(b).

**"HIPAA"**—the Health Insurance Portability and Accountability Act of 1996, as amended by the Health Information Technology for Economic and Clinical Health Act, and their implementing regulations set forth at 45 C.F.R. Parts 160, 162 and 164.

"HSR Act"—the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Intellectual Property" — any and all rights in, arising out of, or associated with any of the following in any jurisdiction throughout the world: (a) all patent applications, and all patents issuing thereon, and all inventions and improvements disclosed therein (b) trademarks, service marks, trade names, trade dress, corporate names, and other indicia of source, including all registrations, applications, and renewals in connection therewith, and all goodwill associated with the foregoing, (c) copyrights and works of authorship (whether or not copyrightable), rights of publicity, and all applications, registrations, and renewals in connection therewith, (d) domain names, social media accounts or user names (including "handles") (whether or not they are trademarks), phone numbers, registrations for any of the foregoing, (e) trade secrets, know-how, databases, processes, techniques, protocols and confidential information, (f) any and all other intellectual property rights and/or proprietary rights in any form or medium known or later devised, and (g) all copies and tangible embodiments, goodwill, rights of priority and protection of interests therein, and rights to recover for past, present and future infringement associated with any of the foregoing.

"Interim Balance Sheet Date" — as defined in <u>Section 3.5</u>.

"Interim Financial Statements" — as defined in <u>Section 3.5</u>.

"IP Licenses" — collectively (1) all Contracts to which any member of the Acquired Companies is a party and pursuant to which any member of Acquired Companies is authorized to use any third-party Intellectual Property or pursuant to which any Person is authorized to use, exploit, or is granted a license or any other rights to any Owned Intellectual Property, and (2) any Contracts pursuant to which any member of the Acquired Companies has agreed to any transfer of any Owned Intellectual Property by any member of the Acquired Companies or restriction of that member of the Acquired Companies' right to use or enforce any Owned Intellectual Property.

"**IRCA**" — as defined in <u>Section 3.12(d)</u>.

"**IRS**"—the United States Internal Revenue Service or any successor agency, and, to the extent relevant, the United States Department of the Treasury.

"Key Employee"— Nichole McClain and Kelly Nichols.

**"Knowledge"**— A Person will be deemed to have Knowledge of a particular fact or other matter if that Person has actual knowledge of that fact or matter, or would reasonably be expected to be aware of that fact or matter following a reasonable inquiry with that Person's direct reports concerning the factor or matter at issue.

**"Knowledge of Seller"** or **"Knowledge of the Company"**—Knowledge of Michael Bailey, Jeff Bogle, Robin Bradley or Philip Clark or the Knowledge of Laura Carrico, Barry Shermer, Jonathan Cooper, Nichole McClain and Kelly Nichols solely with respect to the subject matters reasonably associated with their job responsibilities.

"Lease"—as defined in Section 3.16(a).

"Leased Real Property" —as defined in Section 3.16(b).

**"Legal Requirement"**—any constitution, law, ordinance, principle of common law, code, rule, regulation, statute, act, treaty, directive, ordinance, decree or order of general applicability or other legally enforceable requirement, including Data Privacy and Security Laws, including as set forth in guidance, manuals, and policy, of any Governmental Body, including rules and regulations promulgated thereunder.

"Lookback Period"—six (6) years prior to the date hereof.

"Loss"—any cost, loss, liability, obligation, claim, cause of action, damage, deficiency, expense (including reasonable costs of investigation and defense and reasonable attorneys' fees and expenses), Tax, fine, penalty, judgment, award or assessment.

"Management Agreements"—means that certain Management Agreement, dated effective as of January 1, 2021, by and between Tri-Country Home Health and Hospice, LLC and American Health Partners Management, LLC and that certain Management Agreement, dated effective as of January 1, 2021, by and between Tennessee Valley Home Care, LLC and American Health Partners Management, LLC.

"Material Adverse Effect"—with respect to an Acquired Company, shall mean any change, effect, event, circumstance or development that, individually or when taken together with all other such similar or related changes, effects, events, circumstances or developments (x) has had, or would reasonably be expected to have, a material adverse effect on the business, assets, liabilities, condition (financial or otherwise) or results of operations of the Acquired Companies, taken as a whole, or (y) has, or would reasonably be expected to have, a material adverse effect on the ability of Seller, and/or an Acquired Company to consummate the transactions contemplated hereby, excluding in each case the impact of any changes, effects, events, circumstances or developments arising from: (i) general economic, capital or financial markets or industry conditions (including changes in interest rates or reimbursement rates or bank or other financial institution failures and responses of Governmental Bodies to such events); (ii) acts of God, natural disasters, calamities, national or international political or social conditions, including the engagement by any country in hostility (whether commenced before, on or after the date hereof, and whether or not pursuant to the declaration of a national emergency or war), or the occurrence of a military or terrorist attack; (iii) the impact of epidemics, pandemics and other similar health emergencies, including COVID-19, and responses of health officials and Governmental Bodies to

such events; (iv) changes in or reinterpretations of applicable Legal Requirements, regulatory, political, economic or business conditions, or GAAP (or, in each case, any interpretation thereof) after the date hereof; (v) actions that Seller can demonstrate resulted from Buyer unreasonably withholding, delaying or conditioning its consent with respect to any action requiring Buyer's consent hereunder; (vii) any publicly available statement made by Buyer or any of its Related Persons or Representatives concerning the Acquired Companies or otherwise relating to the Contemplated Transactions; (viii) the announcement, pendency or consummation of the Contemplated Transactions or (ix) any failure, in and of itself, by the Acquired Companies to meet any budgets, projections, forecasts or revenue or earnings predictions for any period ending on or after the date of this Agreement.

"Material Consents"—as defined in Section 8.4.

"Membership Interests"—as defined in the Recitals of this Agreement.

"Net Working Capital"— the current assets (excluding Cash on Hand, but including accounts receivable, prepaid expenses, and inventory) of the Acquired Companies minus the current liabilities (including accounts payable, accrued payroll benefits, and accrued expenses, but excluding intercompany balances, accrued interest, unearned revenue, and accrued taxes) of the Acquired Companies, in each case calculated in a manner consistent with the Acquired Companies' past practices in preparing its financial statements and the assumptions, principles and procedures set forth on <a href="Exhibit A">Exhibit A</a> of this Agreement, and in accordance with GAAP to the extent not inconsistent with the assumptions, principles and procedures set forth on <a href="Exhibit A">Exhibit A</a> sets forth an example, along with the assumptions, principles and procedures relating to the adjustments contained therein, of how Net Working Capital would have been calculated as of the end of the fiscal period of the Acquired Companies set forth therein. Notwithstanding the foregoing, Net Working Capital shall not include any amount or item included in the calculation of Debt.

"Net Working Capital Target" — means zero dollars (\$0.00).

"Objection Notice" — as defined in Section 2.4(b).

"Objection Notice Period" — as defined in Section 2.4(b).

"**Order**"—any order, writ, stipulation, consent, injunction, judgment, decree, ruling, assessment, determination or arbitration award of any Governmental Body or arbitrator.

"Ordinary Course of Business"—an action taken by a Person will be deemed to have been taken in the Ordinary Course of Business if that action is generally consistent with the past practices of such Person and is taken in the ordinary course of its operations.

"Organizational Documents"—(a) the articles or certificate of incorporation and the bylaws of a corporation; (b) the articles of organization and limited liability company agreement, operating agreement, or like agreement of a limited liability company; (c) the partnership agreement and any statement of partnership of a general partnership; (d) the limited partnership agreement and the certificate of limited partnership of a limited partnership; (e) any charter or

agreement or similar document adopted or filed in connection with the creation, formation, or organization of a Person; and (f) any amendment to or restatement of any of the foregoing.

"Owned Intellectual Property"—as defined in Section 3.19(a).

"Permitted Encumbrances"—(a) Encumbrances for Taxes and other governmental charges and assessments that are not yet due and payable; (b) Encumbrances of carriers, warehousemen, mechanics, and materialmen and other like Encumbrances arising in the Ordinary Course of Business (provided lien statements have not been filed or such Encumbrances otherwise perfected); and (c) statutory Encumbrances in favor of lessors arising in connection with any property leased to any Acquired Company.

"**Person**"—an individual, partnership, corporation, business trust, limited liability company, limited liability partnership, joint stock company, trust, unincorporated association, joint venture, other entity, or a Governmental Body.

"Personal Information" — any information processed, collected or otherwise used or disclosed by the Acquired Companies (or any third party on behalf of the Acquired Companies) that identifies a specific natural person, or when used in combination with other data elements is capable of identifying a specific natural person, including, without limitation: (a) a natural person's first and last name, in combination with a (i) social security number or tax identification number, or (ii) credit card number, bank account information and other financial account information, or financial customer or account numbers, account access codes and passwords; (b) Protected Health Information as defined under HIPAA, and (c) any information pertaining to an individual that is regulated or protected by one or more Legal Requirements, including any Data Privacy and Security Laws.

"Pre-Closing Income Tax Return"—as defined in Section 12.1(a).

"Pre-Closing Non-Income Tax Return"—as defined in Section 12.1(a).

"Pre-Closing Reorganization"—means (i) the contribution of the Membership Interests to Seller by Seller Parent, (ii) the distribution by Seller Parent of Seller to Seller Parent's indirect sole member, MFO AHP LLC, a Michigan limited liability company, and (iii) the distribution, assignment, or other transfer of each of Seller Parent, Seller, or certain of the Acquired Companies' interests in Nova Home Care, LLC to Michigan Quality Care Home Health, LLC, a Michigan limited liability company, or another affiliate of Mark T. Mitchell.

"**Pre-Closing Straddle Period**" means the portion of the Straddle Period that ends at the end of the day on the Closing Date.

**"Pre-Closing Tax Accrual Amount"** means the aggregate liability for unpaid Taxes of the Acquired Companies attributable to any Pre-Closing Tax Period (calculated, in the case of Straddle Period, in accordance with <u>Section 12.1(b)</u>), determined on a jurisdiction-by-jurisdiction basis with zero dollars (\$0) being the lowest amount for a jurisdiction.

"**Pre-Closing Tax Period**" means any taxable period ending on or before the Closing Date and the portion of any Straddle Period ending on and including the Closing Date.

"Privacy Consents"—as defined in Section 3.22(k).

"Proceeding"—any action, arbitration, mediation, audit, (including any additional development request and any audit or review conducted by any zone program integrity contractor, unified program integrity contractor, supplemental medical review contractor or similar contractor or agent conducted by or on behalf of a Governmental Body), hearing, investigation, complaint, charge, consent decree, appeal, adjustment, challenge, review, notice of violation, citation, summons subpoena, inquiry, inspection, litigation, suit, claim, proceeding, cause of action, demand of any nature (whether civil, criminal, administrative, judicial, or investigative) and whether at law or in equity.

"**Provider Relief Funds**"—the funds received by the Acquired Companies from the U.S. federal Department of Health and Human Services (the "<u>DHHS</u>") Provider Relief Fund under the CARES Act in the aggregate amount of \$1,112,777.25.

"**Provider Relief Fund Terms and Conditions**"—the terms and conditions established by the DHHS for the receipt of the Provider Relief Funds

"Purchase Price"—as defined in Section 2.2(a).

"Record"—information that is inscribed on a tangible medium or that is stored in an electronic or other medium.

## "Related Person"—

- (f) With respect to an individual:
  - (i) each other member of such individual's Family;
  - (ii) any Person that is directly or indirectly controlled by such individual; and
  - (iii) any Person with respect to which such individual serves as a director, officer, manager, executor, or trustee (or in a similar capacity).
- (g) With respect to a Person other than an individual:
  - (i) any Person that directly or indirectly controls, is directly or indirectly controlled by, or is directly or indirectly under common control with, such specified Person;
  - (ii) each Person that serves as a director, officer, manager, executor, or trustee of such specified Person (or in a similar capacity); and
  - (iii) any Person with respect to which such specified Person serves as a general partner, manager, or a trustee (or in a similar capacity).

- (h) For purposes of this definition:
  - (i) "control" (including "controlling," "controlled by," and "under common control with") means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract, or otherwise, and shall be construed as such term is used in the rules promulgated under the Exchange Act; and
  - (ii) the "Family" of an individual includes (A) the individual, (B) the individual's spouse, (C) any other natural person who is related to the individual or the individual's spouse within the second degree, and (D) any other natural person who resides with such individual.

"**Release**"—any release, spill, emission, leaking, pumping, pouring, dumping, emptying, injection, deposit, disposal, discharge, dispersal, leaching, or migration on or into the Environment, or into or out of any property.

"Representative"—with respect to a particular Person, includes any director, officer, manager, employee, agent, consultant, advisor, accountant, financial advisor, or legal counsel of such Person.

**"Restricted Cash"**— means any (a) Cash on Hand of the Acquired Companies, wherever and however held, where usage of such Cash on Hand is restricted by any Legal Requirement, contract (including customer deposits and security deposits and the ARP Funds) or otherwise, *plus* (b) Cash on Hand equal to the amount of all outstanding checks or drafts written on an account of the Acquired Companies that have not yet cleared, *plus* (c) Cash on Hand, wherever and however held, where such amounts have been received but additional goods or services are required to be supplied in order for the Acquired Companies to be entitled to keep such Cash on Hand.

"**R&W Binder**"—the Binder Agreement for the R&W Policy issued by the R&W Insurer, a copy of which is attached as <u>Exhibit B</u> attached hereto.

"R&W Insurer"—the insurer under the R&W Policy.

"R&W Policy"—the Representations and Warranties Insurance Policy, together with endorsements related thereto, issued by the R&W Insurer, a draft of which is attached as an exhibit to the R&W Binder, as in effect on the Closing Date.

"Securities Act"—the Securities Act of 1933, as amended.

"**Seller**"—as defined in the first paragraph of this Agreement.

"**Seller's Closing Documents**"—the documents required to be executed or delivered by Seller at the Closing pursuant to <u>Section 2.6(a)</u>.

**"Seller's Expenses"**—any and all transaction related fees, costs and expenses incurred by or on behalf of the Seller Parent, Seller or any of the Acquired Companies, including without

limitation fees, costs, commissions and expenses of any brokers, investment bankers, attorneys, accountants, financial advisors and other advisors, consultants or service providers, as well as any payments owing under any appreciation rights, phantom equity or similar plans or any deferred compensation, retention, severance, deferred purchase price, annual or special bonus or profit sharing payments attributable to pre-Closing periods and all bonuses and other payments payable by Seller Parent, Seller or any Acquired Company to any employee of an Acquired Company as a result of the transactions contemplated by this Agreement, including without limitation, those bonus payments contemplated by that certain Incentive Bonus Agreement, dated February 2023, by and between Brandy McClain and Tennessee Valley Home Care, LLC, and the Incentive Bonus Agreement, dated February 2023, by and between Kelly Nichols and American Health Partners Management, LLC, plus the employer's share of any payroll Taxes due in connection with such payments, and all vacation, sick-time and other paid time off as of the Closing to the extent not included in the calculation of Net Working Capital. Seller's Expenses shall also include obligations to provide any compensation, benefits or other amounts due to any current or former employee, officer, director, independent contractor or consultant of the Acquired Companies under any Company Benefit Plan that relate to events or any time period on or prior to the Closing Date, which amounts shall include, but not be limited to, the satisfaction of all claims for medical, dental, life insurance, health accident or disability benefits that relate to events occurring on or prior to the Closing Date. Further, Seller's Expenses shall also include any fees owned to Mitchell Family Office, Inc., a Michigan corporation and Affiliate of Seller Parent, pursuant to that certain Agreement for Management Assistance Services, dated April 6, 2021 to the extent not included in the calculation of Net Working Capital.

"**Software**"— all computer software, programs, operating systems, applications, firmware, and similar systems owned by or licensed to the Acquired Companies or used in the conduct of the business of the Acquired Companies.

"Straddle Period" means any taxable year or period beginning before the Closing Date and ending after the Closing Date.

"Straddle Period Return"—as defined in Section 12.1(a).

"Subsidiary"—with respect to any Person (the "Owner"), any other Person of which securities or other interests having the power to elect a majority of that other Person's board of directors or similar governing body, or otherwise having the power to direct the business and policies of that other Person (other than Equity Securities or other interests having such power only upon the happening of a contingency that has not occurred) are held by the Owner or one or more of its Subsidiaries; when used without reference to a particular Person.

"Tax"— means (i) any and all taxes, fees, levies, duties, tariffs, imposts and other similar charges, imposed by any taxing authority or other Governmental Body, including taxes or other charges on, measured by, or with respect to income, franchise, windfall or other profits, gross receipts, property, sales, use, capital stock, payroll, employment, social security, workers' compensation, escheat, unclaimed property, unemployment compensation or net worth; and taxes or other charges in the nature of excise, withholding, ad valorem, stamp, transfer, value-added or gains taxes; license, registration and documentation fees; and customs duties, tariffs and similar charges; (ii) any and all interest, penalties, additions to tax and additional amounts imposed in

connection with or with respect to the foregoing; (iii) any liability for payment of amounts described in clauses (i) or (ii), whether as a result of transferee liability, of being a member of an affiliated, consolidated, combined or unitary group for any taxable period (including, without limitation, any liability pursuant to Treasury Regulations Section 1.1502-6 or any similar provision of state, local, or non-U.S. Legal Requirement), by Contract, or otherwise through operation of any Legal Requirement; (iv) any liability for the payment of amounts described in clauses (i), (ii) or (iii) as a result of any tax sharing, tax indemnity or tax allocation agreement or any other agreement to pay any Taxes of any other Person.

**"Tax Proceeding"** means any pending or, to the Knowledge of Seller, threatened, in writing, federal, state, local or foreign Tax audits, examinations or assessments relating to Taxes or Tax Returns of the Acquired Companies or Seller for the Pre-Closing Tax Period.

"**Tax Return**"— any return, report, information return or other document (including schedules or any related or supporting information) filed or required to be filed with any Governmental Body or other authority in connection with the determination, assessment or collection of any Tax or the administration of any laws, regulations or administrative requirements relating to any Tax.

**"Third-Party Payor"**—any fiscal intermediary acting on behalf of any Government Program, or any private, commercial or other non-governmental insurance program, managed care organization or other third-party payor, including, without limitation, the Medicare Advantage and the Medicaid Managed Care programs.

"**Threat of Release**"—a reasonable possibility of a Release that could require action (including triggering notification or reporting under Environmental Law) in order to prevent or mitigate damage to the Environment that could result from such Release.

"Transfer Taxes"—as defined in Section 12.1(g).

"Transition Services Agreement"—as defined in Section 2.6(a)(xiii).

"**Treasury Regulations**" — means the proposed, temporary and final regulations promulgated under the Code by the U.S. Department of the Treasury, as amended.

## 1.2Usage.

- (a) In this Agreement, unless expressly stated otherwise:
  - (i) the singular includes the plural and vice versa;
  - (ii) reference to any Person includes such Person's successors and assigns, if applicable, but only if such successors and assigns are permitted by this Agreement, and reference to a Person in a particular capacity excludes such Person in any other capacity;
  - (iii) reference to a gender includes any other gender;

- (iv) reference to any agreement, document, or instrument means such agreement, document, or instrument as amended or modified and in effect from time to time in accordance with its terms;
- (v) reference to any Legal Requirement means that Legal Requirement as from time to time in effect, including any amendment, modification, codification, replacement, or reenactment of such Legal Requirement;
- (vi) reference to any section or other provision of any Legal Requirement means that provision of such Legal Requirement as from time to time in effect, including any amendment, modification, codification, replacement, or reenactment of such section or other provision;
- (vii) "hereunder," "hereof," "hereto," and words of similar import refer to this Agreement as a whole and not to any particular Article, Section, or other provision of this Agreement;
- (viii) "including" (and with correlative meaning "include") means including without limiting the generality of any description preceding such term;
- (ix) "or" is used in the inclusive sense of "and/or";
- (x) "any" means "any and all";
- (xi) with respect to the determination of any period of time, "from" means "from and including" and "to" means "to but excluding";
- (xii) a reference to a document, instrument, or agreement also refers to all addenda, exhibits, or schedules thereto;
- (xiii) a reference to a "copy" or "copies" of any document, instrument, or agreement means a copy or copies that are complete and correct; and
- (xiv) a reference to a list, or any like compilation (whether in the Seller's Disclosure Statement or elsewhere), means that the item referred to is complete and correct.
- (b) Unless otherwise specified in this Agreement, all accounting terms used in this Agreement will be interpreted, and all accounting determinations under this Agreement will be made, in accordance with GAAP.
- (c) This Agreement was negotiated by the Parties with the benefit of legal representation, and any rule of construction or interpretation otherwise requiring this Agreement to be construed or interpreted against any party as having been drafted by it will not apply to any construction or interpretation of this Agreement.

(d) The headings contained in this Agreement are for convenience of reference only, shall not be deemed to be part of this Agreement, and shall not be referred to in connection with the construction or interpretation of this Agreement.

## 2. SALE AND TRANSFER OF MEMBERSHIP INTERESTS; CLOSING

**2.1Membership Interests**. Subject to the terms and conditions of this Agreement, and in reliance upon the representations, warranties, and covenants contained in this Agreement, at the Closing, Buyer shall purchase the Membership Interests from Seller, and Seller shall sell and transfer the Membership Interests to Buyer, free and clear of any Encumbrance.

#### 2.2Purchase Price.

- (a) The purchase price for the Membership Interests (the "**Purchase Price**") is One Hundred and Six Million and no/100 dollars (\$106,000,000.00) (the "**Gross Purchase Price**") (subject to adjustments pursuant to <u>Sections 2.3</u> and <u>2.4</u> of this Agreement).
- (b) At the Closing, Buyer shall pay to Seller by wire transfer of immediately available funds a net amount equal to (A) the Gross Purchase Price, minus (B) the Estimated Closing Debt Amount, minus (C) the Estimated Seller's Expenses Amount, plus (D) the Estimated Cash on Hand Amount plus (E) the excess, if any, of the amount of Estimated Net Working Capital Amount over the amount of the Net Working Capital Target, minus (F) the excess, if any, of the amount of the Net Working Capital Target over the amount of the Estimated Net Working Capital Amount (such aggregate net amount being, the "Closing Cash Amount").
  - (c) [intentionally omitted]
- (d) At the Closing, Buyer shall pay, on behalf of the Acquired Companies, to the Persons identified in the Debt Payoff Letters, the Estimated Closing Debt Amount attributable to the third party Debt in accordance with the wire instructions set forth in the Debt Payoff Letters.
- (e) At the Closing, Buyer shall pay, on behalf of the Acquired Companies or Seller, as the case may be, to the Persons identified in the Expense Payoff Letters, the Estimated Seller's Expenses Amount in accordance with the wire instructions set forth in the Expense Payoff Letters.

## 2.3Closing Purchase Price Adjustments.

(a) Seller shall prepare in good faith and, not less than four (4) Business Days prior to the Closing Date, deliver to Buyer, an estimated consolidated balance sheet of the Acquired Companies as of the Effective Time (the "Estimated Closing Balance Sheet") which will be accompanied by a statement (the "Estimated Closing Statement") setting forth Seller's calculation of (i) the Net Working Capital as of the Effective Time (the "Estimated Net Working Capital Amount"), (ii) the Debt of the Acquired Companies as of the Effective Time (the "Estimated Closing Debt Amount"), (iii) Seller's Expenses as of the Effective Time (the "Estimated Seller's Expenses Amount"), (iv) Cash on Hand as of the Effective Time, and (v) the Closing Cash Amount (the "Estimated Cash on Hand Amount"), in each case, together with reasonably detailed supporting calculations and documentation. Seller shall prepare the Estimated

Closing Balance Sheet and Estimated Closing Statement in accordance with GAAP and the assumptions and procedures set forth on Exhibit A of this Agreement, which is attached solely for illustrative purposes to demonstrate the manner in which the Estimated Closing Balance Sheet, the Estimated Closing Statement, the Closing Balance Sheet and the Closing Statement are to be prepared. Upon delivery of the Estimated Closing Balance Sheet and Estimated Closing Statement, Seller shall provide Buyer and its Representatives with reasonable access to the relevant employees, accountants and books and records of the Acquired Companies to review the accuracy of such amounts. In the event that Buyer does not agree in good faith with Seller's calculation of the Estimated Net Working Capital Amount, Estimated Closing Debt Amount, Estimated Seller's Expenses Amount, and/or Estimated Cash on Hand Amount, Buyer shall so notify Seller in writing no later than two (2) Business Days after Buyer receives Seller's calculations, and Seller and Buyer shall negotiate in good faith to mutually agree upon acceptable estimates and each such Party shall consider in good faith any proposed comments or changes that the other Party may reasonably suggest; provided, however, that Seller's failure to include any changes by Buyer or Buyer's failure to accept changes or positions of Seller, or the acceptance by any Party of the Estimated Closing Balance Sheet and Estimated Closing Statement, shall not limit or otherwise affect any such Party's right to include such changes or other changes in the Closing Balance Sheet or Closing Statement, or constitute an acknowledgement by any Party of the accuracy of the Estimated Net Working Capital Amount, Estimated Closing Debt Amount, the Estimated Seller's Expenses Amount or Estimated Cash on Hand Amount; provided, further, however, that if Seller and Buyer are unable to agree on the Estimated Net Working Capital Amount, the Estimated Closing Debt Amount, or the Estimated Seller's Expenses Amount, then the Closing shall occur on the Closing Date and the Seller's calculations shall be used at Closing, subject to adjustments, if any, pursuant to Section 2.4.

#### 2.4Post-Closing Purchase Price Adjustment.

- (a) No later than one hundred twenty (120) days after the Closing Date, Buyer shall cause a consolidated balance sheet of the Acquired Companies as of the Effective Time (the "Closing Balance Sheet") to be prepared and delivered to Seller, which will be accompanied by a statement (the "Closing Statement") setting forth Buyer's good faith calculation of (i) the Net Working Capital as of the Effective Time, (ii) the outstanding Debt of the Acquired Companies as of the Effective Time, (iii) outstanding Seller's Expenses as of the Effective Time, (iv) the Cash on Hand as of the Effective Time, and (v) the Adjusted Closing Cash Amount. Buyer shall prepare the Closing Balance Sheet and the Closing Statement in accordance with GAAP and the assumptions and procedures set forth on Exhibit A of this Agreement. Buyer shall provide a reasonable opportunity for Seller and its Representatives to review the books and records of the Acquired Companies used in preparing or supporting the Closing Balance Sheet and the Closing Statement and the numbers contained therein and shall provide Seller with reasonable access to Buyer's personnel in connection therewith, subject to reasonable limitations to comply with Legal Requirements or protect confidential information or legal privilege.
- (b) If, within forty-five (45) days following delivery of the Closing Balance Sheet and the Closing Statement (the "**Objection Notice Period**"), Seller has not given Buyer written notice of any good faith objection to the Closing Statement (which notice is to state in reasonable detail the basis of its objection and the matters in dispute and the amount of any proposed adjustments in accordance with <u>Section 2.4(c)</u> of this Agreement) (an "**Objection Notice**"), then

Buyer's calculations of the Adjusted Closing Cash Amount is to be deemed to be binding and conclusive on the Parties for all purposes under this Agreement and not subject to further dispute or challenge.

- If Seller gives Buyer an Objection Notice with respect to the Closing Statement within the Objection Notice Period, then Seller and Buyer shall in good faith attempt to resolve the specific issues set forth in reasonable detail in such Objection Notice (and, for avoidance of doubt, all other calculations with respect to the Adjusted Closing Cash Amount are to be binding and conclusive on the Parties). Any resolution agreed to in writing by Seller and Buyer shall be final, binding and conclusive on the Parties and is to be used in the calculation of the Adjusted Closing Cash Amount. If Seller and Buyer fail to resolve such issues within thirty (30) days following Buyer's receipt of such Objection Notice, Seller and Buyer shall submit the specific issues (and only such issues) set forth in reasonable detail in such Objection Notice remaining in dispute to a nationally recognized certified public accounting firm mutually selected by Seller and Buyer (the "Independent Accountants") for resolution in accordance with the guidelines and procedures set forth in this Agreement. If issues are submitted to the Independent Accountants for resolution, (i) Seller and Buyer shall furnish or cause to be furnished to the Independent Accountants such work papers and other documents and information relating to the disputed issues as the Independent Accountants request and are available to such Party or such Party's Representatives and are to be afforded the opportunity to present to the Independent Accountants any material relating to the disputed issues, (ii) Seller and Buyer shall direct the Independent Accountants to notify Seller and Buyer of their determination within thirty (30) days of the submission to the Independent Accountants of the issues remaining in dispute, and (iii) the determination by the Independent Accountants is to be final, binding and conclusive on the Parties and is to be used in the calculation of the Adjusted Closing Cash Amount, (iv) in no event shall any determination of the Independent Accountants be outside of the range of Buyer's proposed final calculations of such disputed issues set forth in the Closing Statement and Seller's proposed final calculations set forth in the Objection Notice, and (v) the fees and expenses of the Independent Accountants are to be allocated and payable by Buyer, on the one hand, and Seller on the other hand, in proportion to the amounts by which the proposals of Buyer and Seller, respectively, differed from the Independent Accountants' final determination of the matters in dispute, and the Independent Accountants are to determine such proportions in the Independent Accountant's final determination.
- (d) Within five (5) Business Days after the earlier of (i) the expiration of the Objection Notice Period in the event no Objection Notice is delivered to Buyer prior thereto, (ii) the joint final resolution of Buyer and Seller of all disputed items set forth in an Objection Notice, or (iii) the final determination by the Independent Accountants in accordance with Section 2.4(c):
  - (i) If the Adjusted Closing Cash Amount is equal to the Closing Cash Amount, then no further payment will be required.
  - (ii) If the Adjusted Closing Cash Amount is greater than the Closing Cash Amount, then Buyer will pay to Seller the full amount of such excess.

(iii) If the Adjusted Closing Cash Amount is less than the Closing Cash Amount, then Seller and Seller Parent, jointly and severally, will pay to Buyer the full amount of such deficit.

Any payments made under this <u>Section 2.4</u> shall be treated by the Parties as an adjustment to the Purchase Price for income Tax purposes, unless a final determination (which is to include the execution of an IRS Form 870-AD or successor form) with respect to such payment causes any such payment not to be treated as an adjustment to the Purchase Price for Tax purposes. Nothing in this <u>Section 2.4</u> is intended to diminish Seller's and Seller Parent's obligations for all Seller's Expenses and Debt of the Acquired Companies.

**2.5Closing**. The closing of the purchase and sale (the "Closing") provided for in this Agreement will take place at the offices of Bass, Berry & Sims PLC at 150 Third Avenue South, Suite 2800, Nashville, Tennessee 37201, commencing at 10:00 a.m. (local time) on the first day of the month following the date on which all conditions set forth in Articles 8 and 9 have been satisfied or waived and the Closing shall be deemed effective for all purposes as of the Effective Time. All deliveries, payments and other transactions and documents relating to the Closing shall not be effective unless and until all are effective (or waived in writing by the applicable Party) and shall be deemed to be consummated simultaneously.

## **2.6Closing Obligations**.

- (a) At the Closing Seller shall deliver to Buyer:
  - (i) a bill of sale and assignment of membership interest agreement transferring title to the Membership Interests to Buyer;
  - (ii) the Organizational Documents of each Acquired Company filed with any Governmental Body in connection with its organization, duly certified as of a recent date by the Secretary of State or other appropriate authority of the jurisdiction of its incorporation or organization, together with a certificate dated as of the Closing Date from the Secretary of each Acquired Company to the effect that no amendments to such Organizational Documents have been filed since the date referred to above:
  - (iii) the Organizational Documents, if any, of each Acquired Company not filed with a Governmental Body in connection with its organization, certified as of the Closing Date by the Secretary of each Acquired Company;
  - (iv) certificates dated as of a date not more than ten (10) days prior to the Closing Date as to the good standing of each Acquired Company, issued by the appropriate Governmental Body of the jurisdiction of the Acquired Company's organization and each jurisdiction in which the Acquired Company is licensed or qualified to do business as a foreign entity;

- (v) the certificate referred to in <u>Section 8.3</u>;
- (vi) payoff letters for the Debt set forth on <u>Section 2.6(a)(vi)</u> of Seller's Disclosure Statement (the "**Debt Payoff Letters**");
- (vii) payoff letters or invoices with respect to Seller's Expenses (the "Expense Payoff Letters");
- (viii) resignations of offices from such officers of the Acquired Companies as requested by Buyer no later than five (5) Business Days before Closing;
- (ix) [intentionally omitted];
- (x) all required third-party consents and approvals set forth on <u>Section 3.4</u> of the Seller's Disclosure Statement;
- (xi) restrictive covenant agreements in the forms set forth in <u>Exhibit 2.6(a)(xi)</u>, executed by each of Seller Parent, Mark T. Mitchell and Michael Bailey (the "**Restrictive Covenant Agreements**");
- (xii) a duly executed IRS Form W-9 from Seller
- (xiii) evidence that the Management Agreements and any other Contracts identified on <u>Section 2.6(a)(xiii)</u> of Seller's Disclosure Statement have been terminated;
- (xiv) a transition services agreement in the form of Exhibit 2.6(a)(xiv), executed by Seller (the "Transition Services Agreement");
- (xv) evidence, in form and substance reasonably satisfactory to Buyer, that Seller has changed the address of the Acquired Companies with the IRS to 2300 Warrenville Rd, Suite 100, Downers Grove, IL 60515;
- (xvi) evidence, in form and substance reasonably satisfactory to Buyer, that Seller has changed the registered agent of the Acquired Companies to CSC Global by filing a duly executed Form SS-4534 for each Acquired Company with the Secretary of State of Tennessee;
- (xvii) evidence (A) of discharge of all patients of American National Home Health Care, Inc. ("<u>ANHC</u>") and Continuous Care Services, LLC ("<u>CCS</u>") from the care of such entities and readmission to the Acquired Companies, as applicable and subject to patient choice, and (b) that such entities have surrendered their licenses to the Tennessee Health Facilities Commission;

- (xviii) [intentionally omitted];
- (xix) evidence, in form and substance reasonably satisfactory to Buyer, that Seller has transferred the employment of those individuals set forth on Section 2.6(a)(xix) of Seller's Disclosure Statement from Seller or any of its Affiliates to an Acquired Company prior to Closing;
- (xx) evidence, in form and substance reasonably satisfactory to Buyer, that Seller Parent has transferred the domain names set forth on <u>Section 2.6(a)(xx)</u> of Seller's Disclosure Statement to reflect the Company as the registrant holder of such domain names as of Closing;
- (xxi) evidence, in form and substance reasonably satisfactory to Buyer, that Seller Parent has transferred the social media accounts set forth on Section 2.6(a)(xxi) of Seller's Disclosure Statement to the Company as of Closing;
- (xxii) evidence, in form and substance reasonably satisfactory to Buyer that Seller has (a) transferred ownership of the vehicles set forth on <u>Section 2.6(a)(xxii)</u> of Seller's Disclosure Statement from Seller or any of its Affiliates to an Acquired Company prior to Closing and (b) provided the original title documents to all vehicles owned by an Acquired Company; and
- (xxiii) such other deeds, bills of sale, assignments, certificates of title, documents and other instruments of transfer and conveyance as may reasonably be requested by Buyer, each in form and substance reasonably satisfactory to Buyer and executed by Seller and/or the Company, as applicable, including those items set forth on <a href="Section 2.6(a)(xxiii)">Section 2.6(a)(xxiii)</a> of Seller's Disclosure Statement.
- (b) At the Closing, Buyer shall deliver to Seller:
  - (i) the Closing Cash Amount;
  - (ii) [intentionally omitted];
  - (iii)[intentionally omitted];
  - (iv)the Restrictive Covenant Agreements, executed by Buyer
  - (v) the Transition Services Agreement, executed by Buyer; and
  - (vi)the certificate referred to in Section 9.3.

**2.7Withholding**. Notwithstanding any other provision of this Agreement, Buyer (or any agent of Buyer) shall be entitled to deduct and withhold from any amounts payable pursuant to this Agreement any amounts required to be deducted and withheld under the Code or any other applicable Legal Requirement with respect to the making of such payment, but, other than in the case of compensatory payments, only after providing reasonable advance written notice to the Person in respect of which such deduction and withholding is proposed to be made. To the extent that any such amounts are so deducted and withheld, such amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding was made.

#### 3. REPRESENTATIONS AND WARRANTIES OF SELLER PARENT AND SELLER

Except as disclosed in, and subject to disclosures contained in, Seller's Disclosure Statement attached hereto, Seller Parent and Seller jointly and severally represent and warrant to Buyer as follows:

#### 3.1Corporate Organization.

- (a) Seller is (i) a limited liability company duly organized, validly existing and in good standing in the State of Michigan, (ii) duly qualified to do business and in good standing in all jurisdictions (whether federal, state, local or foreign) where its ownership or leasing of property or the conduct of its business requires it to be so qualified, except where the failure to be so qualified or in good standing would not reasonably be expected to have a Material Adverse Effect on Seller or the Acquired Companies, and (iii) has all requisite corporate power and authority to own or lease its properties and assets and to carry on its business as now conducted, including performing all of its obligations under this Agreement and Seller's Closing Documents.
- (b) Seller Parent and each of the Acquired Companies is (i) a limited liability company duly organized, validly existing and in good standing under the laws of the State of Tennessee, (ii) duly qualified to do business and in good standing in all jurisdictions (whether federal, state, local or foreign) where its ownership or leasing of property or the conduct of its business requires it to be so qualified, except where the failure to be so qualified or in good standing would not reasonably be expected to have a Material Adverse Effect on such Acquired Company or Seller Parent, and (iii) has all requisite corporate power and authority to own or lease its properties and assets and to carry on its business as now conducted.
- (c) <u>Section 3.1(c)</u> of Seller's Disclosure Statement contains copies of the operating agreements for each of the Acquired Companies, with all amendments, modifications and supplements, in effect as of the date of this Agreement.

#### 3.2Capitalization/ Ownership of Membership Interests.

(a) As of the date here hereof, Seller Parent owns, directly, all of the Membership Interests as set forth on Section 3.2(a) of Seller's Disclosure Statement. The Membership Interests represent all of the issued and outstanding equity interests of the Company. Immediately prior to the Closing, Seller will own, directly, all of the Membership Interests, and the Membership Interests will represent all of the issued and outstanding equity interests of the Company.

- (b) Section 3.2(b) of the Seller's Disclosure Statement lists each of the Company's Subsidiaries, including its issued and outstanding Equity Securities, and each record holder of its issued and outstanding Equity Securities. All of the outstanding Equity Securities in each of the Company's Subsidiaries have been duly authorized and validly issued, are fully paid and nonassessable and are free of preemptive rights, and are wholly owned beneficially and of record by an Acquired Company, free and clear of any Encumbrance. Except for the Acquired Companies (or as contemplated by the Pre-Closing Reorganization), the Company has no Subsidiaries and does not own directly or indirectly, any capital stock, limited liability company interest, partnership interest or other equity interest in any Person.
- (c) Except for this Agreement, there are no options, subscriptions, warrants, calls, rights, commitments or agreements of any character to which any Acquired Company is a party or by which any Acquired Company is bound obligating any Acquired Company to issue, deliver or sell, or cause to be issued, delivered or sold, any Equity Securities of any Acquired Company or any phantom Equity Securities of any Acquired Company or obligating any Acquired Company to extend or enter into any such option, subscription, warrant, call, right, commitment or agreement. There are no outstanding contractual obligations of any Acquired Company (A) to repurchase, redeem or otherwise acquire any Equity Securities of such Acquired Company or (B) pursuant to which any Acquired Company is or could be required to register any securities under the Securities Act.
- (d) As of the date hereof, Seller Parent owns the Membership Interests beneficially and of record, free and clear of all Encumbrances (except Permitted Encumbrances, any restrictions contained in the operating agreement of the Acquired Companies or as listed on Part I of Section 3.2(d) of Seller's Disclosure Statement), restrictions on transfer, rights of first refusal, options, calls or other similar restrictions and has and will have all necessary lawful right to transfer such Membership Interests to Seller immediately prior to the Closing. As of the date hereof, Seller Parent has sole voting power and sole power of disposition with respect to such Membership Interests. Immediately prior to the Closing, Seller will own the Membership Interests beneficially and of record, free and clear of all Encumbrances (except Permitted Encumbrances, any restrictions contained in the operating agreement of the Acquired Companies or as listed on Part II of Section 3.2(d) of Seller's Disclosure Statement), restrictions on transfer, rights of first refusal, options, calls or other similar restrictions and will have all necessary lawful right to transfer such Membership Interests to Buyer pursuant to the terms of this Agreement. Immediately prior to the Closing, Seller will have sole voting power and sole power of disposition with respect to such Membership Interests. At the Closing, Seller shall convey to Buyer valid title to such Membership Interests.

### 3.3Authority; No Violation.

(a) Seller Parent, Seller, and each Acquired Company represent and warrant, that (i) Seller Parent, Seller, and each Acquired Company have the limited liability company power and authority to execute and deliver this Agreement and to consummate the Contemplated Transactions; (ii) the execution and delivery of this Agreement and the consummation of the Contemplated Transactions have been duly and validly approved by the equity owners of Seller Parent, Seller, and each Acquired Company to the extent required under applicable Legal Requirements or pursuant to the Organizational Documents of Seller Parent, Seller and each

Acquired Company; and (iii) this Agreement has been duly and validly executed and delivered by Seller Parent, Seller, and each Acquired Company and (assuming due authorization, execution and delivery by Buyer) constitutes valid and binding obligations of Seller Parent, Seller, and each Acquired Company, enforceable against Seller Parent, Seller, and each Acquired Company in accordance with its terms (except as may be limited by bankruptcy, insolvency, moratorium, reorganization or similar laws affecting the rights of creditors generally and the availability of equitable remedies).

- (b) Neither the execution and delivery of this Agreement by Seller Parent, Seller, and each Acquired Company, nor the consummation by Seller Parent, Seller, and each Acquired Company of the Contemplated Transactions, nor compliance by Seller Parent, Seller, and each Acquired Company with any of the terms or provisions hereof, will (i) violate any provision of the Organizational Documents of any Acquired Company, or (ii) assuming that the Consents referred to in Section 3.4 are duly obtained, (x) violate any statute, code, ordinance, rule, regulation, judgment, order, writ, decree or injunction applicable to any Acquired Company or any of their respective properties or assets or (y) violate, conflict with, result in a breach of any provision of or the loss of any benefit under, constitute a default (or an event which, with notice or lapse of time, or both, would constitute a default) under, result in the termination of, loss of rights under or a right of termination, modification or cancellation under, accelerate the payment or performance required by, or result in the creation of any Encumbrance upon any of the respective properties or assets of any Acquired Company under, any of the terms, conditions or provisions of any Company Material Contract, note, bond, mortgage, indenture, deed of trust, or license to which any Acquired Company is a party, or by which it or any of its properties or assets may be bound or affected, except in the case of clause (ii) above for such violations, conflicts, breaches, losses, defaults, terminations, cancellations, accelerations, or Encumbrances, which would not, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Acquired Companies.
- **3.4Consents and Approvals.** Except (i) as disclosed on <u>Section 3.4</u> of Seller's Disclosure Statement and (ii) any notice or filings required under the HSR Act, no Consents or filings or registrations with any Governmental Body are necessary in connection with (A) the execution and delivery by Seller Parent, Seller, or any Acquired Company of this Agreement and (B) the consummation by Seller Parent, Seller or any Acquired Company of the Contemplated Transactions. Except for any Consents which are listed in <u>Section 3.4</u> of Seller's Disclosure Statement, no Consents of any person, other than a Governmental Body, are necessary under any Company Material Contract in connection with (A) the execution and delivery by Seller Parent, Seller, or any Acquired Company of the Contemplated Transactions.
- **3.5Financial Statements.** Section 3.5 of Seller's Disclosure Statement contains the audited consolidated financial statements of the Acquired Companies for the years ended December 31, 2021 and 2022 (including the related notes) and the unaudited consolidated financial statements of such Acquired Companies (the "Interim Financial Statements") for the fourmonth period ended April 30, 2023 (the "Interim Balance Sheet Date") (collectively the "Financial Statements"). The Financial Statements fairly present in all material respects the consolidated results of operations, changes in shareholders' equity, cash flows and financial position of such Acquired Companies for the respective fiscal periods or as of the respective dates therein set forth;

each of such statements (including the related notes, where applicable) has been prepared in accordance with GAAP consistently applied during the periods involved, except, in each case, as indicated in such statements or in the notes thereto, and except, in the case of the Interim Financial Statements, for the absence of footnotes and subject to typical year-end adjustments the effect of which will not, individually or in the aggregate, be materially adverse. The Acquired Companies maintain internal accounting controls sufficient to provide reasonable assurance that transactions are executed with management's general or specific authorizations and transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP.

**3.6Brokers or Finders.** Except as set forth on <u>Section 3.6</u> of Seller's Disclosure Statement, neither Seller Parent, Seller, the Acquired Companies nor any of their respective officers or directors has employed any broker or finder or incurred any liability for any broker's fees, commissions or finder's fees in connection with the Contemplated Transactions.

#### 3.7Absence of Certain Changes or Events.

Since May 31, 2022 and except, in each case, as set forth on <u>Section 3.7</u> of Seller's Disclosure Statement or as contemplated or permitted by this Agreement (including, without limitation, the transactions contemplated by the Pre-Closing Reorganization), the Acquired Companies have conducted their business only in the Ordinary Course of Business consistent with past practice and none of the Acquired Companies has:

- (a) experienced a change or development in the business, operations, assets, liabilities, condition (financial or otherwise), results of operations or cash flows of any of the Acquired Companies which, individually or in the aggregate, has had, or would reasonably be expected to have, a Material Adverse Effect with respect to the Acquired Companies as a whole;
- (b) other than in the Ordinary Course of Business, incurred any indebtedness for borrowed money, or assumed, guaranteed, endorsed or otherwise as an accommodation become responsible for the obligations of any other person;
- (c) (i) adjusted, split, combined or reclassified the Equity Securities of any Acquired Company; (ii) granted any stock options, warrants, stock appreciation rights, performance shares, restricted stock units, restricted shares or other equity-based awards or interests, or granted any person any right to acquire any Equity Securities of any Acquired Company; or (iii) issued, sold or otherwise permitted to become outstanding any additional Equity Securities or securities convertible or exchangeable into, or exercisable for, any Equity Securities of any Acquired Company;
- (d) settled any claim, suit, action or proceeding against any Acquired Company, except in the Ordinary Course of Business, in an amount and for consideration not in excess of \$100,000 individually or \$500,000 in the aggregate or that did not or would not impose any material restriction on the business of any Acquired Company;
- (e) merged or consolidated any Acquired Company with any other person, or restructured, reorganized or completely or partially liquidated or dissolved any Acquired Company;

- (f) made, or committed to make, any capital expenditures in excess of \$100,000 individually or \$500,000 in the aggregate;
- (g) made, changed or revoked any material Tax election, changed an annual Tax accounting period, adopted or changed any Tax accounting method, filed any amended Tax Return, entered into any "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local, non-U.S. or other Legal Requirement) with respect to Taxes, or settled any Tax claim, audit, assessment or dispute or surrendered any right to claim a refund or credit of Taxes, entered into any Tax sharing, Tax indemnity, Tax allocation or similar agreement or Contract (other than any Contract entered into in the Ordinary Course of Business which does not have Taxes as its primary purpose) or consented to any extension or waiver of the limitation period applicable to any Tax claim or assessment;
- (h) changed its accounting methods, principles or practices, other than changes required by applicable Legal Requirements or GAAP or regulatory accounting as concurred by Acquired Companies' independent public accountants;
- (i) entered into any Contract of more than \$100,000 individually, or \$500,000 in the aggregate, per annum, other than in the Ordinary Course of Business;
- (j) other than as set forth in <u>Section 3.12(a)</u> of Seller's Disclosure Statement, granted any increase in the base pay of any employee or granted any increase in or established or amended any insurance, deferred compensation, pension, retirement, profit sharing, stock option (including, without limitation, the granting of stock options), stock purchase or other employee benefit plan, payable or to become payable to any directors, officers, or employees of the Acquired Companies (other than normal salary adjustments to employees made in the ordinary course of business consistent with past practice), or any grant of severance, termination or change in control pay, or any contract or arrangement entered into to make or grant any severance, termination or change in control pay, any payment of any material bonus, or the taking of any action not in the Ordinary Course of Business with respect to the compensation or employment of directors and officers of any Acquired Company;
  - (k) amended the Organizational Documents of any of the Acquired Companies;
  - (l) entered into any lease of real or personal property, other than in the Ordinary Course of Business;
- (m) declared, set aside or paid any dividend or distribution (whether in cash, securities or property or any combination thereof) in respect of any Equity Securities of any Acquired Company or any securities or obligations convertible into or exchangeable for any equity securities of any Acquired Company or any redemption, purchase or other acquisition of its securities, except for dividends, distributions and other similar capital transactions between or among the Acquired Companies or the distribution of Cash on Hand permitted pursuant to Section 5.2(a); or
- (n) taken any action which, if taken or authorized between the date of this Agreement and the Closing Date, would have required the consent of Buyer pursuant to Section 5.2.

(o) entered into any agreement, whether oral or written, to do or make any commitment to do or adopted any board resolution or committee resolution to do any of the foregoing.

## 3.8Condition and Sufficiency of Assets.

- (a) The Acquired Companies own all tangible personal property reflected as owned on the balance sheet of the Acquired Companies as of the Interim Balance Sheet Date (other than inventory sold since such date in the Ordinary Course of Business), free and clear of all Encumbrances, other than Permitted Encumbrances and those identified on Section 3.8(a) of Seller's Disclosure Statement. All the tangible personal property purchased or otherwise acquired by the Acquired Companies since the Interim Balance Sheet Date (other than inventory acquired and sold since the Interim Balance Sheet Date in the Ordinary Course of Business) is owned by the Acquired Companies free and clear of all Encumbrances other than Permitted Encumbrances and those identified on Section 3.8(a) of Seller's Disclosure Statement. The Acquired Companies have valid leasehold interests in or licenses to (as applicable) the leased or licensed assets or properties used by the Acquired Companies in the Business. The Acquired Companies own or otherwise have the sole and exclusive right to all strategic plans of the Acquired Companies, including all acquisition and transaction prospect lists.
- (b) The assets owned and leased by the Acquired Companies constitute all of the assets used in connection with the business of the Acquired Companies, and such assets constitute all of the assets necessary for the Acquired Companies to continue to conduct their business following the Closing as it is being presently conducted, except for ordinary course business support services provided by Seller Parent or its Affiliates to the Acquired Companies, and to be provided pursuant to the Transition Services Agreement. All of the material tangible assets used in or necessary for the operation of the Business of the Acquired Companies are in good operating condition and repair taking into account the age of such property (normal wear and tear excepted). Seller does not own any assets or properties used in the Business conducted by any Acquired Company.
- **3.9Inventory**. All inventories of the Acquired Companies reflected on the balance sheets of the Acquired Companies as of the Interim Balance Sheet Date consist of a quality and quantity usable in the Ordinary Course of Business, net of any reserves therein. The reserve for obsolescence with respect to inventory is adequate and calculated consistent with past practice.

## 3.10Legal Proceedings.

- (a) Except as disclosed in <u>Section 3.10(a)</u> of the Seller's Disclosure Statement, no Acquired Company is, or has been during the Lookback Period, a party to any, and there are no pending or, to the Knowledge of Seller, threatened in writing, Proceedings or claims of any nature against the Seller Parent, Seller, or any Acquired Company or challenging the validity or propriety of the Contemplated Transactions.
- (b) Except as disclosed in <u>Section 3.10(b)</u> of the Seller's Disclosure Statement, there is, or has been during the Lookback Period, no Order imposed upon Seller, Seller Parent (only with respect to any Acquired Company), or any Acquired Company or the assets of any Acquired Company.

#### **3.11Taxes and Tax Returns.** Except as disclosed in Section 3.11 of Seller's Disclosure Statement:

- (a) Each Acquired Company has duly filed all Tax Returns required to be filed by it on or prior to the date of this Agreement with the appropriate Governmental Body. All such Tax Returns are true, correct and complete in all material respects. Each Acquired Company has timely paid in full to the appropriate Governmental Body all Taxes that are due and owing by such Acquired Companies (whether or not shown on any Tax Return).
- (b) Each Acquired Company has (i) properly and timely deducted and withheld all Taxes required to be so deducted and withheld, and (ii) properly remitted to the proper Governmental Body all Taxes required to be remitted with respect to amounts paid or owing to any employee, independent contractor, stockholder, creditor or other third party, and each Acquired Company has materially complied with all reporting and recordkeeping requirements relating to such Taxes (including, without limitation, the timely filing and delivery of all IRS Forms W-2 and 1099 with respect thereto).
- (c) With respect to any period for which the applicable statute of limitations has not expired, none of the Acquired Companies is delinquent in the payment of any Tax. The Acquired Companies have not executed any agreements or waivers extending any statute of limitations on, or extending the period for the assessment or collection of, any Tax which agreement or waiver is still in effect.
- (d) None of the Acquired Companies is a party to any Contract or other agreement that contains any Tax allocation, Tax sharing or Tax indemnification provision with a Person that is not an Acquired Company (other than commercial agreement entered into in the Ordinary Course of Business and with respect to which Taxes are not a primary purpose).
- (e) There is no Proceeding pending or threatened concerning any liability for Taxes of any Acquired Company. No Acquired Company has received any notice of reassessment, deficiency, claim, adjustment or proposed adjustment or any other notice indicating an intent to open an audit or other review in connection with any Taxes, which notice has not been satisfied by payment or been withdrawn.
- (f) No Acquired Company has any liability for the unpaid Taxes of any Person (other than an Acquired Company) under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local, or non-U.S. Legal Requirements), as a transferee or successor, as a result of joint and several liability, by Contract, or otherwise. No Acquired Company has been a member of an affiliated group filing a consolidated U.S. federal income Tax Return.
- (g) No Acquired Company is, or has ever been, a party to any "reportable transaction" as defined in Section 6707(A)(c)(1) of the Code and Treasury Regulations Section 1.6011-4(b)(1).
- (h) At all times since March 11, 2021, each of the Acquired Companies has been properly classified as a disregarded entity for federal income Tax purposes. At all times prior to March 11, 2021, each of the Acquired Companies was a validly electing S corporation for federal income Tax purposes.

- (i) No Acquired Company is currently the beneficiary of any extension of time within which to file any Tax Return. No written claim has been made by a Governmental Body in a jurisdiction where any of the Acquired Companies does not file Tax Returns that an Acquired Company is or may be subject to taxation by that jurisdiction.
- (j) There are no Encumbrances for any Taxes (other than statutory Encumbrances for Taxes not yet due and payable) on any of the properties or assets of any Acquired Company.
- (k) All Taxes that each Acquired Company has been required to collect in respect of sales Taxes has been collected and have been paid over in the appropriate amounts to the proper Governmental Body in accordance with applicable Legal Requirements, and, to the extent applicable, all documentation and adequate records have been maintained with respect to any customer exemptions with respect thereto.
- (l) Neither Buyer nor any Acquired Company will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any: (i) change in method of accounting, use of the cash method of accounting or use of an improper method of accounting, by any Acquired Company for a taxable period (or portion thereof) ending on or prior to the Closing Date or for any Straddle Period, including under Section 481 of the Code (or any corresponding or similar provision of state, local, non-U.S. or other Legal Requirement); (ii) "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local, non-U.S. or other Legal Requirement) entered into by any Acquired Company on or prior to the Closing; (iii) installment sale or open transaction disposition made by any Acquired Company at or prior to the Closing; (iv) any "long-term contracts" that are subject to a method of accounting provided for in Section 460 of the Code; or (v) any election under, or application of, IRS Revenue Procedure 2004-34, Treasury Regulation Section 1.451-5, Section 455 of the Code or Section 456 of the Code (or any corresponding provision of state or local Legal Requirement), or any other prepaid amounts or advanced payments received or accrued prior to the Closing.
- (m) Fees, charges, costs or expenses pursuant to affiliate services agreements or arrangements to which any Acquired Company is a party have been made on an arms' length basis (within the meaning of Section 482 of the Code) in all material respects. No claim has been asserted by any Governmental Body that any Acquired Company is liable for any Taxes based on Section 482 of the Code (or any corresponding provision of state, local or non-U.S. Legal Requirements).
- (n) Seller is not a "foreign person" as defined in Section 1445(f)(3) of the Code, and the rules and regulations promulgated thereunder.
- (o) Each Acquired Company has complied in all material respects with applicable Legal Requirements regarding escheatment and the reporting of unclaimed property, in each case, to the appropriate Governmental Body.
- (p) No Acquired Company is subject to or has requested any private letter ruling, technical advice memoranda, or any similar ruling with regard to the Tax matters of any Acquired Company.

- (q) No Acquired Company has granted to any Person any power of attorney with respect to any Tax matter that currently is in force.
- (r) No Acquired Company (i) has claimed or received any credits under Sections 7001 through 7005 of the Families First Coronavirus Response Act or Section 2301 of the CARES Act, or (iii) has taken any additional depreciation pursuant to Section 2307 of the CARES Act.
- (s) None of (i) the goodwill, (ii) the going concern value or (iii) the other intangible assets of any Acquired Company that would not be depreciable or amortizable in the hands of Buyer but for Section 197 of the Code, was held or used on or before August 10, 1993 by an Acquired Company or Seller, or any Person related to Buyer, within the meaning of Section 197(f)(9)(C) of the Code, on and after the Closing Date, or could fail to qualify as an "amortizable section 197 intangible" as a result of Section 197(f)(9)(A) of the Code.
- (t) The Acquired Companies have provided to Buyer a true, correct and complete copy of each Acquired Company's U.S. federal income Tax Returns (and all related state income Tax Returns) for all taxable periods beginning after December 31, 2018.
- **3.12Employees.** The sole and exclusive representations and warranties of Seller Parent and Seller with respect to employment law matters are set forth in this <u>Section 3.12</u>.
- (a) <u>Section 3.12(a)</u> of Seller's Disclosure Statement sets forth the following information (to the extent applicable) with respect to each employee of any Acquired Company as of the date of this Agreement, including each employee on leave of absence or layoff status: name, job title, date of hire, current annual base salary or current wage rate (as applicable), 2023 bonus eligibility and/or target, classification as exempt or non-exempt for wage and hour purposes, PTO or vacation leave that is accrued but unused, and services credited for purposes of vesting and eligibility under any Company Benefit Plan. <u>Section 3.12(a)</u> of Seller's Disclosure Statement also sets forth the name, date of engagement and rate of pay of, and services rendered by, any independent contractors who render services on a regular basis to, or are under contract with, any Acquired Company.
- (b) There is no collective bargaining agreement in effect between any Acquired Company and any labor unions or organizations representing any of the employees of the Acquired Companies. None of the Acquired Companies has experienced any organized slowdown, work interruption, strike or work stoppage by its employees, and, to the Knowledge of Seller, there is no strike, labor dispute, unfair labor practice charge, or union organization activity pending or threatened affecting any Acquired Company.
- (c) Except as set forth on Section 3.12(c) of Seller's Disclosure Statement, the employment of each employee of the Acquired Companies is terminable at the will of the respective Acquired Company, as applicable, and no Acquired Company is a party to any employment, non-competition, severance or similar contract or agreement with any employee. No Key Employee of the Acquired Companies has provided written notice of termination of employment, and, to the Knowledge of Seller, no Key Employee of any Acquired Company intends to terminate his or her employment with the Acquired Company. To the Knowledge of Seller, no employee of any Acquired Company is a party to, or is otherwise bound by, any

agreement, including any confidentiality, non-competition or proprietary rights agreement, between such employee and any person other than an Acquired Company that adversely affects the performance of that employee's duties as an employee of the Acquired Company.

- (d) Except as described in Section 3.12(d) of Seller's Disclosure Statement, the Acquired Companies are, and since January 1, 2020, have been, in compliance in all material respects with all Legal Requirements regarding employment and employment practices, including terms and conditions of employment, wages and hours, anti-discrimination and occupational health and safety, collective bargaining, labor practices, classification of employees and contractors, leaves of absence, workers' compensation, unemployment, child labor, background checks, work authorization (including Form I9 requirements), unfair labor practices within the meaning of Section 8 of the National Labor Relations Act, as amended, and the employment of non-residents under the Immigration Reform and Control Act of 1986, as amended ("IRCA"). There is no unfair labor practice charge or any other proceeding brought by or on behalf of any current or former employee or independent contractor of the Acquired Companies under the Fair Labor Standards Act, Title VII of the Civil Rights Act of 1964, the Family Medical Leave Act or any other Legal Requirement pending or, to the Knowledge of Seller, threatened against the Acquired Companies.
- (e) Since January 1, 2018, the Acquired Companies have obtained and retained a complete and accurate copy of each employee's Form I9 (Employment Eligibility Verification Form) and all other records or documents required to be prepared, procured or retained pursuant to IRCA. Since January 1, 2018, the Acquired Companies have not been cited or fined, nor has any legal proceeding been initiated or threatened against the Acquired Companies, by reason of any actual or alleged failure to comply with IRCA.
- **3.13Employee Benefits.** The sole and exclusive representations of Seller Parent and Seller with respect to employee benefits matters are set forth in this <u>Section 3.13</u>.
- (a) Section 3.13(a) of Seller's Disclosure Statement sets forth a true and complete list of all deferred compensation, incentive compensation, stock purchase, stock option or other equity-based, retention, employment, consulting, change in control, severance or termination pay, hospitalization or other medical, life, dental, vision, disability or other insurance, supplemental unemployment benefits, profit-sharing, pension or retirement plans, programs, agreements or arrangements, and each other fringe or other employee benefit plan, program, agreement or arrangement (including any "employee benefit plan" within the meaning of Section 3(3) of ERISA, whether written or unwritten), sponsored, maintained or contributed to or required to be contributed to by any Acquired Company or by any trade or business, whether or not incorporated, which would be deemed a "single-employer" with any Acquired Company under Section 4001 of ERISA (a "Company ERISA Affiliate"), for the benefit of any current or former employee, independent contractor, consultant, officer, manager or director (and/or their dependents or beneficiaries) of the Acquired Companies or any Company ERISA Affiliate, or with respect to which the Acquired Companies or any Company ERISA Affiliate otherwise has any liabilities or obligations (the "Company Benefit Plans").
- (b) No Company Benefit Plan is (i) a "multiemployer plan," as such term is defined in Section 3(37) of ERISA; (ii) a plan that is subject to Title IV of ERISA, Section 302 of ERISA or Section 412 of the Code; (iii) a multiple employer plan as defined in Section 413(c) of

the Code; or (iv) a multiple employer welfare arrangement as defined in Section 3(40) of ERISA, and no Acquired Company nor any Company ERISA Affiliate has ever maintained, contributed to, or been required to contribute to any employee benefit plan described in clauses (i), (ii), (iii) or (iv) above.

- (c) None of the Company Benefit Plans that are "welfare benefit plans," within the meaning of Section 3(1) of ERISA, nor any other written or oral agreement entered into by an Acquired Company, provide for continuing medical, dental, vision, life or disability insurance benefits or coverage after termination or retirement from employment, except for COBRA rights under a "group health plan" as defined in Section 4980B(g) of the Code and Section 607 of ERISA. There exists no basis upon which any Acquired Company or a Company ERISA Affiliate of an Acquired Company would be expected to be subject to any penalties or assessable payments under Section 4980H of the Code, nor has any Acquired Company or any Company ERISA Affiliate of any Acquired Company received any correspondence from the IRS or other agencies indicating that such penalties or assessable payments are or may be due.
- (d) With respect to each Company Benefit Plan, Seller Parent has delivered or made available to Buyer complete copies of each of the following documents: (i) a copy of each written Company Benefit Plan (including any amendments thereto and all administration agreements, insurance policies, investment management or advisory agreements and all prior Company Benefit Plan documents, if amended within the last two (2) years); (ii) a copy of the three (3) most recent Form 5500 annual reports, if any, required under ERISA or the Code; (iii) a copy of the most recent summary plan description (and any summary of material modifications), if any, required under ERISA; (iv) if the Company Benefit Plan is funded through a trust or any third party funding vehicle, a copy of the trust or other funding agreement (including any amendments thereto); (v) if the Company Benefit Plan is intended to be qualified under Section 401(a) of the Code, the most recent determination letter or opinion letter upon which the plan sponsor is entitled to rely received from the IRS; (vi) any actuarial reports; (vii) all material correspondence with the IRS, Department of Labor and the Pension Benefit Guaranty Corporation regarding any Company Benefit Plan; and (viii) all discrimination tests for each Company Benefit Plan for the three (3) most recent plan years, if any. Seller Parent has disclosed or made available to Buyer the terms and conditions of any unwritten Company Benefit Plan.
- (e) Except as identified in Section 3.13(e) of Seller's Disclosure Statement referenced above, during the Lookback Period (i) each of the Company Benefit Plans has been operated and administered in compliance with ERISA and the Code in all material respects, (ii) each of the Company Benefit Plans intended to be "qualified" within the meaning of Section 401(a) of the Code has received a favorable determination letter or opinion letter (upon which the relevant Acquired Company is permitted to rely) from the IRS that such Company Benefit Plan is so qualified, and to the Knowledge of Seller, there are no existing circumstances or any events that would be reasonably likely to adversely affect the qualified status of any such Company Benefit Plan, (iii) all contributions due and payable by the Acquired Companies with respect to each Company Benefit Plan in respect of current or prior plan years have been paid or accrued in accordance with GAAP, (iv) no Acquired Company or Company ERISA Affiliate of an Acquired Company nor, to the Knowledge of Seller, any other person, including any fiduciary, has engaged in a transaction in connection with which the Acquired Companies, any Company Benefit Plan or Buyer will be subject to either a material civil penalty assessed pursuant to applicable Legal

Requirements, including Section 409 or 502(i) of ERISA or a material Tax imposed pursuant to applicable Legal Requirements, including Section 4975 or 4976 of the Code, and (v) there are no pending or, to the Knowledge of Seller, threatened or anticipated claims (other than routine claims for benefits) by, on behalf of or against any of the Company Benefit Plans or any trusts related thereto, or examinations, correction proceedings or audits of any Company Benefit Plan by any Governmental Body.

- (f) Except as identified in Section 3.13(f) of Seller's Disclosure Statement, neither the execution and delivery of this Agreement nor the consummation of the Contemplated Transactions will (either alone or in conjunction with any other event) (i) result (either alone or upon the occurrence of any additional acts or events) in any payment (including, without limitation, severance, unemployment compensation, "excess parachute payment" (within the meaning of Section 280G of the Code), forgiveness of indebtedness or otherwise) becoming due to any director or any employee of any Acquired Company from such Acquired Company under any Company Benefit Plan or otherwise, (ii) increase any benefits otherwise payable under any Company Benefit Plan or (iii) result in any acceleration of the time of payment or vesting of any such benefits. No Acquired Company has announced any type of plan or binding commitment to (1) create any additional Company Benefit Plan, or (2) amend or modify any existing Company Benefit Plan with any current or former employee, independent contractor or director.
- (g) Each Company Benefit Plan that is a "nonqualified deferred compensation plan" (within the meaning of Section 409A(d)(1) of the Code) has been operated in material compliance with Section 409A of the Code, IRS Notice 2005-1, Treasury Regulations issued under Section 409A of the Code, and any subsequent guidance relating thereto, and no additional tax under Section 409A(a)(1)(B) of the Code has been or is reasonably expected to be incurred by a participant in any such Company Benefit Plan, and no employee or former employee of any Acquired Company is entitled to any gross-up or otherwise entitled to indemnification by any Acquired Company or any Company ERISA Affiliate for any violation of Section 409A of the Code.
- (h) There currently is not and never has been any Company Benefit Plan that is or has been subject to the Legal Requirements of a jurisdiction other than the United States.

# 3.14Compliance with Applicable Law.

(a) The Acquired Companies hold, and have at all times during the Lookback Period, held, all material Governmental Authorizations necessary for the lawful conduct of their respective businesses (and have paid all material fees and assessments due and payable in connection therewith) and to the Knowledge of Seller, no suspension or cancellation of any such Governmental Authorization. The Acquired Companies have at all times during the Lookback Period, complied in all material respects with, and are not in default in any material respect under, any applicable Legal Requirements relating to the Acquired Companies, it properties, assets and the operation of its Business.

# 3.15Certain Contracts.

- (a) Except as disclosed in <u>Section 3.15</u> of Seller's Disclosure Statement, no Acquired Company is a party to or bound by any legally binding Contract:
  - (i) with respect to the employment of any directors, officers or employees of an Acquired Company, or the engagement of any independent contractor (excluding any third party vendor),
    - (ii) with respect to a doctor of medicine or other licensed medical provider who is not an employee;
  - (iii)providing for deferred compensation, severance, change in control, "parachute" terms, vesting acceleration, debt forgiveness, transaction bonus, or any similar payments for the benefit of any officers and/or employees;
  - (iv)which, upon the consummation of the Contemplated Transactions will (either alone or upon the occurrence of any additional acts or events) result in any payment (whether of severance pay or otherwise) becoming due from the Acquired Companies or Buyer to any officer or employee thereof,
  - (v) which materially restricts the conduct of any line of business by any Acquired Company or upon consummation of the Contemplated Transactions will materially restrict the ability of Buyer to engage in any line of business.
    - (vi)with or to a labor union or guild (including any collective bargaining agreement),
  - (vii)that requires the payment by the Acquired Companies or Buyer of amounts in excess of \$500,000 individually over the course of their remaining term,
  - (viii)that accounts for receipts by the Acquired Companies or Buyer of amounts in excess of \$500,000 (or \$1,000,0000 with respect to third party payor contracts) individually over the course of their remaining term,
  - (ix)that provides for any Acquired Company's receipt of core services required by the Medicare Conditions of Participation or Conditions of Payment (i.e., respite, general inpatient care, room and board, and outpatient services),
  - (x) with a referral source, including but not limited to any contract with a healthcare provider for the provision of services, rental of space, and/or rental of equipment,
    - (xi)with any Material Contracted Care Partner, Material Supplier or Material Payor,
  - (xii)pursuant to which any Acquired Company manages the operations of any other Person or pursuant to which any Acquired Company has management services provided to it,

(xiii)with a Governmental Body, including, without limitation, any settlement, conciliation or similar contract,

(xiv)(including any stock option plan, stock appreciation rights plan, restricted stock plan or stock purchase plan) any of the benefits under which will be increased, or the vesting of the benefits under which will be accelerated, by the consummation of any of the Contemplated Transactions, or the value of any of the benefits under which will be calculated on the basis of any of the Contemplated Transactions,

(xv) with respect to a lease or license involving any personal properties or personal assets (whether tangible or intangible) in which the annual expense exceeds \$50,000;

(xvi)with respect to an IP License, other than non-exclusive, in-bound licenses of "shrink wrap" and similar generally available commercial Software owned by a third party with an initial or ongoing cost of less than \$50,000 during any twelve (12) month period;

(xvii)under which any Acquired Company has borrowed or loaned money, or any note, bond, indenture, mortgage, installment, obligation or other evidence of indebtedness for borrowed or loaned money or any guarantee of any such indebtedness, in each case, relating to amounts in excess of \$50,000 which are still outstanding;

(xviii)relating to the disposition or acquisition of the assets of, or any Equity Securities in, any business enterprise within the three (3) year period preceding the date of this Agreement, other than the purchase or sale of inventory or assets in the Ordinary Course of Business;

(xix)concerning joint venture or partnership agreements, or the sharing of profits, or any similar arrangement;

(xx)granting an exclusive right to any other party (including any right of first refusal, right of first offer or right of first negotiation);

(xxi)containing a "most favored nation" provision;

(xxii)with any group purchasing organization;

(xxiii)requiring any Acquired Company to indemnify any Person (other than in the Ordinary Course of Business), including any Contract requiring any Acquired Company to act as a guarantor or surety (irrespective of the amount involved); and

(xxiv)with respect to each Lease.

Each Contract of the type described in this <u>Section 3.15(a)</u>, whether or not set forth in Seller's Disclosure Statement, is referred to herein as a "**Company Material Contract.**"

(b) (i) The Acquired Companies have made available to Buyer true and complete copies of each Company Material Contract (inclusive of all amendments thereto), (ii) each Company Material Contract is valid and binding on the Acquired Company that is a party thereto, as applicable, and in full force and effect, (iii) the Acquired Company that is party to the Company Material Contract is in compliance in all material respects with the terms and conditions of each such Company Material Contract is in compliance in all material respects with the terms and conditions of such Company Material Contract, (iv) no event or condition currently exists which constitutes or, after notice or lapse of time or both, will constitute a default on the part of the Acquired Company that is party to the Company Material Contract under any such Company Material Contract and (v) no Acquired Company has received verbal or written notice from any Person who is a party to any Company Material Contract threatening any termination, cancellation, or material change to the terms of any Company Material Contract.

#### 3.16Real Estate.

- (a) None of the Acquired Companies presently owns any real property. Except for the leases and licenses described in Section 3.16(a) of Seller's Disclosure Statement (including all amendments, extensions, renewals, guaranties and other agreements or Contracts with respect thereto) (each a "Lease" and collectively, the "Leases"), no Acquired Company leases any real property. All Leases for real property to which any Acquired Company is a party are in full force and effect and are binding and enforceable against lessee and, to the Knowledge of Seller, the lessors, except as such enforceability may be limited by bankruptcy, insolvency, moratorium or other similar laws affecting or relating to creditors' rights generally and general principles of equity, regardless of whether asserted in a proceeding in equity or at law. True and correct copies of all such Leases, as amended or modified, have been provided or made available to Buyer or its advisors.
- (b) Each of the Acquired Companies is entitled to and has exclusive possession of the real property subject to the Leases (the "Leased Real Property"). The Leased Real Property is not subject to any other legally binding lease, sublease, tenancy, subtenancy or license or any legally binding agreement to grant any such lease, tenancy or license, and there is no person in possession or occupation of, or who has any current right to possession or occupation of, the Leased Real Property other than the Acquired Companies. Each Acquired Company has a good and valid leasehold interest in the applicable Leased Real Property, in each case, free and clear of any Encumbrances.
  - (c) With respect to the Leased Real Property:
    - (i) No Acquired Company is in material default under the terms of the Leases;
    - (ii) to the Knowledge of Seller, the lessor is not in material default under any of the terms of the Leases; and
    - (iii) to the Knowledge of Seller, all existing water, drainage, sewage and utility facilities relating to the Leased Real Property are adequate for

(d) The Leased Real Property is zoned for the purposes for which it is being used by the Acquired Companies.

#### 3.17Undisclosed Liabilities; Indebtedness.

- (a) Except for liabilities and obligations (i) set forth or adequately provided for in the consolidated balance sheet of the Acquired Companies as of the Interim Balance Sheet Date, (ii) incurred in the Ordinary Course of Business since the Interim Balance Sheet Date, (iii) incurred in connection with this Agreement or the Contemplated Transactions, or (iv) set forth in Section 3.17(a) of Seller's Disclosure Statement, no Acquired Company has any liabilities or obligations, whether accrued or fixed, absolute or contingent, matured or unmatured, determined or determinable. Except as set forth in Section 3.17(a) of Seller's Disclosure Statement, none of the Acquired Companies has any Debt.
- (b) Section 3.17(b) of Seller's Disclosure Statement sets forth a list of each loan, application for assistance or stimulus payment that the Acquired Companies received or for which the Acquired Companies applied pursuant to any COVID-19 Measures, including any Paycheck Program Protection loan, Economic Stabilization Fund loan, other United States Small Business Administration loan, Provider Relief Fund or Medicare Advance Payment (collectively, the "COVID-Assistance"). All certifications, representatives, and indications made by or on behalf of the Acquired Companies to any Person, including any Governmental Body, in connection with the COVID-19 Assistance were correct and complete in all material respects and were prepared in material compliance with all applicable Legal Requirements. The Acquired Companies used the proceeds from the Provider Relief Funds solely for the purposes permitted by the CARES Act, as described in the applicable Provider Relief Fund Terms and Conditions and any DHHS guidance related thereto. The Acquired Companies have complied with, and continue to comply with, all aspects of the Provider Relief Fund program under the CARES Act, including any applicable reporting and auditing requirements. The Company and the Acquired Companies have taken commercially reasonable efforts to maintain accounting records associated with the Provider Relief Funds received by or on behalf of the Acquired Companies in compliance with the applicable Provider Relief Fund Terms and Conditions and related guidance available as of the date hereof.
- **3.18Insurance**. Section 3.18 of Seller's Disclosure Statement sets forth a complete and accurate list of all insurance policies under which any of the assets or properties of the Acquired Companies are covered or otherwise relating to the business of the Acquired Companies (the "Insurance Policies"), including policy numbers, names of insurers and type of liability or risk covered, amounts of coverage, limitations and deductions and expirations dates, and whether each Insurance Policy is claimsmade or occurrence-based. Each such policy is in full force and effect, and the Acquired Companies have paid or accrued (to the extent not due and payable) all premiums due, and has otherwise performed in all material respects all of its obligations under, each such Insurance Policy. No Acquired Company is in breach or default under any Insurance Policy. To the Knowledge of the Company, the insurance policies are sufficient for compliance with all Legal Requirements and agreements to which any Acquired Company is bound. There is no claim

pending under any Insurance Policy or bonds as to which coverage has been questioned, denied or disputed by the underwriters of such policies or bonds.

- **3.19Intellectual Property**. The sole representations and warranties of Seller Parent and Seller with respect to matters relating to intellectual property are set forth in this <u>Section 3.19</u>.
- (a) Section 3.19 of Seller's Disclosure Statement sets forth a true, complete and correct list of all (i) domain names of the Acquired Companies, fictional business names, trade names, registered trademarks, service marks and applications for the same of the Acquired Companies (collectively, the "Marks"); (ii) all other registered Intellectual Property owned by or exclusively licensed to the Acquired Companies; and (iii) all material unregistered Intellectual Property owned by the Acquired Companies or used in connection with the business of the Acquired Companies (collectively, the Marks and clauses (ii) and (iii), the "Owned Intellectual Property"). For each registered Mark, Section 3.19 of Seller's Disclosure Statement lists the registration number and the jurisdiction of registration beside each Mark. No Owned Intellectual Property has been or is now involved in any opposition, invalidation or cancellation, and no Acquired Company nor the operation of such Acquired Company's business is currently infringing, misappropriating, diluting or otherwise violating, and no Acquired Company has infringed, misappropriated, diluted or otherwise violated, the Intellectual Property rights of any other Person. No Acquired Company has received any written notice, demand, or claim with respect to any actual or alleged infringement, misappropriation, misuse, or violation of the Intellectual Property rights of a Person. To the Knowledge of Seller, no other Person is in conflict with or in violation or infringement of or misappropriating, the Owned Intellectual Property.
- (b) The Acquired Companies, taken as a whole, exclusively own all right, title and interest in and to Owned Intellectual Property, free and clear of all Encumbrances (other than Permitted Encumbrances or Encumbrances securing Debt being paid off at Closing). All Owned Intellectual Property that is registered with a Governmental Body is subsisting and valid and enforceable. The Acquired Companies have the right to use any third party Intellectual Property or Software that is used in or necessary for the business of the Acquired Companies as currently conducted by the Acquired Companies as of the date hereof. The Company Intellectual Property constitutes all of the Intellectual Property used in or necessary for the conduct of the business of the Acquired Companies as conducted as of the Closing Date.
- (c) The computers, computer networks, information technology systems, software, platforms, data processing systems, software, computer hardware, and communication and storage systems of the Acquired Companies (collectively, the "Company Systems") are sufficient for the needs of the businesses of the Acquired Companies as currently conducted. To the Knowledge of Seller, the Company Systems are free from material bugs and defects and the Company Systems have not suffered any material malfunction, failure, or other adverse event that has not been remediated or replaced in all material respects. All Company Systems are free of, and the Acquired Companies have taken commercially reasonable steps to prevent the introduction of, any virus, malware, spyware or other device or code that could reasonably be expected to disrupt, disable, or otherwise impair the normal operation of, or provide unauthorized access to any Company Systems.

- (d) Each member of the Acquired Companies has implemented and maintained commercially reasonable security and data protection, information security and privacy policies and procedures to safeguard, and maintain the confidentiality, integrity, and security of all Company Systems, Software, confidential information, trade secrets, products and services of the Acquired Companies, and all information, data, and transactions stored or contained therein or transmitted thereby, including Personal Information, financial information, and other sensitive information (as such information or terms are defined and/or regulated under applicable Law) against any unauthorized or improper use, disclosure, access, transmittal, interruption, modification, or corruption. During the Lookback Period, there have been no unauthorized intrusions or breaches of the security of the Company Systems or Personal Information.
- (e) The Acquired Companies have at all times complied in all material respects with the Acquired Companies' privacy policy published or made available to data subjects from which the Acquired Companies collect Personal Information or other data and such privacy policies are in compliance in all material respects with applicable Data Privacy and Security Laws. Neither the execution, delivery or performance by Seller Parent, Seller, and the Acquired Companies of this Agreement or any of the other agreements referred to in this Agreement nor the consummation by Seller Parent, Seller, and the Acquired Companies of any of the transactions contemplated hereby will result in any violation of the Acquired Companies' privacy policies or Data Privacy and Security Laws.
- **3.20Transactions with Related Persons**. Except as set forth on <u>Section 3.20</u> of Seller's Disclosure Statement, there are no outstanding amounts payable to or receivable from, or advances by any Acquired Company to, and no Acquired Company is otherwise a creditor or debtor to or party to any Contract with, any Related Person of any Acquired Company, other than part of the normal and customary terms of such persons' employment or service as an officer, employee, or director with an Acquired Company. Except as set forth on <u>Section 3.20</u> of Seller's Disclosure Statement, no Related Person owns any interest in any property owned by any Acquired Company or has engaged in any transaction with any Acquired Company within the last twelve (12) months other than part of the normal and customary terms of such person's employment.
- **3.21Environmental Liability**. The sole and exclusive representations and warranties of Seller Parent and Seller with respect to Environmental Law matters are set forth in this <u>Section 3.21</u>.
- (a) There are no Proceedings, claims, causes of action, investigations or remediation activities of any nature seeking to impose on the Acquired Companies, or that would reasonably be expected to result in the imposition on the Acquired Companies of, any liability or obligation arising under any Environmental Law or relating to Hazardous Materials pending or, to the Knowledge of Seller, threatened against the Acquired Companies, and to the Knowledge of Seller there is no reasonable basis for any such Proceeding, claim, cause of action or investigation, or remediation activity.
- (b) The Acquired Companies are, and at all times during the Lookback Period have been, in compliance in all material respects with all applicable Environmental Law, including holding and complying with all Governmental Authorizations required under applicable Environmental Law for their respective operations.

- (c) There has been no Release of, and there is no Threat of Release of, any Hazardous Material by the Acquired Companies at any location or by any Person at, on, under, to, or from any real property owned, operated or otherwise used by the Company, including any Leased Real Property.
- **3.22Healthcare Law Matters**. The sole and exclusive representations and warranties of Seller Parent and Seller with respect to Healthcare Law matters are set forth in this <u>Section 3.22</u>.
- (a) The Acquired Companies are and have been at all times during the Lookback Period in material compliance with all applicable Healthcare Laws. In accordance with 42 C.F.R. § 424.550(b), the initial enrollment in the Medicare Program of each home health agency owned by the Acquired Companies or Seller will have occurred at least thirty-six (36) months prior to the Closing Date, and no such agency will have experienced a "change in majority ownership" (as such term is defined at 42 C.F.R. § 424.502) during the thirty-six (36) months preceding the Closing Date, or, if such a change will have occurred for any home health agency owned by the Acquired Companies or Seller, that the requirements of one of the exceptions set forth at 42 C.F.R. § 424.550(b)(2) will have been satisfied.
- (b) The Acquired Companies possess, and have possessed at all times during the Lookback Period, all Governmental Authorizations necessary with respect to the ownership and operation of the businesses operated by the Acquired Companies (collectively, the "**Healthcare Permits**"), a list of which is attached hereto as <u>Section 3.22(b)</u> of Seller's Disclosure Statement. No Proceeding is pending or, to the Knowledge of Seller, threatened, to revoke, suspend, cancel, terminate or otherwise adversely modify any of the Healthcare Permits. Each Healthcare Permit is valid and in full force and effect. Subject to the parties' receipt of those filings, consents and approvals noted on <u>Section 3.22(b)</u> of Seller's Disclosure Statement, none of the Healthcare Permits will lapse, terminate, expire or otherwise be impaired as a result of the performance of this Agreement or the consummation of the transactions contemplated hereby.
- (c) Except as disclosed in <u>Section 3.22(c)</u> of Seller's Disclosure Statement, during the Lookback Period no Governmental Body has conducted or given the Acquired Companies written notice that it intends to conduct any audit or other review of the Acquired Companies' participation in the Medicare or Medicaid programs, excepting audits and surveys for compliance done in the ordinary course of business to determine compliance with Medicare or Medicaid conditions of participation that did not result in a corrective action plan, monetary penalty or assessment, or disciplinary action.
- (d) During the Lookback Period, the Acquired Companies and their respective directors, officers, owners, members, managers, employees, contractors or to Seller's Knowledge, agents (with respect to contractors and agents, during or including any time in which they were providing services for or to the Acquired Companies), have not (i) been excluded, suspended, debarred, precluded, or otherwise prohibited from participation in any Government Program, nor, is any such exclusion, debarment, suspension, preclusion, or prohibition pending or threatened; (ii) been convicted of, charged with or investigated for a Government Program or Third-Party Payor program related offense or a violation of any Legal Requirement related to fraud, theft, embezzlement, breach of fiduciary responsibility, financial misconduct, obstruction of an investigation, or controlled substances; (iii) been subject to any order or consent decree of, or

criminal or civil fine or penalty relating to any Government Program imposed by or on behalf of any Governmental Body, (iv) been listed on the General Services Administration published list of parties excluded from federal procurement programs and non-procurement programs or the "list of Excluded Individuals/Entities" on the website of the United States Department of Health and Human Services Office of Inspector General ("HHS-OIG"), or (v) been designated a Specially Designated National or Blocked Person by the Office of Foreign Asset Control of the U.S. Department of Treasury.

- (e) Except as disclosed in <u>Section 3.22(e)</u> of Seller's Disclosure Statement, neither the Acquired Companies nor any of their respective officers, directors, employees, or, to Seller's Knowledge, any contractors (when they were providing services for or to the Acquired Companies), (a) are or have been during the Lookback Period a party to any material corrective action plan, any Corporate Integrity Agreement, deferred prosecution agreement, settlement agreement or similar undertaking with or imposed by a Governmental Body (including without limitation the HHS-OIG), (b) have (or during the Lookback Period have had) reporting obligations pursuant to any settlement agreement entered into with any Governmental Body, (c) have made any voluntary or other self-disclosure to any Governmental Body of any potential or actual noncompliance with any Healthcare Law or Healthcare Permit (excluding the making of any repayment or refunds in the ordinary course of business), or (d) to the Knowledge of the Seller is the subject of or a defendant in any unsealed *qui tam* proceeding.
- (f) The Acquired Companies, as applicable, are, and have been to the extent required throughout the Lookback Period, certified for participation in, and party to, such provider agreements for payment by Government Program and Third-Party Payor program as disclosed in <a href="Section 3.22(f">Section 3.22(f</a>) of Seller's Disclosure Statement for the provision of home health, hospice, and private duty nursing services, as applicable, and such provider agreements are current and valid. All associated provider numbers for such Government Programs and Third-Party Payor programs are listed in <a href="Section 3.22(f">Section 3.22(f</a>) of Seller's Disclosure Statement. Each Acquired Company is in good standing in each Government Program and Third-Party Payor program in which it participates. With respect to each contract with a Government Program and Third-Party Payor program to which any Acquired Company is a party, throughout the Lookback Period, such Acquired Company has materially complied with all applicable requirements, including conditions of payment or participation, and are not in material breach of or material default under any such contract. No event has occurred, or, to the Knowledge of the Seller, is pending or has been threatened, which, after the giving of notice, lapse of time or otherwise, would constitute a breach or default by any Acquired Company under any such contract. No Acquired Company has received any notice that there is any investigation, audit, claim, review, inquiry or proceeding pending or threatened against any of the Acquired Companies that would reasonably be expected to result in a revocation, suspension, termination, probation, or substantial restriction of any participation by such Acquired Company in any Third-Party Payor program.
- (g) During the Lookback Period, services billed by each Acquired Company to a Government Program or Third-Party Payor have been supported in all material respects by documentation maintained in accordance with such Acquired Company's applicable policies and procedures, consistent with industry standards and requirements of the applicable Government Program or Third-Party Payor such that no failure to comply has resulted in or would reasonably be expected to result in a recoupment or required repayment, criminal penalty, civil penalty, fine

or act of or against such Company. During the Lookback Period, claims submitted to any Government Program or Third-Party Payor by, and all billing practices (including, without limitation, billing, coding, filing, and claims practices, and the related reports and filings, including cost and cap reports) of, each Acquired Company have been filed in compliance in all material respects with all Healthcare Laws applicable to such Government Program and each Acquired Company has paid or caused to be paid all known and undisputed refunds, penalties, overpayments, or other financial assessments which have become due to any Government Program or Third-Party Payor in compliance with the repayment requirements set forth by such Government Program or Third-Party Payor. Except as disclosed in Section 3.22(g) of Seller's Disclosure Statement, no Acquired Company has any reimbursement, payment or payment rate appeals, disputes or contested positions pending before any Governmental Body and, to the Knowledge of the Seller, none are threatened. Each Acquired Company has filed all cost reports required to be filed under the Medicare Program and the Medicaid Program during the Lookback Period. During the Lookback Period, no Acquired Company has received notice of audits or actual, alleged, possible or potential violations with respect to any cost or cap reports, and to the Knowledge of the Seller, no such audits or actual, alleged, possible or potential violations are threatened. Except as set forth in Section 3.22(g) of Seller's Disclosure Statement, during the Lookback Period, no deficiency in any such billing practices, including claims for overpayments, take-backs, prepayment denials, setoff or recoupments, or deficiencies for late filings, has been asserted or to the Seller's Knowledge threatened by any Governmental Body or any other Third-Party Payor and there is no basis for any such claims or deficiencies.

- (h) During the Lookback Period, no Acquired Company nor any of their respective officers, member, directors, manager or employees, or contractors or, to Seller's Knowledge, any agents (with respect to contractors and agents, during or including any time in which they were providing services to the Acquired Companies) has, in violation of any applicable Healthcare Laws, directly or indirectly, (a) offered to pay, paid or agreed to pay any remuneration, in cash or in kind, to, or made any financial arrangements with, any potential, current or former patients or payors, potential, current or former suppliers, potential, current or former clients, contractors or third party payors of the Acquired Company or any other Person in order to obtain business or payments from any such Person, (b) made or agreed to make, any contribution, payment or gift of funds or property to, or for the private use of, any governmental official, or employee or (c) paid or offered to pay any illegal remuneration for any referral to the Acquired Company or received or, solicited any illegal remuneration to refer any client, patient, payor or any other Person to the Acquired Company or any healthcare provider in violation of any applicable Healthcare Laws.
- (i) Each physician, physician's assistant, nurse practitioner, nurse, mid-level provider, technician, assistant, research coordinator, data coordinator or other clinical staff or healthcare professional employed or contracted by or otherwise providing healthcare services on behalf of any Acquired Company is duly licensed (to the extent required) to practice his or her profession in the state(s) in which such services are being provided and is validly registered (to the extent required) with the United States Drug Enforcement Administration under Title II of the Comprehensive Drug Abuse Prevention and Control Act of 1970, 21 U.S.C. § 801, et seq. (commonly known as the Controlled Substances Act) and with any comparable state controlled substances registration requirements to the extent required by law. Throughout the Lookback Period, to the Seller's Knowledge, each such personnel has complied in all material respects with

all Healthcare Laws relating to the services rendered by him or her on behalf of the Acquired Companies. Throughout the Lookback Period, to the Seller's Knowledge, no such personnel (while he or she was providing services to or for the Acquired Companies) has (i) had any professional license, Drug Enforcement Agency number (if applicable and to the extent required), Medicare, Medicaid or TRICARE provider number suspended or revoked, or (ii) been reprimanded, sanctioned or disciplined by any state licensing board or any Governmental Body, hospital, Third-Party Payor or specialty board. Each Acquired Company has (x) verified, or in the case of independent contractors, required verification from such independent contractors with respect to, the credentials of employees and independent contractors providing patient care or other services reimbursable under Government Programs, (y) conducted criminal background checks on all such employees and required its independent contractors to conduct such checks in its related independent contractor agreements, and (z) screened all such employees under the List of Excluded Individuals/Entities maintained by the HHS-OIG and the System for Award Management's excluded parties data and required its independent contractors to conduct such screenings in its related independent contractor agreements. During the Lookback Period no event has occurred, and no fact, circumstance or condition exists, that has resulted in the denial, loss, restriction, revocation or rescission of any such professional license or registration of such personnel. No such personnel (i) is subject to a pending disciplinary proceeding, inquiry, or monitoring or (ii) to Seller's Knowledge, is subject to investigation nor is such proceeding, inquiry or monitoring threatened, in any case under the bylaws or rules of procedure of any Acquired Company, the Drug Enforcement Administration, or any other federal or state professional board or agency charged with regulating the professional activities of such Person or any other Healthcare Law. There is no action pending or threatened with respect to denial, suspension or revocation of medical staff privileges or licensure of any healthcare professional employed with or contracted by any Acquired Company.

- (j) Since the Lookback Period, each Acquired Company has maintained a compliance program that meets all applicable Legal Requirements and compliance program guidance issued by the HHS-OIG, designed to promote compliance with applicable Legal Requirements and ethical standards, to improve the quality and performance of operations, and to detect, prevent, and address violations of legal or ethical standards applicable to its operations (the "Compliance Program"). A correct and complete copy of the Acquired Companies' current Compliance Program has been made available to Buyer, and each Acquired Company is, and has been, in compliance with such program at all times since the Lookback Period.
- (k) The Acquired Companies comply, and have throughout the Lookback Period complied, in all material respects with HIPAA. The Acquired Companies have in place, and comply and have complied with, written policies to protect the security and privacy of Protected Health Information (as defined in 45 CFR §160.103), and such policies comply in all material respects with HIPAA, including with respect to the Acquired Companies' uses and disclosures of, security protections for, and individual rights regarding, Protected Health Information. The Acquired Companies have conducted all security management processes required by HIPAA, including a risk assessment and information system activity review. The Acquired Companies have in effect with each entity acting as a Business Associate (as defined in HIPAA) of such Acquired Company an agreement that satisfies all of the requirements of 45 CFR §§ 164.504(e) and 164.314(a). When acting as a Business Associate, the Acquired Companies have in effect with the entity on whose behalf such Acquired Company creates, receives, maintains, or transmits Protected

Health Information an agreement that satisfies all of the requirements of 45 CFR §§ 164.504(e) and 164.314(a), and the Acquired Companies are and have complied in all material respects with such agreements. The Acquired Companies maintain, and have maintained during the Lookback Period, administrative, technical, and physical safeguards designed to protect Protected Health Information from unauthorized access or disclosure that are commercially reasonable and, in any event, in material compliance with HIPAA. During the Lookback Period, (x) no Acquired Company has (i) received any claim, information request or notice from any Governmental Body alleging or referencing a violation of HIPAA or the investigation of any breach or the improper use, disclosure or access to any Protected Health Information in its possession, custody or control; or (ii) received any written communication from any Person or Governmental Body alleging that such Company is not in compliance with HIPAA, and (y) without limitation of the foregoing, there has been no unauthorized access, use, disclosure or transfer of any Protected Health Information in the possession, custody or control of the Acquired Companies or a contractor acting on behalf of the Acquired Companies requiring notice to any affected individual or Governmental Body under any applicable Legal Requirement. During the Lookback Period, the Acquired Companies have not experienced any (i) Breach of Unsecured Protected Health Information (as "Breach," "Unsecured Protected Health Information," and "Protected Health Information" are defined by HIPAA), or (ii) any Security Incident (as "Security Incident" is defined by HIPAA). The Acquired Companies have been and are in compliance with, and have since the Lookback Period maintained any been in compliance with, any consents, authorizations, waiver of authorization or other permission pursuant to which the Acquired Companies access, use, disclose, or have accessed, used or disclosed, Protected Health Information ("Privacy Consents"). Neither the execution, delivery, or performance of this Agreement, nor the consummation of any of the transactions contemplated by this Agreement, including any direct or indirect transfer of Protected Health Information resulting from such transactions, will violate HIPAA, any policies, or Privacy Consents of the Acquired Companies as such currently exist or as existed at any time during which any of such Protected Health Information or customer information was collected or obtained.

**3.23Directors and Officers**. Section 3.23 of Seller's Disclosure Statement lists each of the directors and officers of each of the Acquired Companies.

#### 3.24 Material Contracted Care Partners, Suppliers and Payors.

- (a) <u>Section 3.24(a)</u> of Seller's Disclosure Statement sets forth a true and complete list of the 15 largest contracted care partners of the Acquired Companies providing hospice services, measured by the total amount of Acquired Company hospice patients residing at such contracted care partner locations during the most recently completed fiscal year of the Acquired Companies (each, a "<u>Material Contracted Care Partner</u>"), and the number of such Acquired Company hospice patients residing at each such Material Contracted Care Partner. No Material Contracted Care Partner has terminated, cancelled, adversely modified, or failed to renew, or given any Acquired Company notice that references its intention to terminate, cancel, adversely modify, or fail to renew, its business relationship with any Acquired Company (whether or not subject to a contract).
- (b) <u>Section 3.24(b)</u> of Seller's Disclosure Statement sets forth a true and complete list of the 15 largest suppliers of the Acquired Companies (*i.e.*, which provide the Acquired Companies with products or services), measured by the aggregate amount of sales made by such

suppliers to the Acquired Companies during the most recently completed fiscal year of the Acquired Companies (each, a "<u>Material Suppliers</u>"), and the aggregate amount of such sales made by each such Material Supplier during such fiscal year. No Material Supplier has terminated, cancelled, adversely modified, or failed to renew, or given any Acquired Company notice that references its intention to terminate, cancel, adversely modify, or fail to renew, its business relationship with any Acquired Company (whether or not subject to a contract).

- (c) <u>Section 3.24(c)</u> of Seller's Disclosure Statement sets forth a true and complete list of the 15 largest payors of the Acquired Companies, measured by the aggregate amount of revenue recognized by the Acquired Companies during the most recently completed fiscal year of the Acquired Companies (each, a "<u>Material Payors</u>"), and the aggregate amount of revenues recognized by the Acquired Companies from each such Material Payor during such fiscal year. No Material Payor has terminated, cancelled, adversely modified, or failed to renew, or given any Acquired Company notice that references its intention to terminate, cancel, adversely modify, or fail to renew, its business relationship with any Acquired Company (whether or not subject to a contract).
- **3.25No Further Representations**. Except for the representations and warranties set forth in <u>Article 3</u> of this Agreement, each of Seller Parent and Seller does not make, and shall not be deemed to have made, any representation or warranty to Buyer, express or implied, with respect to the Acquired Companies, their respective business or assets, the Contemplated Transactions or otherwise, and each of Seller Parent and Seller hereby disclaims any such representation or warranty not set forth in <u>Article 3</u> of this Agreement, including any representation or warranty as to the accuracy or completeness of any information regarding the Acquired Companies furnished or made available to Buyer or its Representatives, including any confidential information presentation prepared or distributed by Houlihan Lokey, and any information, documents, or material made available to Buyer in the data room or in any other form in expectation of the Contemplated Transactions or as to the condition, value, or quality of the business of the Acquired Companies or their respective assets.

#### 4. REPRESENTATIONS AND WARRANTIES OF BUYER

Except as disclosed in Buyer's Disclosure Statement attached hereto, Buyer represents and warrants to Seller Parent and Seller as follows:

**4.10rganization and Good Standing**. Buyer (i) is a corporation duly organized, validly existing, and in good standing under the laws of the State of Illinois; (ii) is duly qualified to do business and in good standing in all jurisdictions (whether federal, state, local or foreign) where its ownership or leasing of property or the conduct of its business requires it to be so qualified; and (iii) has all requisite corporate power and authority to own or lease its properties and assets and to carry on its business as now conducted, including performing all of its obligations under this Agreement and the Buyer's Closing Documents.

# 4.2Enforceability and Authority; No Conflict.

(a) Buyer represents and warrants, that (i) Buyer has the corporate power and authority to execute and deliver this Agreement and to consummate the Contemplated

Transactions; (ii) the execution and delivery of this Agreement and the consummation of the Contemplated Transactions have been duly and validly approved by the equity owners of Buyer to the extent required under applicable Legal Requirements or pursuant to the Organizational Documents of Buyer; and (iii) this Agreement has been duly and validly executed and delivered by Buyer and (assuming due authorization, execution and delivery by Seller Parent, Seller and the Acquired Companies) constitutes valid and binding obligations of Buyer, enforceable against Buyer in accordance with its terms (except as may be limited by bankruptcy, insolvency, moratorium, reorganization or similar laws affecting the rights of creditors generally and the availability of equitable remedies).

- (b) Neither the execution and delivery of this Agreement nor the consummation or performance of any Contemplated Transactions will directly or indirectly (with or without notice or lapse of time): (i) contravene, conflict with, or violate (A) any Organizational Document of Buyer, or (B) any resolution adopted by the board of directors (or other similar body) or the owners of Buyer; (ii) contravene, conflict with, or violate, or give any Governmental Body or other Person the right to challenge any Contemplated Transactions, or to exercise any remedy or obtain any relief under, any Legal Requirement or any Order to which Buyer, or any assets owned or used by Buyer, is subject; or (iii) breach, or give any Person the right to declare a default or exercise any remedy or to obtain any additional rights under, or to accelerate the maturity or performance of, or payment under, or to cancel, terminate, or modify, any Contract to which Buyer is a party.
- (c) Except for the approval of the board of directors of Addus HomeCare Corporation, which has been obtained prior to the date hereof, Buyer is not required to give notice to or obtain Consent from any Person in connection with the execution and delivery of this Agreement or the consummation or performance of any Contemplated Transactions.
- **4.3Investment Intent**. Buyer is acquiring the Membership Interests for its own account and not with a view to their distribution within the meaning of Section 2(11) of the Securities Act.
- **4.4Certain Proceedings**. There is no Proceeding pending against Buyer that challenges, or could have the effect of preventing, delaying, making illegal, imposing limitations or conditions on, or otherwise interfering with, any Contemplated Transactions. To Buyer's Knowledge, no such Proceeding has been threatened.
- **4.5Brokers or Finders**. Neither Buyer nor any of its Representatives has incurred any obligation or liability, contingent or otherwise, for any brokerage or finder's fee, agent's commission, or other similar payment in connection with this Agreement or the Contemplated Transactions.
- **4.6Due Diligence.** Buyer has conducted such investigations of the Acquired Companies as it deems necessary and appropriate and has received all of the information that it has requested from the Seller Parent, Seller, and the Acquired Companies in connection with the execution and delivery of this Agreement and the consummation of the Contemplated Transactions.

**4.7Non-reliance of Buyer**. Except for the specific representations and warranties expressly made by Seller Parent and Seller in Article 3 of this Agreement or in the Seller's Closing Documents, (a) Buyer represents and warrants that (i) the Seller Parent and Seller are not making and have not made any representation or warranty, expressed or implied, at law or in equity, in respect of the Seller Parent, Seller, or any of the Acquired Companies, or any of the Seller Parent's, Seller's, or the Acquired Companies' respective businesses, assets, liabilities, operations, prospects, or condition (financial or otherwise), including with respect to merchantability or fitness for any particular purpose of any assets, the nature or extent of any liabilities, the prospects of the business, the effectiveness or the success of any operations, or the accuracy or completeness of any confidential information memoranda, documents, projections, material or other information (financial or otherwise) regarding the Seller Parent, Seller, or the Acquired Companies furnished to Buyer or its Representatives or made available to Buyer or its Representatives in any "data rooms," "virtual data rooms," management presentations or in any other form in expectation of, or in connection with, the Contemplated Transactions, or in respect of any other matter or thing whatsoever, and (ii) neither the Seller Parent, Seller, nor any of their Representative or any of the Acquired Companies has any authority, express or implied, to make any representations, warranties or agreements not specifically set forth in this Agreement or in the Seller's Closing Documents and subject to the limited remedies herein provided, (b) the Buyer specifically disclaims that it is relying upon or has relied upon any such other representations or warranties that may have been made by any Person and not set forth in this Agreement or the Seller's Closing Documents, and acknowledges and agrees that the Seller Parent, Seller, and the Acquired Companies and their respective Affiliates have specifically disclaimed and do hereby specifically disclaim any such other representation or warranty made by any Person, (c) Buyer specifically disclaims any obligation or duty by the Seller Parent, Seller, or any Acquired Company or any of their respective Affiliates to make any disclosures of fact not required to be disclosed pursuant to the specific representations and warranties set forth in <u>Article 3</u> of this Agreement, and (d) Buyer is entering into the Contemplated Transactions and acquiring the Membership Interests in connection with the Contemplated Transactions subject only to the specific representations and warranties set forth in Article 3 of this Agreement.

**4.8Sufficiency of Funds**. Buyer has sufficient cash on hand or other sources of immediately available funds to enable it to make payment of the Purchase Price and consummate the Contemplated Transactions.

**4.9R&W Binder**. Subject only to the execution and delivery of this Agreement, the R&W Binder is in full force and effect (provided that coverage thereunder is subject to the conditions to the insurer's obligations thereunder as set forth therein). Neither Buyer, nor to the knowledge of Buyer, the other party to R&W Binder is in default or breach of the R&W Binder.

#### 5. COVENANTS OF SELLER PRIOR TO CLOSING DATE

**5.1Access and Investigation**. After execution of this Agreement and prior to the Closing Date, and upon reasonable notice from Buyer, Seller Parent and Seller shall, and shall cause each Acquired Company to: (i) furnish Buyer and its Representatives reasonable access during normal business hours to its key employees and books and records and such additional information as Buyer may reasonably request, and (ii) to otherwise cooperate with and assist Buyer

to the extent reasonably necessary to obtain applicable Consents of any Governmental Body or third party and to implement the orderly Closing of the Contemplated Transactions.

- **5.2Operation of the Businesses of the Acquired Companies**. Except for the transactions contemplated by the Pre-Closing Reorganization, after execution of this Agreement and prior to the Closing Date, Seller Parent and Seller shall, and shall cause each Acquired Company (i) not to take any action listed in Section 3.7 hereof and (ii) to take the following actions:
  - (a) conduct the business of Seller and such Acquired Company only in the Ordinary Course of Business;
- (b) preserve intact the current business organization of such Acquired Company, use its commercially reasonable efforts to keep available the services of the officers, employees, and agents of such Acquired Company, and maintain its relations and goodwill with customers, landlords, creditors, employees, agents, and others having business relationships with such Acquired Company;
- (c) maintain the assets owned or used by such Acquired Company in a state of repair and condition that complies with Legal Requirements and Contracts and is consistent with the requirements and normal conduct of the business of such Acquired Company;
- (d) keep in full force and effect, without amendment, all material rights relating to the business of such Acquired Company;
  - (e) comply in all material respects with all Legal Requirements applicable to such Acquired Company;
  - (f) refrain from taking any actions described in Section 3.7(g); and
  - (g) maintain all records of such Acquired Company consistent with past practice.

**5.3Filings and Notifications; Cooperation.** As promptly as practicable after the date of this Agreement, and in any event within the applicable time period prescribed by Legal Requirements, Seller Parent and Seller shall, and shall cause each Acquired Company and each of their Related Persons to, (y) obtain all consents required by this Agreement and to make all filings and notifications required by Legal Requirements to be made by them in connection with the Contemplated Transactions (including all filings under the HSR Act); and (z) promptly notify Buyer in writing of any notice or other communication from any Governmental Body in connection with the transactions contemplated by this Agreement. Seller Parent and Seller shall, and shall cause each Acquired Company and each of their Related Persons to, cooperate with, and timely respond to requests from, Buyer, its Related Persons, and their respective Representatives, (a) with respect to all filings and notifications that Buyer or its Related Persons elect to make or shall be required by Legal Requirements to make in connection with the Contemplated Transactions; (b) in identifying and obtaining the Governmental Authorizations required by Buyer to own and operate each Acquired Company from and after the Closing Date; and (c) in obtaining all Consents identified in Section 9.4 of Seller's Disclosure Statement (including taking all actions requested by Buyer to cause early termination of any applicable waiting period under the HSR Act). Seller Parent and Seller shall and shall cause each Acquired Company to use commercially

reasonable efforts to do all things necessary to cause the conditions set forth in Article 8 to be satisfied.

**5.4Exclusive Dealing**. After execution of this Agreement and until this Agreement shall have been terminated pursuant to Section 10.1, Seller Parent and Seller shall not, and Seller Parent and Seller shall cause each Acquired Company, Seller Parent and Seller's other Related Persons, and each of their respective Representatives not to, directly or indirectly, solicit, initiate, encourage, or entertain any inquiries or proposals from, discuss or negotiate with, provide any nonpublic information to, consider the merits of any inquiries or sale proposals from any Person (other than Buyer), or enter into any Contract or non-binding agreement, relating to any business combination transaction involving Seller or an Acquired Company, however structured, including the direct or indirect (including via a transaction involving a Related Person) sale of the business or assets (other than in the Ordinary Course of Business) of any Acquired Company, any Equity Security of any Acquired Company, or any merger, consolidation, or similar transaction or arrangement.

**5.5Financial Information**. After execution of this Agreement, Seller Parent and Seller shall deliver to Buyer within thirty (30) days after the end of each month a copy of each Acquired Company's financial statements for such month prepared in a manner and containing information consistent with such Acquired Company's historical past practices consistently applied. For the avoidance of doubt, Seller Parent and Seller makes no representation or warranty with respect to such financial statements.

**5.6401(k) Plan.** Seller Parent shall remove the Acquired Companies as participating employers in the AHC 401(k) Profit Sharing Plan (the "401(k) Plan") immediately prior to the Closing Date and in connection therewith fully vest the 401(k) Plan accounts of all participants who are employed by one of the Acquired Companies and provide for the distribution of such accounts pursuant to applicable Legal Requirements. Seller or Seller Parent shall deliver to Buyer at Closing duly executed resolutions of the Board of Directors of Seller Parent reflecting the actions discussed in this <u>Section 5.6</u>.

#### 6. COVENANTS OF BUYER PRIOR TO CLOSING DATE

**6.1Filings and Notifications; Cooperation.** As promptly as practicable after the date of this Agreement, and in any event within the applicable time period prescribed by Legal Requirements, Buyer shall, and shall cause each of its Related Persons to, make all filings and notifications required by Legal Requirements to be made by it in connection with the Contemplated Transactions (including all filings under the HSR Act). Buyer shall, and shall cause each of its Related Persons to, cooperate with, and timely respond to requests from, Seller Parent, Seller, each Acquired Company, their Related Persons and their respective Representatives, (a) with respect to all filings and notifications that Seller Parent, Seller, any Acquired Company, or their Related Persons shall be required by Legal Requirements to make in connection with the Contemplated Transactions; and (b) in obtaining all Material Consents; provided, however, that Buyer shall not be required to dispose of or make any change to its business or incur any other material post-Closing obligation in order to comply with this Section 6.1. Buyer agrees not to contact any employee of the Acquired Companies, except after having first obtained written consent from Seller Parent or Seller (which may be in the form of email).

**6.2R&W Binder**. Immediately after the execution and delivery of this Agreement, Buyer shall cause the insurer and underwriter of the R&W Policy to bind the R&W Policy and Buyer covenants and agrees to take all action necessary to cause the issuance of the R&W Policy as soon as practicable thereafter, including, without limitation, timely paying all premiums, underwriting costs and other charges associated therewith. Buyer acknowledges and agrees that the R&W Policy shall not be amended or modified from that set forth on Exhibit B in any manner that is or would reasonably be materially adverse to the Seller Parent or Seller without the prior written consent of the Seller Parent or Seller, in each of their sole discretion. The R&W Policy shall provide that the insurance carriers shall not have any rights of subrogation, contribution against or with respect to the Seller Parent, Seller or any of their Affiliates, except in connection with Fraud by Seller Parent or Seller. Further, the R&W Policy shall not provide for any other rights by Buyer or the insurance carriers against the Seller Parent or Seller. Notwithstanding anything to the contrary contained in this Agreement, (a) neither Buyer nor any of its Affiliates or their respective Representatives shall have recourse to the Seller Parent or Seller for Losses as a result of, arising out of, or related to any breach of or inaccuracy in any representation or warranty of the Seller Parent or Seller in this Agreement (or in any certificate delivered hereunder), except in the case of Fraud by the Seller Parent or Seller, and (b) the sole and exclusive remedy and recourse of Buyer, any of its Affiliates and their respective Representatives for Losses as a result of, arising out of, or related to any breach of or inaccuracy in any representation or warranty of the Seller Parent or Seller in this Agreement (or in any certificate delivered hereunder) shall be to recover under the R&W Policy, except in the case of Fraud by the Seller Parent or Seller. From and after the date hereof, Buyer shall not (and shall cause its Affiliates to not) grant any right of subrogation or otherwise amend, modify, or otherwise change, terminate, or waive any term or condition of the R&W Policy in a manner inconsistent with this Section 6.2, or in a manner reasonably expected to be adverse to the Seller Parent, Seller, its Affiliates, or its respective Representatives, as applicable, without the prior written consent of the Seller Parent and Seller. Buyer shall pay or cause to be paid all costs and expenses related to the R&W Policy as they come due, including the total premium, underwriting costs, taxes, brokerage commission, and other fees and expenses of the R&W Policy. Buyer shall also satisfy on a timely basis all conditions necessary for the issuance of or continuance of coverage under the R&W Policy. From and after the Closing, Buyer shall provide the Seller Parent and Seller with written notice of any claim made against the R&W Policy. Notwithstanding the foregoing, for the avoidance of doubt, the Parties acknowledge and agree that the obtaining of the R&W Policy is not a condition to the Closing, and Buyer shall remain obligated, subject only to the satisfaction or waiver of the conditions set forth in Article 8 to consummate the transactions contemplated by this Agreement.

#### 7. POST-CLOSING COVENANTS

#### 7.1Cooperation and Proceedings; Access to Records.

(a) After the Closing, each of Seller Parent and Seller shall, and shall cause its Related Persons to, cooperate with Buyer and its counsel and make itself and their Representatives available to Buyer and the Acquired Companies in connection with the institution or defense of any Proceeding involving or relating to any Acquired Company, including providing testimony, Records, and other information, all at Buyer's expense.

(b) For a period of six (6) years after the Closing Date, each of Seller Parent and Seller and Buyer will (and Seller Parent and/or Seller shall cause its Related Persons to) make available to each other any Records in the non-requesting party's custody or control for the purpose of preparing any financial statement or Tax Return or preparing for or defending any tax-related examination of the requesting party or any Acquired Company by any Governmental Body. The party requesting such Records will reimburse the non-requesting party for the reasonable out-of-pocket costs and expenses incurred by the non-requesting party. The non-requesting party will afford access to such Records during normal business hours, upon reasonable advance notice given by the requesting party, and subject to such reasonable limitations as the non-requesting party may impose to delete competitively sensitive or privileged information.

#### 7.2Insurance.

- (a) Seller Parent and Seller shall cause an endorsement to be made to each Insurance Policy naming the Acquired Companies as additional insureds under such Insurance Policies.
- (b) For three (3) years following the Closing, Seller Parent and Seller shall enable the Acquired Companies to make direct claims pertaining to pre-Closing occurrences under the Insurance Policies, and shall cause any such claims to be covered under the terms of such Insurance Policies as if the Acquired Companies were still owned by Seller Parent and Seller with retentions, limitations and terms consistent with those in place under the Insurance Policies in place as of the date of this Agreement.
- (c) Seller Parent and Seller shall not permit the exercise of any rights of subrogation, contribution or other rights against the Acquired Companies in connection with any claims made under the Insurance Policies pursuant to this <u>Section 7.2</u>.

#### 7.3Confidentiality.

- (a) As used in this <u>Section 7.3</u>, the term "**Confidential Information**" means all information concerning the business and affairs of any Acquired Company, including historical and current financial statements, financial projections and budgets, tax returns and accountants' materials, historical, current, and projected sales, capital spending budgets and plans, business plans, strategic plans, marketing and advertising plans, publications, client and customer and acquisition and transaction prospect lists and files, current and anticipated customer requirements, price lists, market studies, Contracts, the names and backgrounds of key personnel and personnel training techniques and materials, however documented.
- (b) Each of Seller Parent and Seller acknowledges the confidential and proprietary nature of the Confidential Information and agrees that each of them shall, and shall cause its Representatives to, (i) keep the Confidential Information confidential; (ii) not use the Confidential Information for any reason or purpose; and (iii) without limiting the foregoing, not disclose the Confidential Information to any Person, except as set forth in Section 7.3(e), or with Buyer's written consent.
- (c) Section 7.3(b) does not apply to that part of the Confidential Information that becomes generally available to the public other than as a result of a breach of this Section 7.3 by

Seller Parent, Seller, or a third party acting on the behalf of Seller Parent or Seller. Confidential Information shall not be deemed "generally available to the public" merely because it is included or incorporated in more general information that is publicly available or because it combines features which individually may be publicly available.

- (d) If Seller Parent or Seller becomes compelled in any Proceeding to make any disclosure that is prohibited by this Section 7.3, Seller Parent or Seller shall, to the extent legally permissible, provide Buyer with prompt notice of such compulsion so that Buyer may seek an appropriate protective order or other appropriate remedy or waive compliance with the provisions of this Section 7.3, and upon the request of Buyer, Seller Parent or Seller shall provide reasonable assistance, at Buyer sole cost and expense, in obtaining such protective order or other appropriate remedy. In the absence of a protective order or other remedy, Seller Parent or Seller may disclose that portion (and only that portion) of the Confidential Information that, based upon the reasonable opinion of Seller Parent's or Seller's counsel, Seller Parent or Seller is legally compelled to disclose; provided, however, that Seller Parent or Seller shall use its best efforts to obtain written assurance that any Person to whom any Confidential Information is so disclosed shall accord confidential treatment to such Confidential Information.
- (e) The confidentiality restrictions of this <u>Section 7.3</u> shall not apply to Seller Parent's, Seller's, or Seller's Representatives' disclosure or use of Confidential Information to the extent legally necessary to make any filing required by any Legal Requirement applicable to Seller Parent, Seller, or its Representatives; file of any Tax Return or any other required filing with a Governmental Body or respond to any proceedings involving any Tax Returns or filings with any Governmental Body; evaluate the basis for, prosecuting, or defending any Proceeding arising out of or involving this Agreement or any dispute with Buyer; or communicating with Seller Parent's or Seller's shareholders and their Representatives with respect to any of the foregoing.

#### 7.4Exculpation, Indemnification, D&O Insurance.

- (a) For a period of six (6) years after the Closing, Buyer will not, and will not permit any of the Acquired Companies to amend, repeal or modify any provision of such entity's Organizational Documents relating to the exculpation, indemnification or advancement of expenses of any officers, directors or employees of the Acquired Company (unless required by applicable Legal Requirements), it being the intent of the Parties that the officers and directors of the Acquired Companies as of the date of this Agreement and as of Closing Date will continue to be entitled to such exculpation, indemnification and advancement of expenses. Further, from and after the Closing Date, Buyer agrees to cause the Acquired Companies to comply with and honor any such rights to exculpation, indemnification and advancement; *provided* that if any claims are asserted or made within such period, all rights to indemnification and exculpation (and to advancement of expenses) in respect of such claims shall continue, without diminution, until disposition of any and all such claims.
- (b) In the event of any threatened or actual claim or Proceeding, whether civil, criminal or administrative, including, without limitation, any such claim or Proceeding in which any individual who is now, or has been at any time prior to the date of this Agreement, or who becomes prior to the Closing Date, a director, officer, or employee of any Acquired Company (the "Indemnified Parties"), is, or is threatened to be, made a party based in whole or in part on, or

arising in whole or in part out of, or pertaining to the fact that he is or was a director, officer or employee of an Acquired Company or any of their predecessors, whether in any case asserted or arising before or after the Closing Date, the Parties agree to cooperate and use their commercially reasonable efforts to defend against and respond thereto. It is understood and agreed that after the Closing Date, the Acquired Companies shall, subject to any limitations of any Legal Requirements, indemnify, defend and hold harmless each such Indemnified Party to the same extent as such Indemnified Party is entitled to be indemnified as of the date of this Agreement by the Acquired Companies pursuant to the Acquired Company's Organizational Documents in existence as of the date hereof against any Losses (including expenses in advance of the final disposition of any claim or Proceeding to each Indemnified Party upon receipt of an undertaking (in reasonable and customary form) to repay such advances if it is ultimately determined that such Indemnified Party is not entitled to indemnification hereunder) and amounts paid in settlement in connection with any such threatened or actual claim or Proceeding.

(c) In the event Buyer or the Acquired Companies or any of their successors or assigns (i) consolidates with or merges into any other person and shall not be the continuing or surviving corporation or entity of such consolidation or merger, or (ii) transfers or conveys all or substantially all of its properties and assets to any person, then, and in each such case, to the extent necessary, proper provision shall be made so that the successors and assigns of Buyer or any of the Acquired Companies, as appropriate, assume the obligations set forth in this <u>Section 7.4</u>.

The provisions of this <u>Section 7.4</u> shall survive the Closing and are intended to be for the benefit of, and shall be enforceable by, each Indemnified Party and his or her heirs and representatives.

**7.5Public Announcements**. Neither Buyer nor Seller Parent or Seller will, and each will cause its respective Affiliates not to, without the prior written consent of, in the case of Buyer, the Seller Parent and Seller, or in the case of the Seller Parent or Seller, Buyer, make any public statement or press release or any other announcement or communication to the employees, customers, or suppliers of the Acquired Companies with respect to this Agreement or the transactions contemplated hereby, or otherwise disclose to any Person the existence, terms, content, or effect of this Agreement, except to the extent that disclosure is required by a Legal Requirement or to comply with the obligations set forth in this Agreement.

**7.6COBRA Liability**. Seller Parent and its Affiliates shall be responsible for all liabilities and obligations in connection with claims for post-employment medical, vision and dental benefits that may be required under Section 4980B of the Code regarding any "M&A qualified beneficiaries" with regard to the Contemplated Transactions, as such term in defined in Treasury Regulation 54.4980B-9 Q&A4(b).

#### 8. CONDITIONS PRECEDENT TO BUYER'S OBLIGATION TO CLOSE

Buyer's obligations to purchase the Membership Interests and to take the other actions required pursuant to this Agreement to be taken by Buyer at the Closing are subject to the satisfaction, at or prior to the Closing, of each of the following conditions (any of which may be waived in whole or in part by Buyer):

- **8.1Accuracy of Seller Parent's and Seller's Representations**. The representations and warranties set forth in Section 3.1 ("Corporate Organization"), Section 3.2 ("Capitalization/Ownership of Membership Interests"), Section 3.3 ("Authority; No Violation"), Section 3.4 ("Consents and Approvals"), Section 3.6 ("Brokers or Finders"), and Section 3.11 ("Taxes and Tax Returns"), of this Agreement will be true and correct in all respects (other than *de minimis* exceptions) at and as of the Closing as if made at and as of such time (other than any representation or warranty that expressly relates to a specific date, which representations and warranties set forth in Article 3 will be true and correct (without giving effect to any limitation as to "materiality" or "Material Adverse Effect") at and as of the Closing as if made at and as of such time (other than any representation or warranty that expressly relates to a specific date, which representation and warranty will be true and correct on the date so specified), except where the failure of such representations and warranties to be so true and correct has not had a Material Adverse Effect.
- **8.2Seller Parent's and Seller's Performance**. The covenants and obligations that Seller Parent or Seller is required to perform or to comply with pursuant to this Agreement at or prior to the Closing will have been duly performed and complied with in all material respects.
- **8.3Bring Down Certificate**. Buyer will have received a certificate executed by Seller Parent and Seller confirming, (a) the accuracy of its representations and warranties as of the date of this Agreement and as of the Closing Date in accordance with Section 8.1; and (b) the performance of and compliance with its covenants and obligations to be performed or complied with at or prior to the Closing in accordance with Section 8.2.
- **8.4Consents**. Each of the Consents identified in <u>Section 8.4</u> of Seller's Disclosure Statement (the "Material Consents") will have been obtained in form and substance reasonably satisfactory to Buyer and will be in full force and effect.
- **8.5Governmental Authorizations**. Buyer will have received such Governmental Authorizations as are necessary to allow Buyer to acquire and own the Membership Interests.
- **8.6No Legal Prohibition**. There will not be in effect any Legal Requirement or Order that prohibits the sale of the Membership Interests by Seller to Buyer or the consummation of any of the other Contemplated Transactions.
- **8.7Closing Deliveries**. Seller must have caused the documents and instruments required by <u>Section 2.6(a)</u> to be delivered (or tendered subject only to the Closing) to Buyer.
- **8.8Frustration of Closing Conditions**. No Party may rely on the failure of any condition set forth in this <u>Article 8</u>, as the case may be, to refuse to consummate the transactions contemplated by this Agreement if such failure was caused by such Party's failure to comply with any provision of this Agreement.

#### 9. CONDITIONS PRECEDENT TO SELLER PARENT'S AND SELLER'S OBLIGATIONS TO CLOSE

Seller Parent and Seller's obligations to sell the Membership Interests and to take the other actions required pursuant to this Agreement to be taken by Seller Parent or Seller at the Closing, as applicable, are subject to the satisfaction, at or prior to the Closing, of each of the following conditions (any of which may be waived in whole or in part by Seller or Seller Parent):

- **9.1Accuracy of Buyer's Representations**. Each of the representations and warranties of Buyer set forth in <u>Article 4</u> of this Agreement will be true and correct in all respects at and as of the Closing as if made at and as of such time (other than any representation or warranty that expressly relates to a specific date, which representation and warranty will be true and correct as of the date so specified), except where the failure of such representations and warranties to be so true and correct has not had an adverse effect that could materially impair or delay Buyer's ability to consummate the transactions contemplated hereby.
- **9.2Buyer's Performance**. The covenants and obligations that Buyer is required to perform or to comply with pursuant to this Agreement at or prior to the Closing will have been duly performed and complied with in all material respects.
- **9.3Bring Down Certificate**. Seller will have received a certificate executed by Buyer confirming, (a) the accuracy of its representations and warranties as of the date of this Agreement and as of the Closing Date in accordance with Section 9.1; and (b) the performance of and compliance with its covenants and obligations to be performed or complied with at or prior to the Closing in accordance with Section 9.2.
- **9.4Consents**. Each of the Consents identified in <u>Section 9.4</u> of Seller's Disclosure Statement will have been obtained in form and substance satisfactory to Seller and will be in full force and effect.
- **9.5No Legal Prohibition**. There will not be in effect any Legal Requirement or Order that prohibits the sale of the Membership Interests by Seller to Buyer or the consummation of any of the other Contemplated Transactions.
- **9.6Closing Deliveries**. Buyer must have caused the documents and instruments required by  $\underline{\text{Section 2.6(b)}}$  to be delivered (or tendered subject only to the Closing) to Seller.
- **9.7Frustration of Closing Conditions**. No Party may rely on the failure of any condition set forth in this <u>Article 9</u>, as the case may be, to refuse to consummate the transactions contemplated by this Agreement if such failure was caused by such Party's failure to comply with any provision of this Agreement.

# 10. TERMINATION

**10.1Termination Events.** Subject to <u>Section 10.2</u>, by notice given prior to or at the Closing, this Agreement may be terminated as follows:

- (a) by mutual written consent of Buyer, Seller, and Seller Parent;
- (b) by Buyer if a breach of any provision of this Agreement has been committed by Seller or Seller Parent and Seller or Seller Parent has failed to cure such breach by the End Date, in each case such that the conditions set forth in Section 8.1 or Section 8.2 could not be satisfied;
- (c) by Seller or Seller Parent if a breach of any provision of this Agreement has been committed by Buyer and Buyer has failed to cure such breach by the End Date, in each case such that the conditions set forth in <u>Section 9.1</u> or <u>Section 9.2</u> could not be satisfied;
- (d) by Buyer if satisfaction of any condition in <u>Article 8</u> by October 2, 2023, or such later date as the Parties may agree upon (the "**End Date**") becomes impossible (other than through the failure of Buyer to comply with its obligations under this Agreement);
- (e) by Seller or Seller Parent if satisfaction of any condition in <u>Article 9</u> by the End Date becomes impossible (other than through the failure of Seller or Seller Parent to comply with its obligations under this Agreement);
- (f) by Buyer if the Closing has not occurred on or before the End Date, unless the failure of Buyer to perform its obligations under this Agreement has been the primary cause of the failure of Closing to occur before the End Date, and such failure constitutes a material breach of this Agreement; or
- (g) by Seller or Seller Parent if the Closing has not occurred on or before the End Date, unless the failure of Seller or Seller Parent to perform its obligations under this Agreement has been the primary cause of the failure of Closing to occur before the End Date, and such failure constitutes a material breach of this Agreement.
- **10.2Effect of Termination**. Each Party's right of termination under <u>Section 10.1</u> is in addition to any other right it may have under this Agreement (including under <u>Section 13.12</u>) or otherwise, and the exercise of a Party's right of termination will not constitute an election of remedies. If this Agreement is terminated pursuant to <u>Section 10.1</u>, this Agreement will be of no further force or effect; *provided*, *however*, that, (i) this <u>Section 10.2</u> and <u>Article 13</u> will survive the termination of this Agreement and will remain in full force and effect; and (ii) the termination of this Agreement will not relieve any Party from any liability for any willful breach of this Agreement or Fraud occurring prior to termination.

## 11. SURVIVAL; REMEDIES

11.1Survival of Representations, Warranties and Covenants. This Article 11, Section 4.6, Section 4.7, and those covenants and agreements set forth in this Agreement, Buyer's Closing Documents and Seller's Closing Documents (collectively, the "Closing Documents") that by their terms contemplate performance in whole or in part after the Closing shall survive the Closing in accordance with their respective terms. All representations and warranties and all other covenants and agreements in this Agreement and the Closing Documents shall terminate at and not survive the Closing, and there shall be no liability after the Closing in respect thereof (other than in the case of Fraud and for the benefit of the R&W Policy). Buyer acknowledges and agrees

that, except in the case of Fraud, the R&W Policy shall be the sole and exclusive remedy of Buyer and its Affiliates and their respective Representatives, successors and assigns for any breach of a representation or warranty contained in this Agreement or the Closing Documents. No exclusions or limitations applicable to the R&W Policy or any failure of the R&W Policy insurer to make any payments under the R&W Policy based on any claims asserted by Buyer will increase or alter any liabilities or obligations of Seller Parent or Seller under this Agreement. The R&W Policy will not by its terms obligate Seller or Seller Parent to indemnify the R&W Insurer, and will not allow the R&W to subrogate to the rights of Buyer herein except in the case of fraud by Seller or Seller Parent.

# 11.2Remedies; Specific Performance.

- (a) Notwithstanding anything in this Agreement to the contrary, no Party shall be deemed to have suffered any Losses under this Agreement arising from any item or matter that was included in the determination of final adjustment amounts, as it is the Parties' intent that the procedures set forth in Section 2.4(d) shall provide the sole and exclusive remedies for such items and matters and that neither Buyer, its Affiliates nor their respective Representatives, successors and assigns shall have recourse pursuant to Section 11.1 or otherwise for breaches of covenants or agreements of the Seller Parent or Seller contained in this Agreement the subject matter of which were resolved pursuant to Section 2.4(d); provided, however, that Seller and Seller Parent shall be responsible for, and shall satisfy, any remaining Seller's Expense or Debt of the Acquired Companies. Buyer covenants and agrees to use commercially reasonable efforts to mitigate any Debt relating to the obligations set forth on Section 1.1 of the Seller's Disclosure Statement, including pursuing any claims the Acquired Companies may have against the prior owners of the assets purchased whether those claims arise pursuant to contractual indemnification or breach of warranty rights, negligent misrepresentation, fraud, tort or otherwise at law or equity, provided that Seller and Seller Parent shall be responsible for any out of pocket costs (including attorney's fees) incurred by Buyer in pursuing such efforts, and any amounts collected by Buyer from such prior owners shall reduce the amount of Debt otherwise payable by the Seller and Seller Parent or shall be paid over to the Seller or Seller Parent to the extent the Seller or Seller Parent has previously paid such Debt.
- (b) Each of the Parties acknowledges that the rights of each Party to consummate the Contemplated Transactions are unique and recognizes and affirms that irreparable damage would occur, money damages would be inadequate and the non-breaching Party would have no adequate remedy at law in the event that any of the provisions of this Agreement are not performed in accordance with their specific terms or are otherwise breached. Accordingly, the Parties agree that such non-breaching Party shall have the right, in addition to any other rights and remedies existing in its favor at law or in equity, to enforce its rights and the other Party's obligations hereunder not only by an action or actions for damages but also by an action or actions for specific performance, injunctive and/or other equitable relief (without proof of damages and without posting of bond or other security) to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement. If any Party brings any action to enforce specifically the performance of the terms and provisions hereof by any other Party, the Party bringing such action may unilaterally extend the End Date (notwithstanding the termination provisions of Article 10), so long as the Party bringing such action is actively seeking a court order for an injunction or injunctions or to specifically enforce the terms and provisions of this Agreement or any Closing

Document. Each Party hereto agrees that it will not oppose the granting of an injunction, specific performance or other equitable relief on the basis that (a) there is an appropriate remedy at law or (b) an award of specific performance is not an appropriate remedy for any reason at law or in equity. In no event shall the exercise of a Party's right to seek specific performance pursuant to this <u>Section 11.2(b)</u> reduce, restrict or otherwise limit such Party's right to terminate this Agreement pursuant to <u>Article 10</u> or pursue any other applicable remedy at law, including seeking money damages.

(c) No breach of any representation, warranty, covenant or agreement contained herein shall give rise to any right on the part of any Party, after the consummation of the Contemplated Transactions, to rescind this Agreement, any of the Related Documents or any of the Contemplated Transactions.

#### 11.3Exclusive Remedy.

Following the Closing, the sole and exclusive remedy for any and all claims against the Seller Parent or Seller arising under, out of, or related to this Agreement or the Closing Documents (excluding the Restrictive Covenant Agreements and the Transition Services Agreement), except in the case of Fraud by the Seller Parent or Seller, shall be the limited remedies provided in Section 11.1. The Seller Parent and Seller in approving this Agreement has specifically relied upon the provisions of this Section 11.3 and the limited remedies provided in Section 11.1 in agreeing to the Purchase Price and the terms and conditions of this Agreement.

#### 12. TAX MATTERS

#### 12.1General.

- (a) Seller or Seller Parent will prepare or cause to be prepared and timely file or cause to be timely filed all income Tax Returns for the Acquired Companies for all taxable periods ending on or prior to the Closing Date (each, a "Pre-Closing Income Tax Return"). Buyer will prepare or cause to be prepared and timely file or cause to be timely filed all other Tax Returns for any taxable period ending on or prior to the Closing Date with respect to the Acquired Companies that have not been filed as of the Closing Date (each, a "Pre-Closing Non-Income Tax Return"). Buyer will prepare or cause to be prepared and timely file or cause to be timely filed all Tax Returns for the Straddle Period with respect to the Acquired Companies (each, a "Straddle Period Return" and, together with the Pre-Closing Non-Income Tax Returns, the "Buyer Prepared Returns"). All Pre-Closing Income Tax Returns and Buyer Prepared Returns shall be prepared in accordance with past practice of the applicable Acquired Company to the extent supportable on a "more likely than not" (or higher) basis of confidence. Each party shall furnish the other party with a copy of such Tax Returns at least twenty (20) days before such Tax Returns are filed for the other party's review and comment on each such Tax Return for the Pre-Closing Tax Period filed after the Closing Date. Each party shall consider in good faith any changes suggested by the other party on such Tax Returns.
- (b) For purposes of this Agreement, the portion of any Tax that relates to a Straddle Period will be allocated between the Pre-Closing Straddle Period and the period of the Straddle Period that extends from the day immediately after the Closing Date to the end of the

Straddle Period in accordance with this <u>Section 12.1(b)</u>. The portion of such Tax attributable to the Pre-Closing Straddle Period shall (A) in the case of property Taxes and other similar Taxes imposed on a periodic basis, such Taxes shall be deemed to be the amount of such Tax for the entire Straddle Period (or, in the case of such Taxes determined on an arrears basis, the amount of such Taxes for the immediately preceding period) multiplied by a fraction, the numerator of which is the number of days in the Straddle Period ending on the Closing Date and the denominator of which is the total number of days in the Straddle Period, and (B) in the case of all other Taxes, on a "closing of the books basis" by assuming that the books of the Acquired Companies were closed at the end of the day on the Closing Date; provided, however, that for purposes of clause (ii), exemptions, allowances, or deductions that are calculated on an annual basis (including depreciation and amortization deductions) shall be apportioned to the Pre-Closing Straddle Period in proportion to the number of days out of the entire Straddle Period occurring on or prior to the Closing Date.

- (c) Without the written consent of Seller (not to be unreasonably withheld, conditioned or delayed), Buyer and the Acquired Companies shall not, and shall cause their respective affiliates not to, (A) extend or waive, or cause to be extended or waived, or permit the Acquired Companies or extend or waive, any statute of limitations or other period for the assessment of any Tax or deficiency related to a Pre-Closing Tax Period, (B) make or change any Tax election or accounting method or practice that has retroactive effect to any Pre-Closing Tax Period, (C) initiate any voluntary disclosure process regarding any Tax or Tax Return of the Acquired Companies for any Pre-Closing Tax Period, or (D) except as otherwise provided in Section 12.1(a), file, amend or otherwise modify any Tax Return of the Acquired Companies for any Pre-Closing Tax Period.
- (d) The parties will cooperate with each other to provide each other with such assistance as may be reasonably requested by them in connection with the preparation of any Tax Returns or amended Tax Returns, any Tax audit or other examination in connection with an administrative or judicial proceeding involving a taxing authority relating to Taxes, and the enforcement of the provisions of this <u>Article 12</u>. Such cooperation will include, including upon Seller's request, providing records and information that are reasonably relevant to any such matters and making personnel available on a mutually convenient basis to provide additional information.
- (e) Seller shall promptly notify Buyer in writing upon receipt by, and Buyer shall promptly notify Seller in writing upon receipt by Buyer, any of its affiliates, or the Acquired Companies, of notice of any Tax Proceeding; provided, however, that failure to provide notice of a Tax Proceeding shall not relieve any party of its obligations pursuant to this Agreement except to the extent such party was materially prejudiced by such failure. Buyer shall afford Seller, at Seller's expense, the opportunity to control the conduct of such Tax Proceeding relating to any income Tax Returns of the Acquired Companies and Buyer shall have the right to attend or participate in such Tax Proceeding. Neither Buyer nor Seller (if Seller has exercised its right to control such Tax Proceeding) shall settle, compromise or concede any such Tax Proceeding without the written consent of the other party, which such written consent shall not be unreasonably withheld, delayed or conditioned.
- (f) The income Tax Returns for the Acquired Companies for the taxable period ending on the Closing Date shall include any income Tax deductions attributable to a Seller's

Expense that are properly deductible by the Acquired Companies for such period to the extent supported by a "more likely than not" or higher level of confidence.

(g) Any federal, state, local or non-U.S. sales, use, stock transfer, real estate transfer, real estate gains, transfer, stamp, registration, documentary, recording or other similar taxes or fees, including all interest, additions, surcharges, fees or penalties related thereto, arising out of or incurred in connection with the transactions contemplated by this Agreement ("**Transfer Taxes**"), will be paid fifty percent (50%) by Buyer, on the one hand, and fifty percent (50%) by Seller, on the other hand. The party required by applicable Legal Requirements shall timely file any Tax Return or other document with respect to Transfer Taxes (and Seller and Buyer shall cooperate with respect to filing such Tax Returns and reducing or eliminating any Transfer Taxes).

#### 13. MISCELLANEOUS

#### 13.1Expenses.

- (a) Except as otherwise provided in this Agreement or the other documents to be delivered pursuant to this Agreement, each Party will bear its respective fees and expenses incurred in connection with the preparation, negotiation, execution, and performance of this Agreement and the consummation and performance of the Contemplated Transactions, including all fees and expenses of its Representatives. Buyer will pay all of any application and underwriting fees (including expenses of insurer's counsel, if applicable) and the premium for, the R&W Policy. The obligation of each Party to bear its own fees and expenses will be subject to any rights of such Party arising from a breach of this Agreement by another Party. For the avoidance of doubt, Seller shall be responsible for all expenses of the Acquired Companies.
- (b) All stamp, documentary, and other transfer Taxes (including any penalties and interest) incurred in connection with this Agreement, whether pertaining to the Membership Interests or any assets and properties of the Acquired Companies, will be paid by Buyer. Buyer will, at its own expense, file all necessary Tax Returns and other documentation with respect to all such Taxes.
- **13.2Seller's Disclosure Statement**. The information contained in the Seller's Disclosure Statement (or expressly incorporated therein) shall qualify the representations and warranties made in the identically numbered Section or, if applicable, subsection of this Agreement and the representations and warranties made in any other Section or subsection of this Agreement to the extent the applicability of such information in the Seller's Disclosure Statement to such other Section or subsection is reasonably apparent on its face.
- **13.3Further Assurance**. The Parties will (a) execute and deliver to each other such other documents and (b) do such other acts and things as a Party may reasonably request for the purpose of carrying out the intent of this Agreement, the Contemplated Transactions, and the documents to be delivered pursuant to this Agreement.
- **13.4Entire Agreement**. This Agreement supersedes all prior agreements, whether written or oral, between the Parties with respect to its subject matter (including any letter of intent and, upon the Closing, any confidentiality obligation to which Buyer is subject) and constitutes (along with the Seller's Disclosure Statement and Buyer's Disclosure, the exhibits, annexes, and

the other documents and agreements to be delivered pursuant to this Agreement, including the Closing Documents) a complete and exclusive statement of the terms of the agreement between the Parties with respect to the subject matter of this Agreement.

- **13.5Modification**. This Agreement may only be amended, supplemented, or otherwise modified by a writing executed by each of the Parties hereto.
- **13.6Assignments and Successors**. No Party may assign any of its rights or delegate any of its obligations under this Agreement without the prior consent of the other Party, except that Buyer may collaterally assign its rights under this Agreement to any financial institution providing financing in connection with the Contemplated Transactions, to an Affiliate of Buyer, or to a successor in interest to Buyer. Any purported assignment of rights or delegation of obligations in violation of this Section 13.6 will be void. Subject to the foregoing, this Agreement will apply to, be binding in all respects upon, and inure to the benefit of the heirs, executors, administrators, legal representatives, successors, and permitted assigns of the Parties.
- **13.7No Third Party Rights**. Other than the Indemnified Parties and the Parties, no Person will have any legal or equitable right, remedy, or claim under or with respect to this Agreement. This Agreement may be amended or terminated, and any provision of this Agreement may be waived, without the consent of any Person who is not a party to the Agreement.
- **13.8Governing Law**. All matters relating to or arising out of this Agreement or any Contemplated Transactions and the rights of the Parties (whether sounding in contract, tort, or otherwise) will be governed by and construed and interpreted under the laws of the State of Delaware without regard to conflicts of laws principles that would require the application of any other law.
- **13.9Jurisdiction; Service of Process**. Except as otherwise provided in this Agreement, any Proceeding arising out of or relating to this Agreement or any Contemplated Transactions shall be brought in the courts of the State of Tennessee, County of Williamson, or, if it has or can acquire jurisdiction, in the United States District Court for the Middle District of Tennessee, and each of the Parties irrevocably submits to the exclusive jurisdiction of each such court in any such Proceeding, waives any objection it may now or hereafter have to venue or to convenience of forum, agrees that all claims in respect of such Proceeding shall be heard and determined only in any such court, and agrees not to bring any Proceeding arising out of or relating to this Agreement or any Contemplated Transactions in any other court. Each Party acknowledges and agrees that this Section 13.9 constitutes a voluntary and bargained-for agreement between the Parties. Process in any Proceeding referred to in the first sentence of this Section may be served on any Party anywhere in the world, including by sending or delivering a copy of the process to the Party to be served at the address and in the manner provided for the giving of notices in Section 13.14. Nothing in this Section 13.9 will affect the right of any Party to serve legal process in any other manner permitted by law or at equity.

13.10Waiver of Jury Trail. EACH PARTY, KNOWINGLY, VOLUNTARILY, AND INTENTIONALLY, WAIVES ITS RIGHT TO TRIAL BY JURY IN ANY PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY CONTEMPLATED

#### TRANSACTIONS, WHETHER SOUNDING IN CONTRACT, TORT, EQUITY, OR OTHERWISE.

- **13.11Attorneys' Fees**. In the event any Proceeding is brought in respect of this Agreement or any of the documents referred to in this Agreement, the prevailing party will be entitled to recover reasonable attorneys' fees and other reasonable costs incurred in such Proceeding, in addition to any relief to which such party may be entitled.
- **13.12Enforcement of Agreement**. Each Party acknowledges and agrees that the other Party would be irreparably harmed if any of the provisions of this Agreement or any Closing Document are not performed in accordance with their specific terms and that any breach of this Agreement or any Closing Document by a Party could not be adequately compensated in all cases by monetary damages alone. Accordingly, each Party agrees that, in addition to any other right or remedy to which a Party may be entitled at law or in equity, each Party shall be entitled to enforce any provision of this Agreement or any Closing Document by a decree of specific performance and to obtain temporary, preliminary, and permanent injunctive relief to prevent breaches or threatened breaches, without posting any bond or giving any other undertaking.
- **13.13No Waiver**. Neither any failure nor any delay by any Party in exercising any right, power, or privilege under this Agreement or any of the documents referred to in this Agreement will operate as a waiver of such right, power, or privilege, and no single or partial exercise of any such right, power, or privilege will preclude any other or further exercise of such right, power, or privilege or the exercise of any other right, power, or privilege. To the maximum extent permitted by applicable Legal Requirements, (a) no claim or right arising out of this Agreement or any of the documents referred to in this Agreement can be waived by a Party, in whole or in part, unless made in a writing signed by such Party; (b) a waiver given by a Party will only be applicable to the specific instance for which it is given; and (c) no notice to or demand on a Party will (i) waive or otherwise affect any obligation of that Party or (ii) affect the right of the Party giving such notice or demand to take further action without notice or demand as provided in this Agreement or the documents referred to in this Agreement.
- **13.14Notices.** All notices and other communications required or permitted by this Agreement shall be in writing and will be effective, and any applicable time period shall commence, when (a) delivered to the following address by hand or by a nationally recognized overnight courier service (costs prepaid) addressed to the following address or (b) transmitted electronically to the following e-mail addresses, in each case marked to the attention of the Person (by name or title) designated below (or to such other address, e-mail address, or Person as a Party may designate by notice to the other parties):

Seller Parent or Seller:

American Health Companies, LLC 201 Jordan Road, Suite 200 Franklin, TN 37067 Attention: President

Email: MBailev@amhealthpartners.com

with a copy to:

Philip Clark General Counsel 201 Jordan Road, Suite 200 Franklin, TN 37067 Email: PClark@amhealthpartners.com

and to

**Bradley Arant Boult Cummings LLP** Attention: John W. Titus 1600 Division Street, Suite 700 Nashville, TN 37203 E-mail address: jtitus@bradley.com

**Buyer:** 

Addus HealthCare, Inc. 6303 Cowboys Way, Suite 600 Frisco, Texas 75034 Attn: Sean Gaffney, EVP & Chief Legal Officer Email: sgaffney@addus.com

with a copy to:

Bass, Berry & Sims PLC 150 Third Ave. South, Suite 2800 Nashville, Tennessee 37201 Attn: David Cox

Email: dcox@bassberry.com

13.15Post-Closing Representation of Seller. Each of the Parties hereto acknowledges and agrees, on its own behalf and on behalf of Representatives and Related Persons that Bradley Arant Boult Cummings LLP ("Bradley") has acted as counsel to Seller Parent, Seller, and the Acquired Companies in connection with the negotiation, preparation, execution and delivery of this Agreement and the consummation of Contemplated Transactions. Buyer agrees, and shall cause the Acquired Companies to agree, that, following the Closing, such representation and any prior representation of the Acquired Companies by Bradley shall not preclude Bradley from serving as counsel to Seller, Seller Parent, or any Related Person of Seller or Seller Parent in connection with any litigation, claim or obligation arising out of or relating to this Agreement or the Contemplated Transactions. Each of the Parties hereby consents thereto and waives any conflict of interest arising from such prior representation, and each of such Parties shall cause any of its Related Persons to consent to waive any conflict of interest arising from such representation.

- 13.16Attorney Client Privilege Carve Out. All communications involving attorney-client confidences between Seller Parent, Seller, Acquired Companies, or their Related Persons and Bradley pertaining to the negotiation of this Agreement and the documentation and consummation of the Contemplated Transactions that are legally privileged (the "Privileged Communications") belong solely to Seller Parent, Seller, and their Related Persons (and not the Acquired Companies) and the expectation of client confidence relating thereto shall belong solely to Seller Parent and Seller and shall not pass to or be claimed by Buyer or the Acquired Companies. Accordingly, Buyer acknowledges that neither the Acquired Companies nor Buyer shall have access to any such communications or to the files of Bradley relating to such matters, whether or not the Closing shall have occurred, and the Buyer and the Acquired Companies hereby covenant and agree that they shall use commercially reasonable efforts to avoid a knowing waiver of any Privileged Communications that may be held in any computer files, email files, servers, or physical books and records that may be in the possession of Buyer or the Acquired Companies after the Closing.
- **13.17Severability**. If any provision of this Agreement is held invalid or unenforceable by any court of competent jurisdiction, the other provisions of this Agreement will remain in full force and effect. Any provision of this Agreement held invalid or unenforceable only in part or degree will remain in full force and effect to the extent not held invalid or unenforceable. Upon determination that any term or other provision is invalid, illegal, or incapable of being enforced, the parties hereto will negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner to the end that the transactions contemplated hereby are fulfilled to the fullest extent possible.
- **13.18Time of Essence**. With regard to all dates and time periods set forth or referred to in this Agreement, time is of the essence.

#### 13.19Counterparts and Electronic Signature

- (a) This Agreement and other documents to be delivered pursuant to this Agreement may be executed in one or more counterparts, each of which will be deemed to be an original copy and all of which, when taken together, will be deemed to constitute one and the same agreement or document, and will be effective when counterparts have been signed by each of the Parties and delivered to the other Parties.
- (b) A manual signature on this Agreement or other documents to be delivered pursuant to this Agreement, an image of which shall have been transmitted electronically, will constitute an original signature for all purposes. The delivery of copies of this Agreement or other documents to be delivered pursuant to this Agreement, including executed signature pages where required, by electronic transmission will constitute effective delivery of this Agreement or such other document for all purposes.

[signatures on following page]

**IN WITNESS WHEREOF**, the Parties have executed and delivered this Membership Interests Purchase Agreement as of the date first written above.

## **BUYER:**

Addus HealthCare, Inc.

By: <u>/s/ R. Dirk Allison</u>
Name: <u>R. Dirk Allison</u>

Title: <u>Chief Executive Officer</u>

# **SELLER PARENT:**

American Health Companies, LLC

By: <u>/s/ Michael Bailey</u>
Name: <u>Michael Bailey</u>

Title: <u>Chief Executive Officer</u>

## **SELLER:**

HHH Newco Holdings, LLC

By: <u>/s/ Michael Bailey</u>
Name: <u>Michael Bailey</u>

Title: <u>Chief Executive Officer</u>

# **ACQUIRED COMPANIES:**

American Home Care, LLC

By: <u>/s/ Michael Bailey</u>
Name: <u>Michael Bailey</u>

Title: <u>Chief Executive Officer</u>

## Homecare, LLC

By: <u>/s/ Michael Bailey</u>
Name: <u>Michael Bailey</u>

Title: <u>Chief Executive Officer</u>

Tri-County Home Health and Hospice, LLC

By: <u>/s/ Michael Bailey</u>
Name: <u>Michael Bailey</u>

Title: <u>Chief Executive Officer</u>

Tennessee Valley Home Care, LLC

By: <u>/s/ Michael Bailey</u>
Name: <u>Michael Bailey</u>

Title: Chief Executive Officer

# CERTIFICATIONS OF CHIEF EXECUTIVE OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

#### I, R. Dirk Allison, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Addus HomeCare Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to
    provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance
    with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 1, 2023 By: /s/ R. Dirk Allison

R. Dirk Allison

Chairman and Chief Executive Officer

# CERTIFICATIONS OF CHIEF FINANCIAL OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

#### I, Brian Poff, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Addus HomeCare Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 1, 2023

By: /s/ Brian Poff

Brian Poff

Chief Financial Officer

# CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. SECTION 1350 (AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002)

In connection with the Quarterly Report on Form 10-Q for the fiscal quarter ended June 30, 2023 of Addus HomeCare Corporation (the "Company") as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, R. Dirk Allison, Chairman and Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 1, 2023 By: /s/ R. Dirk Allison

R. Dirk Allison

Chairman and Chief Executive Officer

# CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350 (AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002)

In connection with the Quarterly Report on Form 10-Q for the fiscal quarter ended June 30, 2023 of Addus HomeCare Corporation (the "Company") as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Brian Poff, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 1, 2023 By: /s/ Brian Poff

Brian Poff

Chief Financial Officer